UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

		FORM 10-Q	
X	QUARTERLY REPORT PURSUANT TO S	SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE AC	T OF
	For the qu	narterly period ended June 28, 2014 or	
	TRANSITION REPORT PURSUANT TO S	SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE AC	CT OF
	For the transition	on period from to	
		ssion file number: 001-32891	
	Har	nesbrands Inc.	
	(Exact name	of registrant as specified in its charter)	
	Maryland	20-3552316	
	(State of incorporation)	(I.R.S. employer identification no.)	
	1000 East Hanes Mill Road	25425	
	Winston-Salem, North Carolina	27105	
	(Address of principal executive office)	(Zip code) (336) 519-8080	
	(Registran	(330) 313-3000 t's telephone number including area code)	
.934 d		all reports required to be filed by Section 13 or 15(d) of the Securities Exchange A hat the registrant was required to file such reports), and (2) has been subject to suc	
equire		l electronically and posted on its corporate Web site, if any, every Interactive Data lation S-T during the preceding 12 months (or for such shorter period that the region	
		elerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting co and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one)	
Large	accelerated filer x	Accelerated filer	
Non-a	ccelerated filer \Box (Do not check if a smaller	reporting company) Smaller reporting company	
I	ndicate by check mark whether the registrant is a shell com	pany (as defined in Rule 12b-2 of the Exchange Act). Yes \Box No x	
I	As of July 18, 2014, there were 99,631,359 shares of the reg	gistrant's common stock outstanding.	

TABLE OF CONTENTS

		Page
Forward-Lo	poking Statements	<u>1</u>
Where You	Can Find More Information	<u>1</u>
PART I		
Item 1.	Financial Statements (unaudited):	
10111 11	Condensed Consolidated Statements of Income for the quarters and six months ended June 28, 2014 and June 29, 2013	<u>2</u>
	Condensed Consolidated Statements of Comprehensive Income for the quarters and six months ended June 28, 2014 and June 29, 2013	<u>=</u> <u>3</u>
	Condensed Consolidated Balance Sheets at June 28, 2014 and December 28, 2013	
		<u>4</u>
	Condensed Consolidated Statements of Cash Flows for the six months ended June 28, 2014 and June 29, 2013	<u>5</u>
	Notes to Condensed Consolidated Financial Statements	<u>6</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>20</u>
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	<u>28</u>
Item 4.	Controls and Procedures	<u>28</u>
PART II		
Item 1.	<u>Legal Proceedings</u>	<u>29</u>
Item 1A.	Risk Factors	<u>29</u>
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	<u>29</u>
Item 3.	<u>Defaults Upon Senior Securities</u>	<u>29</u>
Item 4.	Mine Safety Disclosures	<u>29</u>
Item 5.	Other Information	<u>29</u>
Item 6.	<u>Exhibits</u>	<u>29</u>
<u>Signatures</u>		<u>30</u>
Index to Ex	hibits	E-1

Trademarks, Trade Names and Service Marks

We own or have rights to use the trademarks, service marks and trade names that we use in conjunction with the operation of our business. Some of the more important trademarks that we own or have rights to use that may appear in this Quarterly Report on Form 10-Q include the *Hanes, Champion, C9 by Champion, Bali, Playtex, Maidenform, JMS/Just My Size, L'eggs, Flexees, barely there, Wonderbra, Gear for Sports, Lilyette, Zorba, Rinbros and Sol y Oro marks, which may be registered in the United States and other jurisdictions. We do not own any trademark, trade name or service mark of any other company appearing in this Quarterly Report on Form 10-Q.*

FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q includes forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Forward-looking statements include all statements that do not relate solely to historical or current facts, and can generally be identified by the use of words such as "may," "believe," "will," "expect," "project," "estimate," "intend," "anticipate," "plan," "continue" or similar expressions. In particular, statements under the heading "Outlook" and other information appearing under "Management's Discussion and Analysis of Financial Condition and Results of Operations" include forward-looking statements. Forward-looking statements inherently involve many risks and uncertainties that could cause actual results to differ materially from those projected in these statements.

Where, in any forward-looking statement, we express an expectation or belief as to future results or events, such expectation or belief is based on the current plans and expectations of our management, expressed in good faith and believed to have a reasonable basis. However, there can be no assurance that the expectation or belief will result or will be achieved or accomplished. More information on factors that could cause actual results or events to differ materially from those anticipated is included from time to time in our reports filed with the Securities and Exchange Commission (the "SEC"), including our Annual Report on Form 10-K for the year ended December 28, 2013, under the caption "Risk Factors," as well in the "Investors" section of our corporate website, www.Hanes.com/investors.

All forward-looking statements speak only as of the date of this Quarterly Report on Form 10-Q and are expressly qualified in their entirety by the cautionary statements included in this Quarterly Report on Form 10-Q or our Annual Report on Form 10-K for the year ended December 28, 2013, particularly under the caption "Risk Factors." We undertake no obligation to update or revise forward-looking statements that may be made to reflect events or circumstances that arise after the date made or to reflect the occurrence of unanticipated events, other than as required by law.

WHERE YOU CAN FIND MORE INFORMATION

We file annual, quarterly and current reports, proxy statements and other information with the SEC. You can read our SEC filings over the Internet at the SEC's website at *www.sec.gov*. To receive copies of public records not posted to the SEC's web site at prescribed rates, you may complete an online form at *www.sec.gov*, send a fax to (202) 772-9337 or submit a written request to the SEC, Office of FOIA/PA Operations, 100 F Street, N.E., Washington, D.C. 20549. Please call the SEC at 1-800-SEC-0330 for further information.

We make available free of charge at www.Hanes.com/investors (in the "Investors" section) copies of materials we file with, or furnish to, the SEC. By referring to our corporate website, www.Hanes.com/corporate, or any of our other websites, we do not incorporate any such website or its contents into this Quarterly Report on Form 10-Q.

PART I

Item 1. Financial Statements

HANESBRANDS INC.

Condensed Consolidated Statements of Income (in thousands, except per share amounts) (unaudited)

	Quarter Ended				Six Months Ended			
		June 28, 2014		June 29, 2013		June 28, 2014		June 29, 2013
Net sales	\$	1,342,052	\$	1,199,205	\$	2,401,422	\$	2,144,666
Cost of sales		837,698		763,723		1,540,291		1,381,885
Gross profit		504,354		435,482		861,131		762,781
Selling, general and administrative expenses		297,230		254,035		582,219		496,191
Operating profit		207,124		181,447		278,912		266,590
Other expenses		660		751		1,095		1,215
Interest expense, net		21,119		25,221		42,937		50,844
Income before income tax expense		185,345		155,475		234,880		214,531
Income tax expense		30,767		33,889		38,742		41,566
Net income	\$	154,578	\$	121,586	\$	196,138	\$	172,965
Earnings per share:	<u></u>							
Basic	\$	1.54	\$	1.22	\$	1.95	\$	1.74
Diluted	\$	1.51	\$	1.19	\$	1.92	\$	1.70

Condensed Consolidated Statements of Comprehensive Income (in thousands) (unaudited)

		Quarter Ended				Six Months Ended			
	June 28, 2014			June 29, 2013		June 28, 2014		June 29, 2013	
Net income	\$	154,578	\$	121,586	\$	196,138	\$	172,965	
Other comprehensive income (loss), net of tax of \$481, \$2,175, \$1,288 and \$3,671, respectively		3,968		(3,260)		3,187		(1,904)	
Comprehensive income	\$	158,546	\$	118,326	\$	199,325	\$	171,061	

Condensed Consolidated Balance Sheets (in thousands, except share and per share amounts) (unaudited)

	June 28, 2014	December 28, 2013
Assets		
Cash and cash equivalents	\$ 143,617	\$ 115,863
Trade accounts receivable, net	779,520	578,558
Inventories	1,401,615	1,283,331
Deferred tax assets	198,149	197,260
Other current assets	64,835	68,654
Total current assets	2,587,736	2,243,666
Property, net	571,744	579,883
Trademarks and other identifiable intangibles, net	367,820	377,751
Goodwill	626,568	626,392
Deferred tax assets	208,703	207,426
Other noncurrent assets	52,447	54,930
Total assets	\$ 4,415,018	\$ 4,090,048
Liabilities and Stockholders' Equity		
Accounts payable	\$ 563,758	\$ 466,270
Accrued liabilities	325,778	315,026
Notes payable	40,802	36,192
Accounts Receivable Securitization Facility	225,000	181,790
Total current liabilities	1,155,338	999,278
Long-term debt	 1,523,000	1,467,000
Pension and postretirement benefits	220,014	263,819
Other noncurrent liabilities	138,238	129,328
Total liabilities	3,036,590	2,859,425
Stockholders' equity:		
Preferred stock (50,000,000 authorized shares; \$.01 par value)		
Issued and outstanding — None	_	_
Common stock (500,000,000 authorized shares; \$.01 par value)		
Issued and outstanding — 99,630,343 and 99,455,478, respectively	996	995
Additional paid-in capital	294,514	285,227
Retained earnings	1,316,748	1,181,418
Accumulated other comprehensive loss	(233,830)	(237,017)
Total stockholders' equity	1,378,428	1,230,623
Total liabilities and stockholders' equity	\$ 4,415,018	\$ 4,090,048

Condensed Consolidated Statements of Cash Flows (in thousands) (unaudited)

	Six Mo	onths Ended
	June 28, 2014	June 29, 2013
Operating activities:		
Net income	\$ 196,138	\$ \$ 172,965
Adjustments to reconcile net income to net cash from operating activities:		
Depreciation and amortization of long-lived assets	46,040	45,630
Amortization of debt issuance costs	2,853	3,358
Stock compensation expense	8,554	5,003
Deferred taxes and other	(4,611	2,444
Changes in assets and liabilities:		
Accounts receivable	(200,973	(115,907)
Inventories	(117,576	(91,466)
Other assets	6,560	(4,813)
Accounts payable	93,954	42,808
Accrued liabilities and other	(26,001	(41,696)
Net cash from operating activities	4,938	18,326
Investing activities:		
Purchases of property, plant and equipment	(24,585	(19,790)
Proceeds from sales of assets	4,918	3,617
Net cash from investing activities	(19,667	(16,173)
Financing activities:		
Borrowings on notes payable	66,737	62,954
Repayments on notes payable	(61,957	(58,671)
Borrowings on Accounts Receivable Securitization Facility	115,609	81,358
Repayments on Accounts Receivable Securitization Facility	(72,399	(84,715)
Borrowings on Revolving Loan Facility	1,782,500	1,970,000
Repayments on Revolving Loan Facility	(1,726,500	(1,913,000)
Cash dividends paid	(59,731) (19,797)
Proceeds from stock options exercised	_	5,279
Taxes paid related to net shares settlement of equity awards	(10,342	(20,004)
Excess tax benefit from stock-based compensation	7,895	14,892
Other	806	259
Net cash from financing activities	42,618	38,555
Effect of changes in foreign exchange rates on cash	(135	(1,199)
Change in cash and cash equivalents	27,754	
Cash and cash equivalents at beginning of year	115,863	
Cash and cash equivalents at end of period	\$ 143,617	

Notes to Condensed Consolidated Financial Statements (dollars and shares in thousands, except per share data) (unaudited)

(1) Basis of Presentation

These statements have been prepared pursuant to the rules and regulations of the Securities and Exchange Commission (the "SEC") and, in accordance with those rules and regulations, do not include all information and footnote disclosures normally included in annual financial statements prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP"). Management believes that the disclosures made are adequate for a fair statement of the results of operations, financial condition and cash flows of Hanesbrands Inc., a Maryland corporation, and its consolidated subsidiaries (the "Company" or "Hanesbrands"). In the opinion of management, the condensed consolidated interim financial statements reflect all adjustments, which consist only of normal recurring adjustments, necessary to state fairly the results of operations, financial condition and cash flows for the interim periods presented herein. The preparation of condensed consolidated financial statements in conformity with GAAP requires management to make use of estimates and assumptions that affect the reported amounts and disclosures. Actual results may vary from these estimates.

These condensed consolidated interim financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's most recent Annual Report on Form 10-K. The results of operations for any interim period are not necessarily indicative of the results of operations to be expected for the full year.

(2) Recent Accounting Pronouncements

Disclosures About Offsetting Assets and Liabilities

In December 2011, the Financial Accounting Standards Board (the "FASB") issued new accounting rules related to new disclosure requirements regarding the nature of an entity's rights of setoff and related arrangements associated with its financial instruments and derivative instruments. The new rules were effective for the Company in the first quarter of 2014 with retrospective application required. The adoption of the new accounting rules did not have a material effect on the Company's financial condition, results of operations or cash flows.

Presentation of an Unrecognized Tax Benefit

In July 2013, the FASB issued new accounting rules related to standardizing the financial statement presentation of an unrecognized tax benefit, or a portion thereof, when a net operating loss carryforward, a similar tax loss, or a tax credit carryforward exists. The new rules were effective for the Company in the first quarter of 2014 and applied prospectively. The adoption of the new accounting rules did not have a material effect on the Company's financial condition, results of operations or cash flows.

Discontinued Operations

In April 2014, the FASB issued new accounting rules related to updating the criteria for reporting discontinued operations and enhancing related disclosures requirements. The new rules are effective for the Company in the first quarter of 2015. The Company does not expect the adoption of the new accounting rules to have a material impact on the Company's financial condition, results of operations or cash flows.

Revenue from Contracts with Customers

In May 2014, the FASB issued new accounting rules related to revenue recognition for contracts with customers requiring revenue recognition based on the transfer of promised goods or services to customers in an amount that reflects consideration the Company expects to be entitled to in exchange for goods or services. The new rules supercede prior revenue recognition requirements and most industry-specific accounting guidance. The new rules will be effective for the Company in the first quarter of 2017 with retrospective application required. The Company does not expect the adoption of the new accounting rules to have a material impact on the Company's financial condition, results of operations or cash flows.

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

(3) Acquisitions

On June 25, 2014, the Company announced its intentions to acquire DBA Lux Holding S.A. ("DBA") from SLB Brands Holdings, Ltd and certain individual DBA shareholders, provided consultation with European and French works councils representing DBA employees is completed and customary closing conditions are met. DBA is a leading marketer of intimate apparel, hosiery and underwear in Europe. The all-cash transaction offer values DBA at €400,000 on an enterprise basis (approximately \$550,000 at current exchange rates). The Company intends to finance the transaction with cash on hand and third-party borrowings. The acquisition could close as soon as the third quarter 2014. For additional information regarding the transaction, see the Company's Current Report on Form 8-K filed on June 25, 2014.

In October 2013, the Company acquired 100% of the outstanding shares of Maidenform Brands, Inc. ("Maidenform") at \$23.50 per share for a total purchase price of \$580,505. The acquisition was financed through a combination of cash on hand and short-term borrowing on the Company's revolving credit facility.

Maidenform is a global intimate apparel brand with a portfolio of well-known brands including *Maidenform*, *Flexees* and *Lilyette*. The Company believes the acquisition will create growth and cost savings opportunities and increased scale to serve retailers. Maidenform sourced all of its products from manufacturers, while the Company utilizes its low cost supply chain supplemented by third party manufacturing to maximize the value of Maidenform to retailers and consumers.

The allocation of the Maidenform purchase price is preliminary and subject to change. For the six months ended June 28, 2014, the Company has not recorded any purchase price adjustments. The primary areas of the purchase price that are not yet finalized are related to certain income taxes and residual goodwill. Accordingly, adjustments will be made to the values of the assets acquired and liabilities assumed as additional information is obtained about the facts and circumstances which existed at the valuation date.

(4) Earnings Per Share

Basic earnings per share ("EPS") was computed by dividing net income by the number of weighted average shares of common stock outstanding. Diluted EPS was calculated to give effect to all potentially dilutive shares of common stock using the treasury stock method. The reconciliation of basic to diluted weighted average shares outstanding is as follows:

	Quarter	Ended	Six Months Ended			
	June 28, 2014	June 29, 2013	June 28, 2014	June 29, 2013		
Basic weighted average shares outstanding	100,480	99,855	100,432	99,624		
Effect of potentially dilutive securities:						
Stock options	1,163	1,494	1,211	1,437		
Restricted stock units	414	663	393	668		
Employee stock purchase plan and other	_	1	_	_		
Diluted weighted average shares outstanding	102,057	102,013	102,036	101,729		

For the quarters and six months ended June 28, 2014 and June 29, 2013, there were no options or restricted stock units that were excluded from the diluted earnings per share calculation because their effect would be anti-dilutive.

(5) Inventories

Inventories consisted of the following:

	June 28, 2014	December 28, 2013
Raw materials	\$ 192,592	\$ 170,524
Work in process	150,860	142,713
Finished goods	1,058,163	970,094
	\$ 1,401,615	\$ 1,283,331

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

(6) Debt

Debt consisted of the following:

	Interest Rate as of	 Principa	ıl Am	ount	
	June 28, 2014	June 28, 2014		December 28, 2013	Maturity Date
Revolving Loan Facility	1.91%	\$ 523,000	\$	467,000	July 2018
6.375% Senior Notes	6.38%	1,000,000		1,000,000	December 2020
Accounts Receivable Securitization Facility	1.11%	225,000		181,790	March 2015
		 1,748,000		1,648,790	
Less current maturities		225,000		181,790	
		\$ 1,523,000	\$	1,467,000	

As of June 28, 2014, the Company had \$566,113 of borrowing availability under the \$1,100,000 revolving credit facility (the "Revolving Loan Facility") under its senior secured credit facility after taking into account outstanding borrowings and \$10,887 of standby and trade letters of credit issued and outstanding under this facility.

In March 2014, the Company amended the accounts receivable securitization facility that it entered into in November 2007 (the "Accounts Receivable Securitization Facility"). This amendment decreased certain fee rates, revised certain concentration limits and dilution triggers and extended the termination date to March 2015.

As of June 28, 2014, the Company was in compliance with all financial covenants under its credit facilities.

(7) Accumulated Other Comprehensive Loss

The components of Accumulated other comprehensive loss ("AOCI") are as follows:

	Cumulative Translation Adjustment]	Foreign Exchange Contracts	I	Defined Benefit Plans	Income Taxes	 ccumulated Other omprehensive Loss
Balance at December 28, 2013	\$ (21,928)	\$	2,042	\$	(357,503)	\$ 140,372	\$ (237,017)
Amounts reclassified from accumulated other comprehensive loss	_		(1,030)		5,203	(1,629)	2,544
Current-period other comprehensive income (loss) activity	1,157		(855)		_	341	643
Balance at June 28, 2014	\$ (20,771)	\$	157	\$	(352,300)	\$ 139,084	\$ (233,830)

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

The Company had the following reclassifications out of Accumulated other comprehensive loss:

			Amount of Reclassification from AOCI Quarter Ended			 Amount of Reclassification from AOCI				
						Six Mon	ths Ended			
Component of AOCI	Location of Reclassification into Income		June 28, 2014		June 29, 2013	June 28, 2014		June 29, 2013		
Gain on foreign exchange										
contracts	Cost of sales	\$	355	\$	47	\$ 1,030	\$	5		
Gain on foreign exchange contracts	Income tax		(142)		(19)	(411)		(2)		
Net of tax			213		28	619		3		
Amortization of deferred actuarial loss and prior service cost	Selling, general and administrative expenses		(2,602)		(3,847)	(5,203)		(7,709)		
Amortization of deferred actuarial loss and prior service cost	Income tax		1,020		1,485	2,040		3,025		
Net of tax			(1,582)		(2,362)	 (3,163)		(4,684)		
							-			
Total reclassifications		\$	(1,369)	\$	(2,334)	\$ (2,544)	\$	(4,681)		

(8) Financial Instruments and Risk Management

The Company uses forward foreign exchange contracts to manage its exposures to movements in foreign exchange rates. As of June 28, 2014, the notional U.S. dollar equivalent of commitments to sell and purchase foreign currencies within the Company's derivative portfolio was \$61,298 and \$9,922 respectively, primarily consisting of contracts hedging exposures to the Mexican peso, Canadian dollar, Australian dollar, Brazilian real and Japanese yen.

Fair Values of Derivative Instruments

The fair values of derivative financial instruments recognized in the Condensed Consolidated Balance Sheets of the Company were as follows:

		Fair	Value		
	Balance Sheet Location	ine 28, 2014	De	ecember 28, 2013	
Hedges	Other current assets	\$ 49	\$	32	
Non-hedges	Other current assets	21		970	
Total derivative assets		 70		1,002	
Hedges	Accrued liabilities	(856)		_	
Non-hedges	Accrued liabilities	(286)		(28)	
Total derivative liabilities		 (1,142)		(28)	
Net derivative asset (liability)		\$ (1,072)	\$	974	

Cash Flow Hedges

The Company uses forward foreign exchange contracts to reduce the effect of fluctuating foreign currencies on short-term foreign currency-denominated transactions, foreign currency-denominated investments and other known foreign currency exposures. Gains and losses on these contracts are intended to offset losses and gains on the hedged transaction in an effort to reduce the earnings volatility resulting from fluctuating foreign currency exchange rates.

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

The Company expects to reclassify into earnings during the next 12 months a net loss from Accumulated other comprehensive loss of approximately \$521.

The changes in fair value of derivatives excluded from the Company's effectiveness assessments and the ineffective portion of the changes in the fair value of derivatives used as cash flow hedges are reported in the "Selling, general and administrative expenses" line in the Condensed Consolidated Statements of Income.

The effect of cash flow hedge derivative instruments on the Condensed Consolidated Statements of Income and Accumulated other comprehensive loss is as follows:

			Amou Gain (Recogn Accumula Comprehe (Effective	Loss) ized in ted Othe nsive Lo	SS		Gain Recog Accumul Compreh	ount of (Loss) nized in ated Other ensive Loss e Portion)			
			Quarter	Ended			Six Mon	ths Ended			
			June 28, 2014		June 29, 2013		June 28, 2014		ne 29, 2013		
Foreign exchange contracts		\$	(997)	\$	1,775	\$	(855)	\$	1,624		
		Amoi Gain Reclassif Accum Other Com Loss into (Effective	(Loss) ied from ulated prehensi Income	ive		Gain Reclassi Accur Other Cor Loss int	ount of (Loss) ified from nulated nprehensive to Income re Portion)	è			
	Accumulated Other Comprehensive		Quartei	Ended			Six Mon	ths Ended			
	Loss into Income (Effective Portion)		June 28, June 29, 2014 2013							June 29 2013	
Foreign exchange contracts	Cost of sales	\$	355	\$	47	\$	1,030	\$	5		

Derivative Contracts Not Designated As Hedges

The Company uses foreign exchange derivative contracts as economic hedges against the impact of foreign exchange fluctuations on existing accounts receivable and payable balances and intercompany lending transactions denominated in foreign currencies. These contracts are not designated as hedges under the accounting standards and are recorded at fair value in the Condensed Consolidated Balance Sheet. Any gains or losses resulting from changes in fair value are recognized directly into earnings. Gains or losses on these contracts largely offset the net remeasurement gains or losses on the related assets and liabilities.

The effect of derivative contracts not designated as hedges on the Condensed Consolidated Statements of Income is as follows:

			Amount of Recognized				Amount of Recognize		
	Location of Loss		Quarte	r End	ed		Six Mon	ths En	ded
	Recognized in Income on Derivative		June 28, 2014		June 29, 2013		June 28, 2014		June 29, 2013
Foreign evelopes contracts	Selling, general and administrative expenses	c	(322)	¢	2,349	¢	(372)	¢	563
Foreign exchange contracts	adillillistrative expenses	Ф	(322)	Ф	2,349	Ф	(3/2)	Ф	303

(9) Fair Value of Assets and Liabilities

As of June 28, 2014, the Company held certain financial assets and liabilities related to foreign exchange derivative contracts that are required to be measured at fair value on a recurring basis. The fair values of foreign currency derivatives are determined using the cash flows of the foreign exchange contract, discount rates to account for the passage of time and current foreign exchange market data and are categorized as Level 2. The Company's defined benefit pension plan investments are not required to be measured at fair value on a recurring basis.

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

There were no changes during the quarter ended June 28, 2014 to the Company's valuation techniques used to measure asset and liability fair values on a recurring basis. There were no transfers between the three level categories and there were no Level 3 assets or liabilities measured on a quarterly basis during the quarter ended June 28, 2014. As of and during the quarter and six months ended June 28, 2014, the Company did not have any non-financial assets or liabilities that were required to be measured at fair value on a recurring or non-recurring basis.

The following tables set forth, by level within the fair value hierarchy, the Company's financial assets and liabilities accounted for at fair value on a recurring basis.

		Assets (Liabilities) at Fair Value as of June 28, 2014									
	-	Activ for	d Prices In e Markets Identical Assets evel 1)		Significant Other Observable Inputs (Level 2)		Significant Unobservable Inputs (Level 3)				
Foreign exchange derivative contracts		\$		\$	70	\$	_				
Foreign exchange derivative contracts			_		(1,142)		_				
					(1,072)		_				
	-				•						
Deferred compensation plan liability			_		(18,128)		_				
	_				_						
Total		\$		\$	(19,200)	\$	_				
	-	_									
	_	Assets (Liabilities) at Fair Value as of December 28, 2013									
			d Prices In		Significant		G! !C! .				

	Assets (Liabilities) at Fair Value as of December 28, 2013									
	Acti	ted Prices In ive Markets r Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)			Significant Unobservable Inputs (Level 3)				
Foreign exchange derivative contracts	\$	_	\$	1,002	\$	_				
Foreign exchange derivative contracts		_		(28)		_				
	-			974		_				
Deferred compensation plan liability		_		(17,036)		_				
Total	\$		\$	(16,062)	\$	_				

Fair Value of Financial Instruments

The carrying amounts of cash and cash equivalents, trade accounts receivable, notes receivable and accounts payable approximated fair value as of June 28, 2014 and December 28, 2013. The carrying amount of trade accounts receivable includes allowance for doubtful accounts, chargebacks and other deductions of \$16,796 and \$13,336 as of June 28, 2014 and December 28, 2013, respectively. The fair value of debt, which is classified as a Level 2 liability, was \$1,831,635 and \$1,744,115 as of June 28, 2014 and December 28, 2013, respectively. Debt had a carrying value of \$1,748,000 and \$1,648,790 as of June 28, 2014 and December 28, 2013, respectively amounts of the Company's notes payable, which is classified as a Level 2 liability, approximated fair value as of June 28, 2014 and December 28, 2013, primarily due to the short-term nature of these instruments.

(10) Income Taxes

The Company's effective income tax rate was 17% and 22% for the quarters ended June 28, 2014 and June 29, 2013, respectively. The Company's effective tax rate was 16% and 19% for the six months ended June 28, 2014 and June 29, 2013, respectively. The lower effective income tax rates for the quarter and six months ended June 28, 2014 compared to the quarter

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

and six months ended June 29, 2013 were primarily due to a lower proportion of earnings attributed to domestic subsidiaries, which are taxed at rates higher than foreign subsidiaries.

(11) Dividends

As part of the Company's cash deployment strategy, in July 2014 the Company's Board of Directors authorized a regular quarterly dividend of \$0.30 per share to be paid September 3, 2014 to stockholders of record at the close of business on August 12, 2014. In January 2014 and April 2014, the Board of Directors also declared dividends of \$0.30 per share on outstanding common stock which were paid on March 11, 2014 and June 3, 2014, respectively.

Cash paid for dividends was \$29,881 and \$59,731 for the quarter and six months ended June 28, 2014, respectively, and \$19,797 for the quarter and six months ended June 29, 2013.

(12) Business Segment Information

The Company's operations are managed and reported in four operating segments, each of which is a reportable segment for financial reporting purposes: Innerwear, Activewear, Direct to Consumer and International. These segments are organized principally by product category, geographic location and distribution channel. Each segment has its own management that is responsible for the operations of the segment's businesses, but the segments share a common supply chain and media and marketing platforms.

The types of products and services from which each reportable segment derives its revenues are as follows:

- Innerwear sells basic branded products that are replenishment in nature under the product categories of men's underwear, children's underwear, socks, panties, hosiery and intimates, which includes bras and shapewear.
- Activewear sells basic branded products that are primarily seasonal in nature under the product categories of branded printwear and retail
 activewear, as well as licensed logo apparel in collegiate bookstores and other channels.
- Direct to Consumer includes the Company's value-based ("outlet") stores and Internet operations which sell products from the Company's portfolio of leading brands. The Company's Internet operations are supported by its catalogs.
- International primarily relates to the Asia, Latin America, Canada and Australia geographic locations that sell products that span across the Innerwear and Activewear reportable segments.

The Company evaluates the operating performance of its segments based upon segment operating profit, which is defined as operating profit before general corporate expenses and amortization of intangibles. The accounting policies of the segments are consistent with those described in Note 2 to the Company's consolidated financial statements included in its Annual Report on Form 10-K for the year ended December 28, 2013.

		Quarte	r End	ed	Six Months Ended				
	June 28, 2014			June 29, 2013		June 28, 2014		June 29, 2013	
Net sales:									
Innerwear	\$	788,330	\$	687,319	\$	1,359,484	\$	1,184,344	
Activewear		317,814		294,231		612,318		561,417	
Direct to Consumer		104,352		92,633		188,066		172,716	
International		131,556		125,022		241,554		226,189	
Total net sales	\$	1,342,052	\$	1,199,205	\$	2,401,422	\$	2,144,666	

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

		Quarte	r End	led	Six Mon			ıded
		June 28, 2014		June 29, 2013	June 28, 2014			June 29, 2013
Segment operating profit:				_				
Innerwear	\$	181,667	\$	152,702	\$	277,422	\$	242,444
Activewear		45,709		37,120		77,704		58,429
Direct to Consumer		11,848		9,064		11,147		9,196
International		16,060		12,732		24,371		15,014
Total segment operating profit	,	255,284		211,618		390,644		325,083
Items not included in segment operating profit:								
General corporate expenses		(19,892)		(26,874)		(36,931)		(51,825)
Acquisition, integration and other action related charges		(24,045)		_		(66,682)		_
Amortization of intangibles		(4,223)		(3,297)		(8,119)		(6,668)
Total operating profit		207,124		181,447		278,912		266,590
Other expenses		(660)		(751)		(1,095)		(1,215)
Interest expense, net		(21,119)		(25,221)		(42,937)		(50,844)
Income before income tax expense	\$	185,345	\$	155,475	\$	234,880	\$	214,531

The results of Maidenform have been included in the Company's consolidated financial statements since the date of acquisition and are reported as part of the Innerwear, Direct to Consumer and International segments based on geographic location and distribution channel. For the quarter ended June 28, 2014, the Company incurred acquisition, integration and other action related charges of \$24,045, of which \$3,835 is reported in the "Cost of sales" line and \$20,210 is reported in the "Selling, general and administrative expenses" line in the Condensed Consolidated Statement of Income. For the six months ended June 28, 2014, the Company incurred acquisition, integration and other action related charges of \$66,682, of which \$18,662 is reported in the "Cost of sales" line and \$48,020 is reported in the "Selling, general and administrative expenses" line in the Condensed Consolidated Statement of Income.

(13) Consolidating Financial Information

In accordance with the indenture governing the Company's \$1,000,000 6.375% Senior Notes issued on November 9, 2010, as supplemented from time to time, certain of the Company's subsidiaries have guaranteed the Company's obligations under the 6.375% Senior Notes. The following presents the condensed consolidating financial information separately for:

- (i) Parent Company, the issuer of the guaranteed obligations. Parent Company includes Hanesbrands Inc. and its 100% owned operating divisions which are not legal entities, and excludes its subsidiaries which are legal entities;
 - (ii) Guarantor subsidiaries, on a combined basis, as specified in the Indentures;
 - (iii) Non-guarantor subsidiaries, on a combined basis;
- (iv) Consolidating entries and eliminations representing adjustments to (a) eliminate intercompany transactions between or among Parent Company, the guarantor subsidiaries and the non-guarantor subsidiaries, (b) eliminate intercompany profit in inventory, (c) eliminate the investments in the Company's subsidiaries and (d) record consolidating entries; and
 - (v) The Company, on a consolidated basis.

The 6.375% Senior Notes are fully and unconditionally guaranteed on a joint and several basis by each guarantor subsidiary, each of which is 100% owned, directly or indirectly, by Hanesbrands Inc. A guarantor subsidiary's guarantee can be released in certain customary circumstances. Each entity in the consolidating financial information follows the same accounting policies as described in the consolidated financial statements, except for the use by the Parent Company and guarantor subsidiaries of the equity method of accounting to reflect ownership interests in subsidiaries which are eliminated upon consolidation.

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

Condensed Consolidating Statement of Comprehensive Income Quarter Ended June 28, 2014

	Quarter Ended June 28, 2014									
		Parent Company		Guarantor Subsidiaries	Non-Guarantor Subsidiaries			Consolidating Entries and Eliminations		Consolidated
Net sales	\$	1,186,489	\$	190,946	\$	607,973	\$	(643,356)	\$	1,342,052
Cost of sales		897,376		90,178		458,668		(608,524)		837,698
Gross profit		289,113		100,768		149,305		(34,832)		504,354
Selling, general and administrative expenses		215,474		53,922		29,619		(1,785)		297,230
Operating profit		73,639		46,846		119,686		(33,047)		207,124
Equity in earnings of subsidiaries		120,322		93,613		_		(213,935)		_
Other expenses		660		_		_		_		660
Interest expense, net		19,058		(88)		1,979		170		21,119
Income before income tax expense		174,243		140,547		117,707		(247,152)		185,345
Income tax expense		19,665		8,070		3,032		_		30,767
Net income	\$	154,578	\$	132,477	\$	114,675	\$	(247,152)	\$	154,578
Comprehensive income	\$	158,546	\$	132,477	\$	117,159	\$	(249,636)	\$	158,546

Condensed Consolidating Statement of Comprehensive Income Quarter Ended June 29, 2013

	Quarter Ended June 25, 2015									
		Parent Company		Guarantor Subsidiaries		Non-Guarantor Subsidiaries		Consolidating Entries and Eliminations		Consolidated
Net sales	\$	1,054,318	\$	168,376	\$	603,380	\$	(626,869)	\$	1,199,205
Cost of sales		813,705		78,781		463,089		(591,852)		763,723
Gross profit		240,613		89,595		140,291		(35,017)		435,482
Selling, general and administrative expenses		188,889		35,823		30,651		(1,328)		254,035
Operating profit		51,724		53,772		109,640		(33,689)		181,447
Equity in earnings of subsidiaries		112,006		78,211		_		(190,217)		_
Other expenses		751		_		_		_		751
Interest expense, net		23,756		_		1,465		_		25,221
Income before income tax expense		139,223		131,983		108,175		(223,906)		155,475
Income tax expense		17,637		7,754		8,498		_		33,889
Net income	\$	121,586	\$	124,229	\$	99,677	\$	(223,906)	\$	121,586
Comprehensive income	\$	118,326	\$	124,229	\$	163,602	\$	(287,831)	\$	118,326

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

Condensed Consolidating Statement of Comprehensive Income Six Months Ended June 28, 2014

	Six Months Ended June 28, 2014									
		Parent Company		Guarantor Subsidiaries		Non-Guarantor Subsidiaries		Consolidating Entries and Eliminations		Consolidated
Net sales	\$	2,078,819	\$	410,896	\$	1,168,159	\$	(1,256,452)	\$	2,401,422
Cost of sales		1,618,522		223,659		889,218		(1,191,108)		1,540,291
Gross profit		460,297		187,237		278,941		(65,344)		861,131
Selling, general and administrative expenses		406,179		123,945		56,098		(4,003)		582,219
Operating profit		54,118		63,292		222,843		(61,341)		278,912
Equity in earnings of subsidiaries		205,387		168,473		_		(373,860)		_
Other expenses		1,095		_		_		_		1,095
Interest expense, net		36,942		1,898		4,035		62		42,937
Income before income tax expense		221,468		229,867		218,808		(435,263)		234,880
Income tax expense		25,330		5,756		7,656				38,742
Net income	\$	196,138	\$	224,111	\$	211,152	\$	(435,263)	\$	196,138
Comprehensive income	\$	199,325	\$	224,111	\$	211,371	\$	(435,482)	\$	199,325

Condensed Consolidating Statement of Comprehensive Income Six Months Ended June 29, 2013

	Six Months Ended June 29, 2013										
	Parent Company		Guarantor Subsidiaries	ľ	Non-Guarantor Subsidiaries		Consolidating Entries and Eliminations		Consolidated		
Net sales	\$ 1,915,073	\$	301,082	\$	1,147,681	\$	(1,219,170)	\$	2,144,666		
Cost of sales	1,497,562		142,259		902,076		(1,160,012)		1,381,885		
Gross profit	417,511		158,823		245,605		(59,158)		762,781		
Selling, general and administrative expenses	362,837		74,131		61,748		(2,525)		496,191		
Operating profit	 54,674		84,692		183,857		(56,633)		266,590		
Equity in earnings of subsidiaries	187,866		128,030		_		(315,896)		_		
Other expenses	1,215		_		_		_		1,215		
Interest expense, net	47,909		_		2,935		_		50,844		
Income before income tax expense	 193,416		212,722		180,922		(372,529)		214,531		
Income tax expense	20,451		9,129		11,986		_		41,566		
Net income	\$ 172,965	\$	203,593	\$	168,936	\$	(372,529)	\$	172,965		
Comprehensive income	\$ 171,061	\$	203,593	\$	162,637	\$	(366,230)	\$	171,061		

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

Condensed Consolidating Balance Sheet June 28, 2014

				June 28, 2014		
	 Parent Company	Guarantor Subsidiaries	N	Ion-Guarantor Subsidiaries	Consolidating Entries and Eliminations	Consolidated
Assets						
Cash and cash equivalents	\$ 3,433	\$ 4,898	\$	135,286	\$ _	\$ 143,617
Trade accounts receivable, net	132,718	58,621		589,182	(1,001)	779,520
Inventories	1,007,306	129,194		467,083	(201,968)	1,401,615
Deferred tax assets	179,483	15,373		3,293	_	198,149
Other current assets	36,539	9,126		19,170	_	64,835
Total current assets	1,359,479	 217,212		1,214,014	 (202,969)	2,587,736
Property, net	83,514	 47,540		440,690	 _	571,744
Trademarks and other identifiable intangibles, net	6,261	83,784		277,775	_	367,820
Goodwill	232,882	124,247		269,439	_	626,568
Investments in subsidiaries	3,112,658	1,703,831		_	(4,816,489)	_
Deferred tax assets	140,190	53,317		15,196	_	208,703
Receivables from related entities	4,799,460	4,215,846		1,985,892	(11,001,198)	_
Other noncurrent assets	50,554	373		1,520		52,447
Total assets	\$ 9,784,998	\$ 6,446,150	\$	4,204,526	\$ (16,020,656)	\$ 4,415,018
Liabilities and Stockholders' Equity						
Accounts payable	\$ 361,694	\$ 17,150	\$	184,914	\$ _	\$ 563,758
Accrued liabilities	209,203	37,972		78,952	(349)	325,778
Notes payable	_	_		40,802	_	40,802
Accounts Receivable Securitization Facility	_	_		225,000	_	225,000
Total current liabilities	570,897	 55,122		529,668	 (349)	1,155,338
Long-term debt	 1,523,000	 _		_	 _	1,523,000
Pension and postretirement benefits	211,642	_		8,372	_	220,014
Payables to related entities	5,986,970	3,155,251		1,571,663	(10,713,884)	_
Other noncurrent liabilities	114,061	12,887		11,143	147	138,238
Total liabilities	8,406,570	3,223,260		2,120,846	(10,714,086)	3,036,590
Stockholders' equity	1,378,428	3,222,890		2,083,680	(5,306,570)	1,378,428
Total liabilities and stockholders' equity	\$ 9,784,998	\$ 6,446,150	\$	4,204,526	\$ (16,020,656)	\$ 4,415,018

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

Condensed Consolidating Balance Sheet December 28, 2013

	December 28, 2013										
	Parent Company			Guarantor Subsidiaries	N	Ion-Guarantor Subsidiaries	Consolidating Entries and Eliminations			Consolidated	
Assets											
Cash and cash equivalents	\$	5,695	\$	7,811	\$	102,357	\$	_	\$	115,863	
Trade accounts receivable, net		44,366		69,944		465,662		(1,414)		578,558	
Inventories		825,300		208,250		405,756		(155,975)		1,283,331	
Deferred tax assets		178,732		15,373		3,155		_		197,260	
Other current assets		37,429		14,354		16,871		_		68,654	
Total current assets		1,091,522		315,732		993,801		(157,389)		2,243,666	
Property, net		82,786		50,193		446,904		_		579,883	
Trademarks and other identifiable intangibles, net		8,385		88,716		280,650		_		377,751	
Goodwill		232,882		124,247		269,263		_		626,392	
Investments in subsidiaries		2,881,739		1,535,404		_		(4,417,143)		_	
Deferred tax assets		139,102		53,317		15,007		_		207,426	
Receivables from related entities		4,706,001		4,065,909		1,987,603		(10,759,513)		_	
Other noncurrent assets		52,712		412		1,806		_		54,930	
Total assets	\$	9,195,129	\$	6,233,930	\$	3,995,034	\$	(15,334,045)	\$	4,090,048	
Liabilities and Stockholders' Equity											
Accounts payable	\$	253,494	\$	61,964	\$	150,812	\$	_	\$	466,270	
Accrued liabilities		184,653		63,906		66,497		(30)		315,026	
Notes payable		_		_		36,192		_		36,192	
Accounts Receivable Securitization Facility		_		_		181,790		_		181,790	
Total current liabilities		438,147		125,870		435,291		(30)		999,278	
Long-term debt		1,467,000		_		_		_		1,467,000	
Pension and postretirement benefits		253,299		2,159		8,361		_		263,819	
Payables to related entities		5,699,670		3,114,701		1,673,828		(10,488,199)		_	
Other noncurrent liabilities		106,390		11,318		11,620		_		129,328	
Total liabilities		7,964,506		3,254,048		2,129,100		(10,488,229)		2,859,425	
Stockholders' equity		1,230,623		2,979,882		1,865,934		(4,845,816)		1,230,623	
Total liabilities and stockholders' equity	\$	9,195,129	\$	6,233,930	\$	3,995,034	\$	(15,334,045)	\$	4,090,048	

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

Condensed Consolidating Statement of Cash Flows Six Months Ended June 28, 2014

		SIX	MIOII	iuis Ended June 20, a	2014	•	
	 Parent Company	Guarantor Subsidiaries		Non-Guarantor Subsidiaries	Consolidating Entries and Eliminations		Consolidated
Net cash from operating activities	\$ 210,832	\$ 150,068	\$	18,064	\$	(374,026)	\$ 4,938
Investing activities:							
Purchases of property, plant and equipment	(6,995)	(3,194)		(14,396)		_	(24,585)
Proceeds from sales of assets	_	_		4,918		_	4,918
Net cash from investing activities	 (6,995)	(3,194)		(9,478)		_	(19,667)
Financing activities:							
Borrowings on notes payable	_	_		66,737		_	66,737
Repayments on notes payable	_	_		(61,957)		_	(61,957)
Borrowings on Accounts Receivable Securitization Facility	_	_		115,609		_	115,609
Repayments on Accounts Receivable Securitization Facility	_	_		(72,399)		_	(72,399)
Borrowings on Revolving Loan Facility	1,782,500	_		_		_	1,782,500
Repayments on Revolving Loan Facility	(1,726,500)			_			(1,726,500)
Cash dividends paid	(59,731)	_		_		_	(59,731)
Taxes paid related to net shares settlement of equity awards	(10,342)	_		_		_	(10,342)
Excess tax benefit from stock-based compensation	7,895	_		_		_	7,895
Other	1,146	_		(486)		146	806
Net transactions with related entities	 (201,067)	 (149,787)		(23,026)		373,880	 _
Net cash from financing activities	(206,099)	(149,787)		24,478		374,026	42,618
Effect of changes in foreign exchange rates on cash				(135)			(135)
Change in cash and cash equivalents	(2,262)	(2,913)		32,929			27,754
Cash and cash equivalents at beginning of year	5,695	7,811		102,357		_	115,863
Cash and cash equivalents at end of period	\$ 3,433	\$ 4,898	\$	135,286	\$	_	\$ 143,617

Notes to Condensed Consolidated Financial Statements — (Continued) (dollars and shares in thousands, except per share data) (unaudited)

Condensed Consolidating Statement of Cash Flows Six Months Ended June 29, 2013

			itns Ended June 29, 2		
	Parent Company	Guarantor Subsidiaries	Non-Guarantor Subsidiaries	Consolidating Entries and Eliminations	Consolidated
Net cash from operating activities	\$ 165,172	\$ 105,792	\$ 63,256	\$ (315,894)	\$ 18,326
nvesting activities:			-		
Purchases of property, plant and equipment	(7,729)	(2,037)	(10,024)	_	(19,790)
Proceeds from sales of assets	3,402	26	189	_	3,617
Net cash from investing activities	 (4,327)	(2,011)	(9,835)		(16,173)
Financing activities:			-		
Borrowings on notes payable	_	_	62,954	_	62,954
Repayments on notes payable	_	_	(58,671)	_	(58,671)
Borrowings on Accounts Receivable Securitization Facility	_	_	81,358	_	81,358
Repayments on Accounts Receivable Securitization Facility	_	_	(84,715)	_	(84,715)
Borrowings on Revolving Loan Facility	1,970,000	_	_	_	1,970,000
Repayments on Revolving Loan Facility	(1,913,000)		_		(1,913,000)
Cash dividends paid	(19,797)	_	_	_	(19,797)
Proceeds from stock options exercised	5,279	_	_	_	5,279
Taxes paid related to net shares settlement of equity awards	(20,004)	_	_	_	(20,004)
Excess tax benefit from stock-based compensation	14,892	_	_	_	14,892
Other	536	_	(274)	(3)	259
Net transactions with related entities	(198,788)	(102,206)	(14,903)	315,897	_
Net cash from financing activities	 (160,882)	(102,206)	(14,251)	315,894	38,555
Effect of changes in foreign exchange rates on cash			(1,199)		(1,199)
Change in cash and cash equivalents	 (37)	1,575	37,971		39,509
Cash and cash equivalents at beginning of year	5,617	1,919	35,260	_	42,796
Cash and cash equivalents at end of period	\$ 5,580	\$ 3,494	\$ 73,231	\$ _	\$ 82,305

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

This management's discussion and analysis of financial condition and results of operations, or MD&A, contains forward-looking statements that involve risks and uncertainties. Please see "Forward-Looking Statements" in this Quarterly Report on Form 10-Q for a discussion of the uncertainties, risks and assumptions associated with these statements. This discussion should be read in conjunction with our historical financial statements and related notes thereto and the other disclosures contained elsewhere in this Quarterly Report on Form 10-Q. The unaudited consolidated financial statements and notes included herein should be read in conjunction with our audited consolidated financial statements and notes for the year ended December 28, 2013, which were included in our Annual Report on Form 10-K filed with the SEC. The results of operations for the periods reflected herein are not necessarily indicative of results that may be expected for future periods, and our actual results may differ materially from those discussed in the forward-looking statements as a result of various factors, including but not limited to those included elsewhere in this Quarterly Report on Form 10-Q and those included in the "Risk Factors" section and elsewhere in our Annual Report on Form 10-K for the year ended December 28, 2013.

Overview

We are a consumer goods company with a portfolio of leading apparel brands, including *Hanes*, *Champion*, *Bali*, *Playtex*, *Maidenform*, *JMS/Just My Size*, *L'eggs*, *Flexees*, *barely there*, *Wonderbra*, *Gear for Sports*, *Lilyette*, *Zorba*, *Rinbros* and *Sol y Oro*. We design, manufacture, source and sell a broad range of basic apparel such as T-shirts, bras, panties, men's underwear, children's underwear, activewear, socks and hosiery.

Our operations are managed and reported in four operating segments, each of which is a reportable segment for financial reporting purposes: Innerwear, Activewear, Direct to Consumer and International. These segments are organized principally by product category, geographic location and distribution channel. Each segment has its own management that is responsible for the operations of the segment's businesses, but the segments share a common supply chain and media and marketing platforms.

Highlights from the Second Quarter and Six Months Ended June 28, 2014

Key financial highlights during the quarter and six months are as follows:

- Total net sales in the second quarter of 2014 were \$1.3 billion, compared with \$1.2 billion in the same period of 2013, representing a 12% increase. Total net sales in the first six months of 2014 were \$2.4 billion, compared with \$2.1 billion in the same period of 2013, representing a 12% increase.
- Operating profit increased 14% to \$207 million in the second quarter of 2014, compared with \$181 million in the same period of 2013. As a percentage of sales, operating profit was 15.4% in the second quarter of 2014 compared to 15.1% in the same period of 2013. Operating profit was \$279 million in the first six months of 2014, compared with \$267 million in the same period of 2013, representing a 5% increase. As a percentage of sales, operating profit was 11.6% in the second quarter of 2014 compared to 12.4% in the same period of 2013.
- Diluted earnings per share increased 27% to \$1.51 in the second quarter of 2014, compared with diluted earnings per share of \$1.19 in the same period of 2013. Diluted earnings per share was \$1.92 in the first six months of 2014, compared with diluted earnings per share of \$1.70 in the same period of 2013, representing a 13% increase.

Outlook

For the full year 2014, we expect net sales of approximately \$5.075 billion, including approximately \$500 million contributed by Maidenform Brands, Inc. ("Maidenform").

Interest and other related expense is expected to be approximately \$85 million for the full year, including approximately \$10 million from higher debt balances associated with the Maidenform acquisition.

We expect our full year tax rate to be in the low teens with higher rates in the first half of the year.

We expect cash flow from operations to be \$500 million to \$600 million for the full year. We typically use cash for the first half of the year and generate most of our cash flow in the second half of the year. We expect our cash deployment strategy in the future will include a mix of dividends, bolt-on acquisitions and share repurchases. For example, as part of our cash deployment strategy, we have authorized regular dividends during 2014 of \$0.90 per share and have paid, as of June 28, 2014, \$0.60 per share.

On June 25, 2014, we announced our intention to acquire DBA Lux Holding S.A. ("DBA") from SLB Brands Holdings, Ltd and certain individual DBA shareholders, provided consultation with European and French works councils representing DBA employees is completed and customary closing conditions are met. DBA is a leading marketer of intimate apparel,

Table of Contents

hosiery and underwear in Europe. The all-cash transaction offer values DBA at €400,000 on an enterprise basis (approximately \$550,000 at current exchange rates). We intend to finance the transaction with cash on hand and third-party borrowings. The acquisition could close as soon as the third quarter of 2014. There can be no assurance that the proposed acquisition will be completed, or if it is completed, that the expected benefits of the transaction will be realized.

Seasonality and Other Factors

Our operating results are subject to some variability due to seasonality and other factors. Generally, our diverse range of product offerings helps mitigate the impact of seasonal changes in demand for certain items. We generally have higher sales during the back-to-school and holiday shopping seasons and during periods of cooler weather, which benefits certain product categories such as fleece. Sales levels in any period are also impacted by customers' decisions to increase or decrease their inventory levels in response to anticipated consumer demand. Our customers may cancel orders, change delivery schedules or change the mix of products ordered with minimal notice to us. Media, advertising and promotion expenses may vary from period to period during a fiscal year depending on the timing of our advertising campaigns for retail selling seasons and product introductions.

Although the majority of our products are replenishment in nature and tend to be purchased by consumers on a planned, rather than on an impulse, basis, our sales are impacted by discretionary spending by consumers. Discretionary spending is affected by many factors, including, among others, general business conditions, interest rates, inflation, consumer debt levels, the availability of consumer credit, taxation, gasoline prices, weather, unemployment trends and other matters that influence consumer confidence and spending. Many of these factors are outside our control. Consumers' purchases of discretionary items, including our products, could decline during periods when disposable income is lower, when prices increase in response to rising costs, or in periods of actual or perceived unfavorable economic conditions. These consumers may choose to purchase fewer of our products or to purchase lower-priced products of our competitors in response to higher prices for our products, or may choose not to purchase our products at prices that reflect our price increases that become effective from time to time.

Changes in product sales mix can impact our gross profit as the percentage of our sales attributable to higher margin products, such as intimate apparel and men's underwear, and lower margin products, such as activewear, fluctuate from time to time. In addition, sales attributable to higher and lower margin products within the same product category fluctuate from time to time. Our customers may change the mix of products ordered with minimal notice to us, which makes trends in product sales mix difficult to predict. However, certain changes in product sales mix are seasonal in nature, as sales of socks, hosiery and fleece products generally have higher sales during the last two quarters (July to December) of each fiscal year as a result of cooler weather, back-to-school shopping and holidays, while other changes in product mix may be attributable to customers' preferences and discretionary spending.

Condensed Consolidated Results of Operations — Second Quarter Ended June 28, 2014 Compared with Second Quarter Ended June 29, 2013

	Quarte	r Enc	led		
	June 28, 2014		June 29, 2013	Higher (Lower)	Percent Change
			(dollars i	n thousands)	
Net sales	\$ 1,342,052	\$	1,199,205	\$ 142,847	11.9 %
Cost of sales	837,698		763,723	73,975	9.7
Gross profit	 504,354		435,482	68,872	15.8
Selling, general and administrative expenses	297,230		254,035	43,195	17.0
Operating profit	 207,124		181,447	25,677	14.2
Other expenses	660		751	(91)	(12.1)
Interest expense, net	21,119		25,221	(4,102)	(16.3)
Income before income tax expense	 185,345		155,475	29,870	19.2
Income tax expense	30,767		33,889	(3,122)	(9.2)
Net income	\$ 154,578	\$	121,586	\$ 32,992	27.1 %

Net Sales

Net sales increased 12% during the second quarter primarily due to the acquisition of Maidenform in October 2013, which added an incremental \$141 million of net sales in the second quarter of 2014. Net sales were 8% higher in our Activewear segment due to higher sales volume and net space gains at retailers. Offsetting the higher net sales were unfavorable foreign currency exchange rates and lower sales volume in our Innerwear segment. Excluding the impact of

Table of Contents

unfavorable foreign currency exchange rates, consolidated net sales and International segment net sales increased 13% and 11%, respectively.

Gross Profit

Our gross profit was higher for the second quarter of 2014 as compared to the same period of 2013 as we continue to improve our gross profit across all segments. The increase in gross profit was attributable to our Innovate-to-Elevate strategy, which combines our brand power, our innovation platforms and our low cost supply chain to drive margin expansion by increasing our price per unit and reducing our cost per unit through supply chain efficiencies. Included with gross profit in the second quarter of 2014 are charges of \$4 million related to Maidenform acquisition, integration and other action related costs

Selling, General and Administrative Expenses

As a percentage of net sales, our selling, general and administrative expenses were 22.1% in the second quarter of 2014 compared to 21.2% in the second quarter of 2013. The higher selling, general and administrative expenses were attributable to charges of \$20 million related to Maidenform acquisition, integration and other action related costs. Additionally, we incurred higher distribution costs due to increased sales volume in the second quarter of 2014 compared to the second quarter of 2013.

Other Highlights

Interest Expense - Interest expense was lower by \$4 million in the second quarter of 2014 compared to the second quarter of 2013 primarily due to the redemption of the 8% Senior Notes in the fourth quarter of 2013 and a lower weighted average interest rate. Our weighted average interest rate on our outstanding debt was 4.02% during the second quarter of 2014, compared to 5.27% in the second quarter of 2013.

Income Tax Expense – Our effective income tax rate was 17% and 22% for the second quarter of 2014 and the second quarter of 2013, respectively. The lower effective income tax rate was primarily attributable to a lower proportion of earnings attributed to domestic subsidiaries, which are taxed at rates higher than foreign subsidiaries.

Operating Results by Business Segment — Second Quarter Ended June 28, 2014 Compared with Second Quarter Ended June 29, 2013

	Net Sales					Operati	Profit		
		Quarte	er End	ed	Quarter I			Ended	
		June 28, 2014		June 29, 2013		June 28, 2014		June 29, 2013	
				(dollars in	thousa	ands)			
Innerwear	\$	788,330	\$	687,319	\$	181,667	\$	152,702	
Activewear		317,814		294,231		45,709		37,120	
Direct to Consumer		104,352		92,633		11,848		9,064	
International		131,556		125,022		16,060		12,732	
Corporate		_		_		(48,160)		(30,171)	
Total	\$	1,342,052	\$	1,199,205	\$	207,124	\$	181,447	

Innerwear

	Quarte	er Ended	i				
	 June 28, 2014		June 29, 2013	•	Higher (Lower)	Percent Change	
			(dollars i	n thousa	nds)		
Net sales	\$ 788,330	\$	687,319	\$	101,011	14	.7%
Segment operating profit	181,667		152,702		28,965	19	.0

Innerwear net sales were \$101 million higher in the second quarter of 2014 compared to the same period in 2013. The higher net sales were driven primarily by incremental sales of Maidenform products. Our intimate apparel category had higher sales due to space gains in our bra category. Offsetting the improvement was the overall sluggish retail environment, specifically in our basics and hosiery categories, resulting in lower sales volume.

Our Innovate-to-Elevate strategy continues to positively impact our Innerwear segment margins as we are able to increase our price per unit with product innovations and reduce our cost per unit through supply chain efficiencies. Offsetting the improvement was higher distribution costs.

Activewear

	Quarte	er Ende	d			
	June 28, 2014		June 29, 2013	Higher (Lower)		Percent Change
			(dollars i	n thous	ands)	_
Net sales	\$ 317,814	\$	294,231	\$	23,583	8.0%
Segment operating profit	45,709		37,120		8,589	23.1

The higher net sales in our Activewear segment were primarily attributable to higher sales volume in our branded printwear category and space gains for our *Champion* products in our retail channel. *Champion* continues to benefit from innovative platforms resulting in higher sales volume and increased channel penetration. Additionally, net sales in our Gear for Sports licensed apparel increased as a result of space gains.

Our Innovate-to-Elevate strategy continues to positively impact our Activewear segment margins as we are able to increase our price per unit with product innovations and reduce our cost per unit through supply chain efficiencies.

Direct to Consumer

	Quarter Ended					
	 June 28, 2014		June 29, 2013		Higher (Lower)	Percent Change
			(dollars i	n thous	ands)	
Net sales	\$ 104,352	\$	92,633	\$	11,719	12.7%
Segment operating profit	11,848		9,064		2,784	30.7

Direct to Consumer segment net sales were higher due to the addition of Maidenform sales. Comparable store sales were 4% lower in the second quarter of 2014 compared to the same period of 2013 as a result of the sluggish retail environment.

Direct to Consumer segment operating margin increased primarily due to higher sales volume.

International

	Quarte	er Ende	d			
	June 28, 2014		June 29, 2013		Higher (Lower)	Percent Change
			(dollars i	n thous	ands)	
Net sales	\$ 131,556	\$	125,022	\$	6,534	5.2%
Segment operating profit	16,060		12,732		3,328	26.1

Sales in the International segment were higher due to the addition of Maidenform sales, partially offset by an unfavorable impact of foreign exchange rates. Excluding the unfavorable impact of foreign exchange rates, International segment net sales were 11% higher.

International segment operating margin increased primarily due to higher sales volume, partially offset by foreign currency exchange rates.

Corporate

Corporate expenses were higher in the second quarter of 2014 compared to the second quarter of 2013 primarily due to acquisition, integration and other action related charges of \$24 million.

Condensed Consolidated Results of Operations — Six Months Ended June 28, 2014 Compared with Six Months Ended June 29, 2013

	Six Months Ended						
	June 28, 2014		June 29, 2013		Higher (Lower)		Percent Change
	(dollars in thou					sands)	
Net sales	\$	2,401,422	\$	2,144,666	\$	256,756	12.0 %
Cost of sales		1,540,291		1,381,885		158,406	11.5
Gross profit		861,131		762,781		98,350	12.9
Selling, general and administrative expenses		582,219		496,191		86,028	17.3
Operating profit		278,912		266,590		12,322	4.6
Other expenses		1,095		1,215		(120)	(9.9)
Interest expense, net		42,937		50,844		(7,907)	(15.6)
Income before income tax expense		234,880		214,531		20,349	9.5
Income tax expense		38,742		41,566		(2,824)	(6.8)
Net income	\$	196,138	\$	172,965	\$	23,173	13.4 %

Net Sales

Net sales increased 12% in the six months of 2014 compared to the same period of 2013 as a result of the acquisition of Maidenform in October 2013, which added an incremental \$266 million of net sales during the six months of 2014. Net sales were 9% higher in our Activewear segment as a result of our Innovate-to-Elevate strategy, which helped drive core-product and new-product success. Offsetting these higher net sales were lower sales volume in our Innerwear segment and unfavorable foreign currency exchange rates. Excluding the impact of unfavorable foreign currency exchange rates, consolidated net sales and International segment net sales increased 13% and 15%, respectively.

Gross Profit

Our gross profit was higher for the first six months of 2014 as compared to the same period in 2013 as we continue to maintain strong profit margins across all segments. Our Innovate-to-Elevate strategy continues to help drive profitable results as we combine our brand and supply chain strengths with product innovation. Included with gross profit in the six months of 2014 are charges of \$19 million related to Maidenform acquisition, integration and other action related costs.

Selling, General and Administrative Expenses

As a percentage of net sales, our selling, general and administrative expenses were 24.2% in the six months of 2014 compared to 23.1% in the same period of 2013. The higher selling, general and administrative expenses were primarily attributable to charges of \$48 million related to Maidenform acquisition, integration and other action related costs and higher distribution costs.

Other Highlights

Interest Expense – lower by \$8 million in the six months of 2014 compared to the same period of 2013 primarily due to the redemption of the 8% Senior Notes in the fourth quarter of 2013 and a lower weighted average interest rate. Our weighted average interest rate on our outstanding debt was 4.07% during the six months of 2014 compared to 5.37% in the same period of 2013.

Income Tax Expense – our effective income tax rate was 16% and 19% for the six months of 2014 and 2013, respectively. The lower effective income tax rate was primarily due to a lower proportion of earnings attributed to domestic subsidiaries, which are taxed at rates higher than foreign subsidiaries. The lower effective income tax rate was partially offset by the benefit of approximately \$6 million in the first quarter of 2013 related to the retroactive application of the American Taxpayer Relief Act of 2012 that was signed into law in January 2013.

Operating Results by Business Segment — Six Months Ended June 28, 2014 Compared with Six Months Ended June 29, 2013

	Net Sales				Operating Profit			
		Six Mon	ths	Ended		Six Mon	ths Ended	
	June 28, 2014			June 29, 2013		June 28, 2014		June 29, 2013
				(dollars in	thous	ands)		
Innerwear	\$	1,359,484	5	\$ 1,184,344	\$	277,422	\$	242,444
Activewear		612,318		561,417		77,704		58,429
Direct to Consumer		188,066		172,716		11,147		9,196
International		241,554		226,189		24,371		15,014
Corporate		_		_		(111,732)		(58,493)
Total net sales	\$	2,401,422	9	\$ 2,144,666	\$	278,912	\$	266,590

Innerwear

	Six Mon	ths En	ded			
	June 28, 2014		June 29, 2013	•	Higher (Lower)	Percent Change
			(dollars i	n thous	ands)	
Net sales	\$ 1,359,484	\$	1,184,344	\$	175,140	14.8%
Segment operating profit	277,422		242,444		34,978	14.4

The higher net sales in our Innerwear segment were driven primarily by incremental sales of Maidenform products. A sluggish retail environment led to lower sales volume in our intimate apparel, basics and hosiery product categories.

Our Innovate-to-Elevate strategy continues to positively impact our Innerwear segment margins as we are able to increase our price per unit with product innovations and reduce our cost per unit through supply chain efficiencies. Offsetting the improvement were higher planned media spending and higher distribution costs.

Activewear

	 Six Mon	ths End	led			
	 June 28, 2014	June 29, 2013		Higher (Lower)		Percent Change
			(dollars i	n thous	ands)	
Net sales	\$ 612,318	\$	561,417	\$	50,901	9.1%
Segment operating profit	77,704		58,429		19,275	33.0

The higher net sales in our Activewear segment is primarily attributable to net space gains and higher sales volume for *Champion* products in our retail channel. Additionally, we had higher net sales of our branded printwear category and Gear for Sports licensed apparel as a result of higher sales volume and space gains, respectively.

Activewear segment operating margin improved due to higher sales volume across all product categories within the segment. Our Innovate-to-Elevate strategy continues to positively impact our Activewear segment margins as we are able to increase our price per unit with product innovations and reduce our cost per unit through supply chain efficiencies. The margin improvement was partially offset by higher planned media spending in the first six months of 2014 as compared to the same period of 2013.

Direct to Consumer

	Six Months Ended						
	 June 28, 2014		June 29, 2013		Higher (Lower)	Percent Change	
	(dollars in thousands)						
Net sales	\$ 188,066	\$	172,716	\$	15,350	8.9%	
Segment operating profit	11,147		9,196		1,951	21.2	

Direct to Consumer segment net sales were higher due to the addition of Maidenform sales. Comparable store sales were 4% lower in the first six months of 2014 compared to the same period of 2013 resulting from the unusually high weather-related temporary store closures occurring in the first quarter of 2014 and overall sluggish retail environment.

Direct to Consumer segment operating margin improved due to higher sales volume.

International

	Six Months Ended						
	June 28, 2014		June 29, 2013	•'	Higher (Lower)	Percent Change	
	(dollars in thousands)						
Net sales	\$ 241,554	\$	226,189	\$	15,365	6.8%	
Segment operating profit	24,371		15,014		9,357	62.3	

Overall net sales in the International segment were higher primarily due to the addition of Maidenform sales, partially offset by an unfavorable impact of foreign exchange rates. Excluding the unfavorable impact of foreign exchange rates, International segment net sales were higher by 15%.

International segment operating margin increased due to higher sales volume, partially offset by the unfavorable impact related to foreign currency exchange rates.

General Corporate Expenses

General corporate expenses were higher in the six months of 2014 compared to the same period of 2013 primarily due to acquisition, integration and other action related charges of \$67 million.

Liquidity and Capital Resources

Trends and Uncertainties Affecting Liquidity

Our primary sources of liquidity are cash generated by operations and availability under the \$1.1 billion revolving credit facility (the "Revolving Loan Facility") under our senior secured credit facility (the "Senior Secured Credit Facility"), our accounts receivable securitization facility (the "Accounts Receivable Securitization Facility") and our international loan facilities.

At June 28, 2014, we had \$566 million of borrowing availability under our Revolving Loan Facility (after taking into account outstanding letters of credit), \$88 million of borrowing availability under our international loan facilities, \$144 million in cash and cash equivalents and no borrowing availability under our Accounts Receivable Securitization Facility. We currently believe that our existing cash balances and cash generated by operations, together with our available credit capacity, will enable us to comply with the terms of our indebtedness and meet foreseeable liquidity requirements.

We typically use cash during the first half of the year and generate most of our cash flow in the second half of the year. We expect our cash deployment strategy in the future will include a mix of dividends, bolt-on acquisitions and share repurchases.

On June 25, 2014, we announced our intention to acquire DBA in a transaction valued at €400 million (approximately \$550 million at current exchange rates). We intend to finance the transaction, which could close as soon as the third quarter of 2014, with cash on hand and third-party borrowings, including a €363 million institutional term loan.

Dividends

As part of our cash deployment strategy, in January 2014 and April 2014 our Board of Directors declared regular quarterly dividends of \$0.30 per share which were paid in March 2014 and June 2014, respectively. In July 2014, our Board of Directors authorized a regular quarterly dividend of \$0.30 per share to be paid September 3, 2014 to stockholders of record at the close of business on August 12, 2014.

Cash Requirements for Our Business

We rely on our cash flows generated from operations and the borrowing capacity under our Revolving Loan Facility, Accounts Receivable Securitization Facility and international loan facilities to meet the cash requirements of our business. The primary cash requirements of our business are payments to vendors in the normal course of business, capital expenditures, maturities of debt and related interest payments, contributions to our pension plans, repurchases of our stock and regular quarterly dividend payments. We believe we have sufficient cash and available borrowings for our foreseeable liquidity needs.

There have been no significant changes in the cash requirements for our business from those described in our Annual Report on Form 10-K for the year ended December 28, 2013.

Sources and Uses of Our Cash

The information presented below regarding the sources and uses of our cash flows for the six months ended June 28, 2014 and June 29, 2013 was derived from our condensed consolidated financial statements.

	Six Months Ended				
		June 28, 2014		June 29, 2013	
		(dollars in thousands)			
Operating activities	\$	4,938	\$	18,326	
Investing activities		(19,667)		(16,173)	
Financing activities		42,618		38,555	
Effect of changes in foreign currency exchange rates on cash		(135)		(1,199)	
Change in cash and cash equivalents		27,754		39,509	
Cash and cash equivalents at beginning of year		115,863		42,796	
Cash and cash equivalents at end of period	\$	143,617	\$	82,305	

Our overall liquidity is primarily driven by our strong cash flow provided by operating activities, which is dependent on net income, as well as changes in our working capital. As compared to prior year, the lower net cash from operating activities is due to changes in working capital, specifically related to inventory, accounts receivable, accounts payable and accrued liabilities. We typically build inventory unit levels in the first half of the year to support the back-to-school shopping season.

The higher net cash used in investing activities resulted from higher net capital expenditures. The higher net cash from financing activities was primarily the result of higher net borrowings on our loan facilities, partially offset by incremental cash dividends paid during 2014 of \$40 million.

Financing Arrangements

In March 2014, we amended the Accounts Receivable Securitization Facility. This amendment decreased certain fee rates, revised certain concentration limits and dilution triggers and extended the termination date to March 2015.

As of June 28, 2014, we were in compliance with all financial covenants under our credit facilities. We expect to maintain compliance with our covenants for the foreseeable future, however economic conditions or the occurrence of events discussed under "Risk Factors" in our Annual Report on Form 10-K for the year ended December 28, 2013 or other SEC filings could cause noncompliance.

In May 2014, Moody's Investors Service ("Moody's") upgraded our corporate credit rating to Ba1 from Ba2 and our probability of default rating to Ba1-PD from Ba2-PD. Moody's also raised the rating on our Revolving Loan Facility to Baa2 from Ba3, and the 6.375% Senior Notes to Ba2 from Ba3. Moody's indicated the upgrade reflects our improved operating margins from a combination of lower overall production costs and modest increase in revenues.

In July 2014, Moody's affirmed our corporate credit rating at Ba1 and our probability of default rating at Ba1-PD. Moody's adjusted the rating on our Revolving Loan Facility to Baa3 and assigned a Baa3 rating to the anticipated Euro term loan due 2021, which will be primarily used to fund a portion of the DB Apparel acquisition. The new Euro term loan will have the same guarantee and security structure as the Revolving Loan Facility and therefore, Moody's indicated that it is unable to warrant a rating differential between the Revolving Loan Facility and the Euro term loan. Additionally, in July 2014, Standard & Poor's Ratings Services ("Standard & Poor's") assigned a credit rating of BBB- to the anticipated Euro term loan. The corporate credit and unsecured debt ratings of BB remain unchanged.

There have been no other significant changes in the financing arrangements from those described in our Annual Report on Form 10-K for the year ended December 28, 2013.

Off-Balance Sheet Arrangements

We do not have any off-balance sheet arrangements within the meaning of Item 303(a)(4) of SEC Regulation S-K.

Critical Accounting Policies and Estimates

We have chosen accounting policies that we believe are appropriate to accurately and fairly report our operating results and financial condition in conformity with accounting principles generally accepted in the United States. We apply these accounting policies in a consistent manner. Our significant accounting policies are discussed in Note 2, "Summary of

Table of Contents

Significant Accounting Policies," to our financial statements included in our Annual Report on Form 10-K for the year ended December 28, 2013.

The application of critical accounting policies requires that we make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues and expenses and related disclosures. These estimates and assumptions are based on historical and other factors believed to be reasonable under the circumstances. We evaluate these estimates and assumptions on an ongoing basis and may retain outside consultants to assist in our evaluation. If actual results ultimately differ from previous estimates, the revisions are included in results of operations in the period in which the actual amounts become known. The critical accounting policies that involve the most significant management judgments and estimates used in preparation of our financial statements, or are the most sensitive to change from outside factors, are discussed in Management's Discussion and Analysis of Financial Condition and Results of Operations in our Annual Report on Form 10-K for the year ended December 28, 2013. There have been no material changes in these policies from those described in our Annual Report on Form 10-K for the year ended December 28, 2013.

Recently Issued Accounting Pronouncements

Discontinued Operations

In April 2014, the FASB issued new accounting rules related to updating the criteria for reporting discontinued operations and enhancing related disclosures requirements. The new rules are effective for us in the first quarter of 2015. We do not expect the adoption of the new accounting rules to have a material impact on our financial condition, results of operations or cash flows.

Revenue from Contracts with Customers

In May 2014, the FASB issued new accounting rules related to revenue recognition for contracts with customers requiring revenue recognition based on the transfer of promised goods or services to customers in an amount that reflects consideration we expect to be entitled in exchange for goods or services. The new rules supercede prior revenue recognition requirements and most industry-specific accounting guidance. The new rules will be effective for us in the first quarter of 2017 with retrospective application required. We do not expect the adoption of the new accounting rules to have a material impact on our financial condition, results of operations or cash flows.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

There have been no significant changes in our market risk exposures from those described in Item 7A of our Annual Report on Form 10-K for the year ended December 28, 2013.

Item 4. Controls and Procedures

As required by Exchange Act Rule 13a-15(b), our management, including our Chief Executive Officer and Chief Financial Officer, conducted an evaluation of the effectiveness of our disclosure controls and procedures, as defined in Exchange Act Rule 13a-15(e), as of the end of the period covered by this report. Based on that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective.

In connection with the evaluation required by Exchange Act Rule 13a-15(d), our management, including our Chief Executive Officer and Chief Financial Officer, concluded that no changes in our internal control over financial reporting occurred during the period covered by this report that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II

Item 1. Legal Proceedings

Although we are subject to various claims and legal actions that occur from time to time in the ordinary course of our business, we are not party to any pending legal proceedings that we believe could have a material adverse effect on our business, results of operations, financial condition or cash flows.

Item 1A. Risk Factors

The risk factors that affect our business and financial results are discussed in Part I, Item 1A, of our Annual Report on Form 10-K for the fiscal year ended December 28, 2013. There are no material changes to the risk factors previously disclosed, nor have we identified any previously undisclosed risks that could materially adversely affect our business and financial results.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

Item 3. Defaults Upon Senior Securities

None.

Item 4. Mine Safety Disclosures

Not applicable.

Item 5. Other Information

None.

Item 6. Exhibits

The exhibits listed in the accompanying Exhibit Index are filed or furnished as part of this Quarterly Report on Form 10-Q.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

HANESBRANDS INC.

By: /s/ Richard D. Moss

Richard D. Moss Chief Financial Officer (Duly authorized officer and principal financial officer)

Date: July 24, 2014

INDEX TO EXHIBITS

Exhibit <u>Number</u>	<u>Description</u>
2.1	Binding Offer Letter, dated June 25, 2014 by and between Hanesbrands Inc. and SLB Brands Holdings, Ltd. (Certain schedules to Exhibit A have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The Registrant agrees to furnish a supplemental copy of any omitted schedule to the SEC upon request).
3.1	Articles of Amendment and Restatement of Hanesbrands Inc. (incorporated by reference from Exhibit 3.1 to the Registrant's Current Report on Form 8-K filed with the Securities and Exchange Commission on September 5, 2006).
3.2	Articles Supplementary (Junior Participating Preferred Stock, Series A) (incorporated by reference from Exhibit 3.2 to the Registrant's Current Report on Form 8-K filed with the Securities and Exchange Commission on September 5, 2006).
3.3	Amended and Restated Bylaws of Hanesbrands Inc. (incorporated by reference from Exhibit 3.1 to the Registrant's Current Report on Form 8-K filed with the Securities and Exchange Commission on December 15, 2008).
31.1	Certification of Richard A. Noll, Chief Executive Officer.
31.2	Certification of Richard D. Moss, Chief Financial Officer.
32.1	Section 1350 Certification of Richard A. Noll, Chief Executive Officer.
32.2	Section 1350 Certification of Richard D. Moss, Chief Financial Officer.
101.INS XBRL	Instance Document
101.SCH XBRL	Taxonomy Extension Schema Document
101.CAL XBRL	Taxonomy Extension Calculation Linkbase Document
101.LAB XBRL	Taxonomy Extension Label Linkbase Document
101.PRE XBRL	Taxonomy Extension Presentation Linkbase Document
101.DEF XBRL	Taxonomy Extension Definition Linkbase Document

MFB INTERNATIONAL HOLDINGS S.À R.L.

33, Rue du Puits Romain, L-8070 Bertrange, Grand Duchy of Luxembourg

HANESBRANDS INC.

1000 East Hanes Mill Road, Winston-Salem, NC 27105

25 June 2014

SLB Brands Holdings, Ltd

c/o Maples Corporate Services Limited PO Box 309 Ugland House George Town Grand Cayman KY1-1104 Cayman Islands

Dear Sirs:

Final Binding Offer for the DBA business (the "Offer Letter")

We, MFB International Holdings s.à r.l., refer to our recent discussions relating to the possible acquisition (the "Transaction") by us, a company incorporated under the laws of the Grand-Duchy of Luxembourg (the "Purchaser"), of 100% of the issued share capital of DBA Lux Holding S.A. (the "Company") and the warrants issued by the Company pursuant to a proposed share purchase agreement to be entered into by SLB Brands Holdings, Ltd (the "Institutional Seller"), certain managers of the Group (the "Management Sellers" and together with the Institutional Seller, the "Sellers"), certain Manager holding companies (the "Manager Holding Companies"), the Purchaser and Hanesbrands Inc. (the "Purchaser's Guarantor") (the "Share Purchase Agreement").

Capitalised terms used but not otherwise defined herein shall have the meanings ascribed to them in the Share Purchase Agreement.

All references in the Offer Letter to time are to Paris time.

1. Offer

1.1 The Purchaser hereby makes a final, binding and irrevocable offer to the Institutional Seller to purchase all of the Shares and Warrants free and clear of all Encumbrances together with all rights attached thereto, on the terms and subject to the conditions of the Share Purchase Agreement and this Offer Letter, and the Purchaser's Guarantor makes a final, binding and irrevocable offer to be a guarantor of the Purchaser on the terms and subject to the conditions of the Share Purchase Agreement and this Offer Letter (together, the "Offer"). A copy of the Share Purchase Agreement (including schedules, exhibits)

- and appendixes), executed by the Purchaser and the Purchaser's Guarantor is attached as Exhibit A to this Offer Letter.
- 1.2 The Offer is made to the Institutional Seller and the covenants and warranties contained in or arising under this Offer Letter are given to the Institutional Seller.
- 1.3 The Purchaser and the Purchaser's Guarantor irrevocably and unconditionally undertake to the Institutional Seller in consideration of the obligations of the Institutional Seller hereunder not to withdraw, revoke, modify or qualify the Offer in any respect whatsoever.
- 1.4 The Institutional Seller's acknowledgement of the Offer by countersigning this Offer Letter shall not constitute in any manner whatsoever an undertaking by the Institutional Seller to accept the Offer or to become bound to sell or transfer any Shares and Warrants or to execute the Share Purchase Agreement or any other agreement in respect of the Transaction but shall only constitute an option for the Institutional Seller, exercisable in its sole discretion, to sell all of the Shares and Warrants to the Purchaser on the terms and subject to the conditions of the Share Purchase Agreement.

2. Offer Acceptance

- 2.1 Once the Institutional Seller has delivered a countersigned copy of this Offer Letter to the Purchaser and the Purchaser's Guarantor, the Offer shall remain valid and in effect, and shall be irrevocable, until 11.59 p.m. on the earlier of:
 - (a) subject to paragraph 3, 31 December 2014; and
 - (b) the date that is 10 calendar days after delivery or deemed delivery of the Works Council Opinions (as defined in <u>Schedule 1</u>), as provided for in <u>Schedule 1</u>,
 - (as such date may be extended by agreement of the Institutional Seller, the Purchaser and the Purchaser's Guarantor, the "Offer Expiry Date").
- 2.2 With effect from the day following the Offer Expiry Date, the Offer will automatically lapse and may no longer be accepted, and all provisions of this Offer Letter shall be of no further force and effect (other than paragraphs 7 (to the extent provided therein) and 9 to 13 inclusive).
- 2.3 The Purchaser and the Purchaser's Guarantor acknowledge and agree that the Institutional Seller may accept the Offer by delivering a written notice to the Purchaser in the form of the Acceptance Notice attached to this Offer Letter at Exhibit B (the "Acceptance Notice") duly executed by the Institutional Seller, accompanied by an original copy of the Share Purchase Agreement duly executed by all of the Sellers and Manager Holding Companies, prior to 11:59 p.m. on the Offer Expiry Date. The Institutional Seller hereby undertakes in favour of the Purchaser and the Purchaser's Guarantor that the copy of the Share Purchase Agreement so delivered shall be executed by all of the Management Sellers and Manager Holding Companies and the Institutional Seller shall procure absolutely (se porte fort avec obligation de résultat) that it is so

executed. Acceptance of the Offer shall not be valid in the absence of the Share Purchase Agreement so delivered and executed. The Acceptance Notice together with the Share Purchase Agreement shall be deemed to have been received by the Purchaser:

- (a) if delivered by hand, at the time of delivery to the Purchaser;
- (b) if sent by post, at 9.30 am on the eighth day after (and excluding) the date of posting by the Institutional Seller to the Purchaser; or
- (c) if sent by fax, at the time of transmission by the sender,

at the contact details provided in Clause 14 of the Share Purchase Agreement, provided that if a notice would otherwise be deemed to have been received outside Normal Business Hours, it shall instead be deemed to have been received at the recommencement of such Normal Business Hours.

- 2.4 Upon delivery of the Acceptance Notice together with the Share Purchase Agreement duly executed as aforesaid, the Share Purchase Agreement shall become a valid and binding agreement of the parties thereto in accordance with its terms.
- 2.5 At the date on which the Share Purchase Agreement becomes binding, all rights and obligations of the Institutional Seller, the Purchaser and the Purchaser's Guarantor under this Offer Letter (except in respect of antecedent breaches) shall terminate.

3. Termination

- 3.1 Notwithstanding anything to the contrary in this Offer Letter, this Offer Letter may be terminated at any time prior to the Offer Expiry Date:
 - 3.1.1 by the mutual consent of the Purchaser and the Institutional Seller; or
 - 3.1.2 by the Institutional Seller or the Purchaser if any Governmental Entity shall have issued an order or taken any other action (which order or other action the parties shall use their commercially reasonable efforts to lift) which permanently restrains, enjoins or otherwise prohibits the completion of the Transaction and such order or other action shall have become final and non-appealable.
- 3.2 In the event this Offer Letter is terminated by any party pursuant to the terms of this paragraph 3, the Offer will automatically lapse and may no longer be accepted, and all provisions of this Offer Letter (other than paragraphs 9, 10, 12 and 13) shall be of no further force and effect.

4. Works Councils and Consultation Process

4.1 The defined terms 'Works Councils' and 'Works Council Opinions' shall have the meanings given to them in <u>Schedule 1</u> hereto.

- 4.2 The Institutional Seller shall use its reasonable efforts and work diligently, and shall exercise all of its voting rights as shareholder of the Company to cause the Group to comply with all appropriate consultation procedures in connection with the delivery or deemed delivery of the Works Council Opinions as provided for in Schedule 1 and, accordingly, to convene meetings with the Works Councils as soon as reasonably practicable with a view to obtaining expeditious delivery or deemed delivery of the Works Council Opinions. The Institutional Seller shall keep the Purchaser regularly informed (subject to applicable confidentiality obligations and/or constraints) of the status of matters relating to the delivery or deemed delivery of the Works Council Opinions, including furnishing the Purchaser promptly (and in any event within two Business Days of receipt by the Institutional Seller) with copies of notices or details of other communications received by the Institutional Seller whether orally or in writing as part of the process relating to the delivery or deemed delivery of the Works Council Opinions and shall provide the Purchaser with such other information relating to the delivery or deemed delivery of the Works Council Opinions as the Purchaser may reasonably require from time to time. Without limiting the generality of the foregoing, the Institutional Seller undertakes to inform the Purchaser in writing as soon as possible (and in any event no later than two Business Days) after any of the Works Councils has delivered or is deemed to have delivered a Works Council Opinion in accordance with Schedule 1.
- 4.3 Within three Business Days from the date of this Offer Letter, the Purchaser shall provide the Institutional Seller with a document containing detailed information regarding the Purchaser, including, but not limited to: (i) the Purchaser's background and history, past investments in Europe, structure chart, current assets/investments, specifically in the apparel sector, and management, employment policy and experience, in each case so far as it is relevant to the Transaction; and (ii) the Purchaser's future plans for the Group, and, if applicable, the group to which the Purchaser belongs, which the Institutional Seller may disclose to the Works Councils. If the Institutional Seller, acting reasonably, considers that such information should be disclosed to any other staff representative bodies in any other jurisdiction, it may do so having first consulted with the Purchaser thereon and provided it does not disclose any inside information concerning the Purchaser (as that term is interpreted in the Market Abuse Directive 2003/6/EC).
- 4.4 The Purchaser agrees to cooperate and procure that each other member of the Purchaser's Group cooperates, as may be reasonably required, with the Institutional Seller and the Group in connection with the delivery or deemed delivery of the Works Council Opinions, including by:
 - 4.4.1 providing any documents and information relating to the Purchaser's Group or the Transaction that may be reasonably requested by the Works Councils, as promptly as practicable but to the extent such information is available in any event within five Business Days following receipt by the Purchaser of written notice thereof;
 - 4.4.2 having senior authorised representatives of the Purchaser's Group attend meetings with the Works Councils and any other staff representative body which the Institutional Seller deems reasonably necessary or beneficial to the Transaction, if

so required or requested by the Institutional Seller, as promptly as reasonably practicable but in any event within five Business Days following receipt by the Purchaser of notice thereof;

- 4.4.3 not doing anything (and procuring that no member of the Purchaser's Group does anything) that could reasonably be expected to prejudice a favourable outcome of the process relating to delivery or deemed delivery of the Works Council Opinions, provided that nothing herein shall restrict any of the members of the Purchaser's Group from conducting their business in the ordinary course consistent with past practice; and
- 4.4.4 consulting with the Institutional Seller and considering in good faith any issues and proposals in relation to the Transaction that may be raised as part of the delivery or deemed delivery of the Works Council Opinions by the Works Councils, provided that the Purchaser's obligation in this respect shall be limited to such consultation and consideration in good faith and the Purchaser shall not be obliged to agree to any modification hereto or to the Share Purchase Agreement nor to make any commitment of any kind.

The Purchaser understands and acknowledges the importance of its diligent, professional and timely participation as and when requested by the Institutional Seller in order to complete the process relating to the delivery or deemed delivery of the Works Council Opinions in a timely manner.

5. Merger Control

- 5.1 The Purchaser agrees that it will be primarily responsible for preparing and making any filings required by Russia's antimonopoly Laws in relation to the Transaction as provided in <u>Schedule 2</u> and the Institutional Seller shall, and shall procure that the Group Companies shall, cooperate in preparing and making such filings in the manner set out in <u>Schedule 2</u>.
- 5.2 Without prejudice to its obligations under paragraph 5.1 if, pursuant to any action taken by a member of the Purchaser's Group after the date of this Offer Letter, any decree or order is granted under any applicable competition or antitrust Law that would have the effect of preventing the Completion of the Transaction on the terms contemplated in the Transaction Documents, the Purchaser shall and shall procure that each relevant member of the Purchaser's Group shall agree to any divestments and/or comply with any prescriptions, conditions, or undertakings imposed by the Federal Antimonopoly Service of the Russian Federation (the "FAS"), or other relevant Governmental Entity having jurisdiction in respect of the Transaction who so imposes such prescription, condition or undertaking, as will permit the Completion of the Transaction on the terms contemplated by the Transaction Documents.

6. Warranties

6.1 The Institutional Seller warrants to the Purchaser:

- 6.1.1 in terms of the Warranties set out in Clause 8.1 and Clause 8.3 (subject to the proviso thereto in Clause 8.5) of the Share Purchase Agreement;
- 6.1.2 that this Offer Letter will, when executed, constitute valid, binding and, subject to applicable bankruptcy, insolvency and other similar laws of general application affecting enforcement of creditors' rights generally, enforceable obligations of the Institutional Seller in accordance with its terms;
- 6.1.3 that the Institutional Seller has full power and authority and has taken all corporate action required of it to enter into, deliver and perform its obligations under this Offer Letter;
- 6.1.4 that the execution and delivery of, and the performance by the Institutional Seller of its obligations under this Offer Letter will not:
 - (a) conflict with or result in a breach of any provision of the articles of association or other constitutional documents of the Institutional Seller;
 - (b) conflict with, result in a breach of or constitute a default (with or without the lapse of time and/or the giving of any notice, certificate, declaration or demand) under any agreement or instrument to which the Institutional Seller is a party;
 - (c) conflict with or result in a breach of any Law or regulation, or of any order, injunction, judgment or decree of any court, that applies to the Institutional Seller; or
 - (d) save as set out in this Offer Letter, require the Institutional Seller to obtain any consent or approval of, or give any notice to or make any registration with, any Governmental Entity that has not been unconditionally and irrevocably obtained or made at the date of this Offer Letter (save for any legal or regulatory entitlement to revoke the same other than by reason of any misrepresentation or misstatement).
- 6.2 The Purchaser warrants to the Institutional Seller:
 - 6.2.1 in terms of the warranties set out in Clause 8.7 of the Share Purchase Agreement;
 - 6.2.2 that the Purchaser has full power and authority and has taken all corporate action required of it to enter into, deliver and perform its obligations under this Offer Letter:
 - 6.2.3 that this Offer Letter will, when executed, constitute valid, binding and, subject to applicable bankruptcy, insolvency and other similar laws of general application affecting enforcement of creditors' rights generally, enforceable obligations of the Purchaser in accordance with its terms; and

- 6.2.4 that the execution and delivery of, and the performance by the Purchaser of its obligations under this Offer Letter will not:
 - (a) conflict with or result in a breach of any provision of the articles of association or other constitutional documents of the Purchaser;
 - (b) conflict with, result in a breach of or constitute a default under (with or without the lapse of time and/or the giving of any notice, certificate, declaration or demand) any agreement or instrument to which the Purchaser is a party;
 - (c) conflict with or result in a breach of any Law or regulation, or of any order, injunction, judgment or decree of any court, that applies to the Purchaser; or
 - (d) save as set out in this Offer Letter, require the Purchaser to obtain any consent or approval of, or give any notice to or make any registration with, any Governmental Entity that has not been unconditionally and irrevocably obtained or made at the date of this Offer Letter (save for any legal or regulatory entitlement to revoke the same other than by reason of any misrepresentation or misstatement).

7. Exclusivity

- 7.1 From the date of this Offer Letter until the earliest to occur of: (i) the execution date of the Share Purchase Agreement, (ii) subject always to any extension of the exclusivity period pursuant to paragraph 11.1(a) hereof, the lapse of the Offer following the Offer Expiry Date, and (iii) the termination of the Offer Letter in accordance with paragraph 3, the Institutional Seller undertakes:
 - (a) not to, directly or indirectly: (1) make, initiate, solicit, encourage, discuss or facilitate any inquiry, proposal or offer by any person other than the Purchaser with respect to (x) the transfer of all or part of the Shares or Warrants or (y) any other transaction similar to, or having the same purpose as the Transaction or which, if completed, would frustrate the Transaction (any such inquiry, proposal or offer, a "Competing Proposal"); (2) enter into, continue, or participate in any discussions or negotiations with any person with respect to any Competing Proposal; (3) make any offer to, or enter into any agreement, binding arrangement or understanding with, any person in respect of any Competing Proposal; or (4) furnish any information to any person in connection with any Competing Proposal;
 - to promptly cease any existing activities, discussions or negotiations with any person, other than the Purchaser with respect to any Competing Proposal; and
 - (c) if it receives, during such period, any inquiry, proposal, offer, bid or invitation for discussion regarding a Competing Proposal, it shall respond by indicating its

current lack of interest for such Competing Proposal and shall not take any further action in relation to such Competing Proposal.

- 7.2 Without prejudice to any reimbursement of costs and expenses that may be due by the Institutional Seller to the Purchaser pursuant to paragraph 11.1(b)11.1(a), if during the period of exclusivity, as extended pursuant to paragraph 11.1(a):
 - (a) despite the delivery or deemed delivery of the Works Council Opinions in accordance with <u>Schedule 1</u> on or before the Offer Expiry Date, the Institutional Seller does not validly accept the Offer in accordance with paragraph 2.3; and
 - (b) the Institutional Seller breaches its obligations under paragraph 7.1,

the Institutional Seller shall promptly pay to the Purchaser an aggregate lump sum amount of damages corresponding to EUR 15,000,000. The Institutional Seller explicitly acknowledges that the provisions of this paragraph 7.2 are reasonable and necessary to protect the legitimate interests of the Purchaser and do not limit the remedies the Purchaser may have under applicable Law including, for the avoidance of doubt, the right to claim indemnification for any additional losses.

7.3 For the avoidance of doubt, nothing in paragraph 7.1 shall prevent the Institutional Seller from performing its undertakings as set out in paragraph 4.

8. Covenants

- 8.1 From the date of this Offer Letter until the earliest to occur of: (i) the execution date of the Share Purchase Agreement; (ii) the Offer Expiry Date; and (iii) termination of the Offer Letter in accordance with paragraph 3, the Institutional Seller undertakes to procure absolutely (se porte fort avec obligation de résultat) compliance with Schedule 3 (Conduct of Business Prior to Completion) to the Share Purchase Agreement (but subject to Clause 5.2 (Period Before Completion)) and Clause 7.3 (Access to Premises), Clause 7.4 (Further Assistance), Clause 7.5 (Information Covenant) and Clause 8.6 (Notifications regarding Warranties) of the Share Purchase Agreement as if such Schedule and Clauses were set out in full herein.
- 8.2 For the avoidance of doubt, nothing in paragraph 8.1 shall prevent the Institutional Seller from performing its undertakings as set out in paragraph 4.
- 8.3 From the date of this Offer Letter until the earliest to occur of: (i) the execution date of the Share Purchase Agreement; (ii) the Offer Expiry Date; and (iii) termination of the Offer Letter in accordance with paragraph 3, the Purchaser's Guarantor undertakes to comply with Clause 13 (Guarantee) of the Share Purchase Agreement as if such Clause was set out in full herein.

9. Confidentiality and Announcements

- 9.1 Subject to paragraphs 9.2 and 9.4, for 12 months after the date of this Offer Letter, the Institutional Seller shall treat as strictly confidential and not disclose or use any information which relates to:
 - (a) the Offer and the provisions of this Offer Letter; and
 - (b) the negotiations relating to the Offer and the provisions of this Offer Letter,

without the prior written approval of the Purchaser.

- 9.2 Paragraph 9.1 shall not prohibit disclosure or use of any information by any member of the Institutional Seller's Group to the extent:
 - (a) the disclosure or use is required by applicable Law, Governmental Entities or any recognised stock exchange on which the shares of any member of the Institutional Seller's Group are listed (including where this is required as part of any actual or potential offering, placing and/or sale of securities of any member of the Institutional Seller's Group);
 - the disclosure or use is required for the purpose of any applicable judicial or other dispute proceedings arising out of this Offer Letter or related documents;
 - (c) the disclosure is made to a Tax Authority in connection with the tax affairs of a member of the Institutional Seller's Group;
 - (d) the disclosure is made to professional advisers, auditors or bankers of any member of the Institutional Seller's Group on a need to know basis and on terms that such professional advisers, auditors or bankers undertake to comply with the provisions of paragraph 9.1 in respect of such information as if they were a party to this Offer Letter;
 - (e) the disclosure is made by the Institutional Seller to a member of the Institutional Seller's Group or, whether by the Institutional Seller or such member of the Institutional Seller's Group, to investors in funds managed or advised by members of the Institutional Seller's Group, subject to the recipient keeping such information confidential and undertaking to comply with the provisions of paragraphs 9.1 and 9.2 in respect of such information as if they were a party to this Offer Letter;
 - the information is or becomes publicly available (other than by breach of the Confidentiality Agreement or this Offer Letter); or
 - (g) the Purchaser has given its prior written consent,

provided that prior to disclosure or use of any information pursuant to paragraph (a), (b) and (c) above, the Institutional Seller shall (unless prohibited by law) promptly notify the

Purchaser of the circumstances of the disclosure and the information to be disclosed with a view to providing the Purchaser with the opportunity to contest, limit or agree the timing and content of such disclosure.

- 9.3 The Purchaser continues to be bound by the terms of the Confidentiality Agreement and, for the avoidance of doubt:
 - 9.3.1 the Offer and the provisions of this Offer Letter (including the Share Purchase Agreement); and
 - 9.3.2 the negotiations relating to the Offer and the provisions of this Offer Letter,

constitute "Evaluation Material" for the purposes of the Confidentiality Agreement.

- 9.4 Notwithstanding paragraphs 9.1 and 9.3, the Purchaser and the Institutional Seller agree that:
 - 9.4.1 upon countersignature of this Offer Letter by the Institutional Seller, the Purchaser (or any member of the Purchaser's Group) and the Institutional Seller may announce the Offer and the contents of this Offer Letter, in each case in such form as may have been previously agreed in writing between them; and
 - 9.4.2 subject always to compliance with its obligations under paragraph 4.4.3, the Purchaser (or any member of the Purchaser's Group) may issue, or cause to be issued, any other public announcement in connection with the existence or the subject matter of this Offer Letter, have any communication with the press in connection with the existence or the subject matter of this Offer Letter or make any other public statement or schedule any press conference or conference call with investors or analysts in connection with the existence or the subject matter of this Offer Letter, provided that to the extent reasonably practicable, the Purchaser shall provide the Institutional Seller with a copy of any such written public announcement, public statement or press release prior to its issue.

10. Counterparts

This Offer Letter may be entered into in any number of counterparts, all of which taken together shall constitute one and the same instrument. Any party may enter into this Offer Letter by executing any such counterpart.

11. Costs

11.1 Except as otherwise expressly provided in this Offer Letter or the Share Purchase Agreement, each party shall pay the fees, costs and expenses incurred by it in connection with the entering into and completion of this Offer Letter and the Transaction, provided that in the event that despite the delivery or deemed delivery of the Works Council Opinions in accordance with Schedule 1 on or before the Offer Expiry Date, the Institutional Seller does not validly accept the Offer in accordance with paragraph 2.3:

- (a) the period of exclusivity referred to in paragraph 7.1 shall automatically be extended for a period of six months and, accordingly, the provisions of paragraph 7.1 shall remain in full force and effect notwithstanding the subsequent lapse of the Offer; and
- (b) the Institutional Seller shall indemnify the Purchaser and the Purchaser's Guarantor for any reasonable and properly documented costs and expenses (including reasonable attorneys' fees) incurred by the Purchaser, the Purchaser's Guarantor or any member of the Purchaser's Group in connection with the Transaction, up to an aggregate maximum amount of EUR 10,000,000.

12. Miscellaneous

- 12.1 Each party agrees that it will promptly provide all information requested in writing by the other party reasonably required to enable the other party to comply with its obligations under the <u>Proceeds of Crime Act 2002</u> and the <u>Money Laundering Regulations 2007</u> (No. 2157) of the United Kingdom (or analogous legislation in any applicable jurisdiction).
- 12.2 The Offer Letter and the Confidentiality Agreement constitute the whole agreement between the parties relating to the Transaction to the exclusion of any terms implied in law that may be excluded by contract. They supersede and extinguish any and all prior discussions, correspondence, negotiations, drafts, arrangements, understandings or agreements relating to the Transaction.
- 12.3 Each party agrees and acknowledges that:
 - it is entering into the Offer Letter in reliance solely on the statements made or incorporated in the aforementioned documents and a letter from the other Sellers of even date herewith;
 - it is not relying on any other statement, representation, warranty, assurance or undertaking made or given by any person, in writing or otherwise, at any time prior to the date of this Offer Letter ("Pre-Contractual Statement");
 - except as expressly provided in this Offer Letter, it is entering into this Offer Letter solely in reliance on its own commercial assessment and investigation and advice from its own professional advisers; and
 - (d) the other party is entering into this Offer Letter in reliance on the acknowledgements given in this paragraph 12.3.
- 12.4 No party shall have any liability whatsoever for any Pre-Contractual Statement, whether in contract, in tort or otherwise, and is agreed that the only liability of any party in respect of those statements, representations, warranties, assurances and undertakings made or given by it and set out or incorporated in this Offer Letter shall be for breach of contract.
- 12.5 This entire agreement clause does not limit or exclude any liability for fraud.

- 12.6 The provisions of this Offer Letter shall inure to the benefit of and shall be binding on the parties hereto and their respective successors and permitted assigns, as the case may be.
- 12.7 The provisions of this Offer Letter will be severable. If any such provision or any part thereof is held to be unenforceable, void, invalid or illegal, in whole or in part, by any judicial or other competent authority, then the enforceability, validity or legality of the other provisions of this Offer Letter shall not be affected and will remain in full force and effect and will not in any way be impaired to the fullest extent permitted by Law.
- 12.8 No amendment or waiver of any provision of this Offer Letter shall be valid or binding unless made in writing and signed by an authorised representative of each of the parties. No failure or delay by a party in exercising any right, power or privilege under this Offer Letter shall operate as a waiver thereof, nor shall any single or partial exercise thereof preclude any other or future exercise of any right, power or privilege under this Offer Letter.
- 12.9 No party may assign, hold on trust, transfer, charge or otherwise deal with all or any part of its rights or obligations under this Offer Letter without the prior written consent of the other parties, provided that this Offer Letter and the benefits arising under it may be assigned in whole or in the part by the Institutional Seller to any member of the Institutional Seller's Group, or by the Purchaser to any member of the Purchaser's Group.
- 12.10 In the case of an assignment pursuant to paragraph 12.9 above, the liability of any party to such an assignee shall not be greater than it would have been had such assignment not taken place, and all the rights, benefits and protections afforded to a party shall continue to apply to the benefit of that party as against the assignee as they would have applied as against the assigning party.
- 12.11 A person who is not a party to this Offer Letter shall have no rights to enforce any of its terms.
- 12.12 Each of the parties acknowledges and agrees that the other parties would be damaged irreparably in the event any of the provisions of this Offer Letter are not performed in accordance with its specific terms or are otherwise breached. Accordingly, each party expressly waives the benefit of Article 1142 of the French Civil Code and agrees that the other parties shall be entitled to an injunction or injunctions to prevent breaches of the provisions of this Offer Letter and to enforce specifically this Offer Letter and the terms and provisions hereof in any action instituted in any court in France or in any state or country having jurisdiction over the parties and the matter in addition to any other remedy to which they may be entitled pursuant hereto.

13. Governing Law

- 13.1 The Offer and this Offer Letter and any non-contractual obligations arising out of or in connection with them shall be governed by and construed in accordance with French law.
- 13.2 Any dispute, claim, difference or controversy arising out of, relating to or having any connection with this Offer Letter, including any dispute as to its existence, validity,

interpretation, performance, breach or termination or the consequences of its nullity and any dispute relating to any non-contractual obligations arising out of or in connection with it, shall be referred to and finally resolved by arbitration under the Rules of Arbitration of the International Chamber of Commerce.

- 13.3 The number of arbitrators shall be three. Each party shall nominate, in the arbitration request and the answer, respectively, one arbitrator for confirmation. The arbitrators nominated by the parties shall jointly nominate the third arbitrator who will act as president of the arbitral tribunal. The arbitral tribunal shall not act as amiable compositeur or ex aequo et bono, but shall solely apply the laws of France in accordance with paragraph 13.1 of this Letter.
- 13.4 The seat or legal place of arbitration shall be London, England. Unless otherwise agreed by the parties, all hearings of the arbitral tribunal shall be conducted in London, England. The arbitral tribunal may deliberate at any location it considers appropriate. The language used in the arbitral proceedings shall be English. The award of the arbitral tribunal shall be final and binding on the parties and may be entered and enforced in any court having jurisdiction.

You	rs faithfully,
	/ Richard D. Mossehalf of MFB International Holdings s.à r.l.
	/ Joia M. Johnsonehalf of Hanesbrands Inc.

The Offer is irrevocable as from the date hereof but the countersignature by the Institutional Seller hereunder shall not be construed as an acceptance of the Offer to acquire the Shares and Warrants contemplated herein and any such acceptance shall only be made in accordance with the terms of paragraph 2 hereof.					
/s/ Lynn Skillen					
On behalf of SLB Brands Holdings, Ltd.					

Schedule 1

Works Council Procedures

- Delivery of Works Council Opinions. The Works Council Opinions will be treated as having been delivered in the following circumstances:
 - (a) in respect of the European Works Council, on the earlier of:
 - receipt of the acknowledgement from the European Works Council that the Exchange of Views with respect to the Transaction is completed; and
 - (ii) in the event that the European Works Council fails to provide such an acknowledgement after the exchange of views (échange de vues) has been completed (which shall be determined by the Institutional Seller acting reasonably, having regard to the European Works Council Agreement and any other legal requirements and past practice), upon such date as the Institutional Seller, acting reasonably and having consulted in good faith with the Purchaser, determines that no further information and consultation process is required; and
 - (b) in respect of the French Works Councils, on the earlier of:
 - receipt of an opinion from each of the French Works Councils with respect to the Transaction; or
 - (ii) deemed delivery of an opinion from each of the French Works Councils with respect to the Transaction in accordance with applicable Law.
- 2 Defined Terms. In this Offer Letter, the following terms have the meanings set out alongside them.
- 2.1 "European Works Council" means the DBA European Works Council as constituted by the European Works Council Agreement;
- 2.2 "European Works Council Agreement" means the DBA European Works Council Agreement, dated 4 June 1996 as amended on 12 December 2007;
- 2.3 "Exchange of Views" means the exchange of views (échange de vues) process of the European Works Council to be completed in accordance with the European Works Council Agreement;
- 2.4 "French Works Councils" means (i) the Central works council ("comite central d'entreprise") of DIM S.A.S and (ii) the "Délégation unique du personnel" of DBApparel Direct Marketing S.A.S;

- 2.5 "Works Council Opinions" means the advice of the Works Councils to be sought in connection with the Transaction in accordance with applicable law and/or as far as the DBA European Works Council is concerned, the completion of the Exchange of Views; and
- 2.6 "Works Councils" means the European Works Council and the French Works Councils and "Works Council" means any one of them as the case may be.

Schedule 2

Merger Control Filings

- 1 The Purchaser shall, and shall procure that each member of the Purchaser's Group shall, do all things legally permissible to satisfy the condition set out in Clause 4.3 of the Share Purchase Agreement, including proposing and agreeing to any divestments, complying with any prescriptions, conditions or undertakings, to ensure the satisfaction of such condition or to avoid, vacate or lift any action, including order or decision, that would otherwise have the effect of preventing satisfaction of such condition or which would otherwise prevent or make illegal Completion of the Transaction, as soon as practicable and in any event prior to 31 December 2015 or such later date as may be agreed by the Institutional Seller and the Purchaser. For the avoidance of doubt, all of the Purchaser's costs and expenses (including filing and advisors' fees) in relation to securing the Russian Antitrust Approval shall be borne by the Purchaser. The Institutional Seller shall be responsible for any costs and expenses (including filing and advisors' fees) incurred by the Institutional Seller in relation to securing Russian Antitrust Approval.
- 2 The Purchaser shall have primary responsibility for obtaining the Russian Antitrust Approval and shall use best endeavours to ensure satisfaction of the Joint Condition on or before the Long Stop Date, and in particular shall:
 - (a) as soon as reasonably practicable following the date of this letter and following the receipt of required or desirable information (in the requisite form) from the Institutional Seller, submit a merger control notification to the FAS;
 - (b) do all things legally permissible to avoid any declaration of incompleteness by the FAS, and any other suspension or setting of the applicable waiting or suspensory periods; and
 - (c) only withdraw the filing with the prior written consent of the Institutional Seller.
- 3 From and after the date of this letter, with respect to the condition set out in Clause 4.3 of the Share Purchase Agreement, the Institutional Seller and the Purchaser shall (to the extent allowed under applicable competition or antitrust Law):
 - (a) cooperate and coordinate in good faith and expeditiously with the other in exchanging information and supplying assistance that is reasonably necessary or requested for the purposes of such filing (on a counsel-tocounsel basis if the party to which the relevant information relates considers disclosure on that basis to be necessary);

- (b) promptly provide the other with advance copies and a reasonable opportunity to comment on all significant applications, notices, submissions, consents, filings, correspondence and materials supplied to or filed with the FAS and copies of all notices, consents, rulings, orders, correspondence and materials received from the FAS;
- consult with the other in advance of any significant meeting or other discussion with the FAS;
- (d) avoid taking any unilateral action in respect of any significant matter with the FAS or any other competent Governmental Entity without first reasonably consulting with the other;
- (e) promptly notify each other upon becoming aware that the Russian Antitrust Approval has been obtained; and
- (f) to the extent permitted by the FAS, permit the other to attend such significant meetings and discussions and where it has not been possible to involve the other, to promptly inform the other of the content of any such meeting or material discussion,

it being understood that each of the Institutional Seller and the Purchaser shall at all times provide the required information comprehensively, accurately and expeditiously so as to avoid any undue delay.

4 The Purchaser shall forthwith, by written notice, inform the Institutional Seller if the Purchaser has reasons to believe that the approval process with the FAS will be materially delayed or that an approval will not be received.

Exhibit A					
Share Purchase Agreement					

EXECUTION VERSION

D.	2014
Date:	, 2014
Date.	. 2014

SHARE PURCHASE AGREEMENT

relating to

DBA LUX HOLDING S.A.

THE INSTITUTIONAL SELLER

THE MANAGEMENT SELLERS

THE MANAGER HOLDING COMPANIES

THE PURCHASER

and

THE PURCHASER'S GUARANTOR

TABLE OF CONTENTS

	Page
1	Definitions and Interpretations4
2	Sale and purchase
3	Consideration
4	Conditions to Completion32
5	Period before Completion34
6	Completion34
7	Release of Guarantees, access to premises and books and records and further assistance
8	Warranties and undertakings44
9	Limitation of liability52
10	Tax information56
11	Restrictions on the Sellers57
12	Announcements and confidentiality59
13	Guarantee61
14	Notices
15	General65
SCHE	DULE 3 Conduct of Business prior to Completion83
SCHE	DULE 4 Completion Obligations86
SCHE	DULE 5 Business Warranties92
SCHE	DULE 6 Post Completion Adjustments112
SCHE	DULE 7 Accounting Policies119

Date:	, 2014
	,

PARTIES

- (1) SLB BRANDS HOLDING, LTD., a Cayman exempt limited company with registered number 161547 and registered office at c/o Maples Corporate Services Limited, PO Box 309, Ugland House, George Town, Grand Cayman, KY1-1104, Cayman Islands (the "Institutional Seller");
- (2) THE INDIVIDUALS set out in rows (b) to (g) in column (1) of Schedule 1 (the "Management Sellers" and each a "Management Seller");
- (3) MFB INTERNATIONAL HOLDINGS S.À R.L., a private limited liability company (societe à responsabilite limitee) established and existing under the laws of the Grand-Duchy of Luxembourg, having its registered office at 33, Rue du Puits Romain, L-8070 Bertrange, Grand Duchy of Luxembourg, with a share capital of USD 74,924,490 and registered with the Luxembourg Trade and Companies Register under number B-182082 (the "Purchaser");
- (4) HANESBRANDS INC. whose principal place of business is at 1000 East Hanes Mill Road, Winston-Salem, NC 27105, the United States of America (the "Purchaser's Guarantor");
- (5) SOCIÉTÉ CIVILE DE LA DUNE, a French société civile with a share capital of 1,000 € having its registered office at 7 allée Henri Marret - 78112 Fourqueux (France), registered with the commercial and companies registry of Versailles under number 788 667 780 (the "Manager Holding Company 1"); and
- (6) GUESHOV INVESTISSEMENT 1 a French société civile with a share capital of 1,000 € having its registered office at 74 rue de la Chaussée de l'Etang - 94520 Perigny-sur-Yerres (France), registered with the commercial and companies registry of Créteil under number 791 472 913 (the "Manager Holding Company 2", and together with the Manager Holding Company 1, the "Manager Holding Companies").

INTRODUCTION

- (A) The Sellers have agreed to sell the Shares and the Warrants and to assume the obligations imposed on the Sellers under this Agreement.
- (B) The Manager Holding Companies have agreed to be parties to this Agreement solely in order to agree to transfer the Warrants previously held by Christian Duval and François Riston respectively (which shall satisfy Christian Duval's and François Riston's obligations under Clause 2.4 below) and they shall have no liability or obligation in respect of any provision of this Agreement other than those in paragraph 2 of Schedule 4 (Completion Obligations).
- (C) The Purchaser has agreed to purchase the Shares and the Warrants and procure repayment of the Existing Shareholder Debt Redemption Amount and to assume the obligations imposed on the Purchaser under this Agreement.

- (D) The Purchaser's Guarantor has agreed to guarantee the payment obligations of the Purchaser under this Agreement.
- (E) There has been an exchange of views with the European Works Council and the opinions of the French Works Councils have been received, or deemed received.

1 DEFINITIONS AND INTERPRETATIONS

Definitions

1.1 In this Agreement, unless the context requires otherwise, the capitalised terms set out below have the following meanings:

"2012 Audited Accounts"

- (i) the consolidated, audited balance sheet, profit and loss statement and cash flow statement of the Group (including any notes thereon) as at 30 June 2012 in accordance with International Financial Reporting Standards as adopted by the European Union; and
- the audited balance sheet, profit and loss statement and cash flow statements (including any notes thereon) of each of the individual Group Companies as at such date prepared under the relevant Local GAAP;

"2013 Audited Accounts"

- the consolidated, audited balance sheet, profit and loss statement and cash flow statement of the Group (including any notes thereon) as at 30 June 2013 in accordance with International Financial Reporting Standards as adopted by the European Union; and
- (ii) other than for the purposes of Schedule 7 (Accounting Policies), the audited balance sheet, profit and loss statement and cash flow statements (including any notes thereon) of each of the individual Group Companies as at such date prepared under the relevant Local GAAP;

"Accounting Policies"

the accounting policies referred to or set out in Schedule 7 (Accounting Policies);

"Accounts Date"

the date of the 2013 Audited Accounts, being 30 June 2013;

"Actual Cash Amount"

the amount calculated in accordance with Part D of the Annex to Schedule 6 (Post Completion Adjustments) and the Accounting Policies and consistent with the categorisations included within the spreadsheet included with the Annexes to Schedule 6, being the aggregate amount of all relevant constituent line items comprising Cash held by or on behalf of the Group as at the Effective Time;

"Actual Cash

a statement setting out the Actual Cash Amount, as agreed or

Statement"

determined in accordance with Part D of the Annex to Schedule 6 (Post Completion Adjustments) and the Accounting Policies;

"Actual Debt Amount"

the amount calculated in accordance with Part C of the Annex to Schedule 6 (Post Completion Adjustments) and the Accounting Policies, being the aggregate amount of all relevant constituent line items comprising Indebtedness, or items specifically recorded therein as 'Debt-like' consistent with the categorisations included within the spreadsheet included with the Annexes to Schedule 6, which constitute liabilities or obligations of (or provisions in respect of the same made by) the Group as at the Effective Time and including, for the avoidance of doubt, the Existing Shareholder Debt;

"Actual Debt Statement"

a statement setting out the Actual Debt Amount, as agreed or determined in accordance with Part C of the Annex to Schedule 6 (Post Completion Adjustments) and the Accounting Policies;

"Actual Net Debt Amount"

the amount calculated by deducting (i) the Actual Cash Amount from (ii) the Actual Debt Amount;

"Actual Working Capital Amount"

the amount calculated in accordance with Part B of the Annex to Schedule 6 (Post Completion Adjustments) and the Accounting Policies and consistent with the categorisations included within the spreadsheet included with the Annexes to Schedule 6, being the aggregate amount of all relevant constituent line items comprising Working Capital of the Group as at the Effective Time;

"Actual Working Capital Statement"

a statement setting out the Actual Working Capital Amount, as agreed or determined in accordance with Part B of the Annex to Schedule 6 (Post Completion Adjustments) and the Accounting Policies;

this "Agreement"

this share purchase agreement, including the Introduction and the Schedules, as amended or restated from time to time;

"Articles"

the articles of association of the Company;

"Assets"

has the meaning given in paragraph 24(a) of Schedule 5 (Business Warranties);

"Balancing Amount" an amount equal to the aggregate of:

- the Working Capital Shortfall (expressed as a negative number) or the Working Capital Excess; and
- (ii) the Net Debt Shortfall or the Net Debt Excess

(expressed as a negative number);

"Bid Value" has the meaning given in Clause 3.1(a);

"Business Day" any day that is not a Saturday or Sunday or a public holiday

in France, the United Kingdom or Luxembourg;

"Business the warranties set out at Sche Warranties"

the warranties set out at Schedule 5 (Business Warranties);

"Business the escrow account at the Escrow Bank in the joint names of Warranties Escrow the Institutional Seller's Solicitors and the Purchaser's Solicitors for the purposes set out in the Business Warranties

Escrow Letter;

"Business

Warranties Escrow Amount" the sum of €2,000,000 to be paid into the Business Warranties Escrow Account in accordance with paragraph 6(e) of Schedule 4 (Completion Obligations);

"Business Warranties Escrow Letter" the letter agreement from the Sellers and the Purchaser to the Institutional Seller's Solicitors and the Purchaser's Solicitors in the agreed form setting out the terms on which the Business Warranties Escrow Account is to be operated;

"Business Warranties Escrow Release Date" the date that is 15 months from the Completion Date;

"Cash"

all cash or cash equivalents, determined in a manner consistent with the Accounting Policies, subject to Clause

3.2;

"Company"

DBA Lux Holding S.A., a société anonyme incorporated and existing under the laws of the Grand Duchy of Luxembourg, registered with the Luxembourg Trade and Companies Register under number B11180 and having its registered office at 5, Rue Guillaume Kroll, L-1882, Luxembourg;

"Company Shareholders' Agreement" the shareholders' agreement dated 15 July 2010 in respect of

the Company;

"Completion"

Initial Completion and Subsequent Completion, taken

together, at 17:30 BST on the Completion Date;

"Completion Announcement" the announcements relating to the Transaction to be made by the Purchaser (and/or a member of the Purchaser's Group) and the Institutional Seller on the Completion Date, which the Party making such announcement shall (i) use reasonable endeavours to disclose to the other Party sufficiently in advance of the proposed disclosure or issue and (ii) in good faith consider any reasonable requests and proposals made by such other Party (if any) in respect of the content of such announcement;

"Completion Date"

the date on which Completion is to take place, determined in accordance with Clause 6.1:

"Completion Statement"

a statement in the form set out in Parts A, B, C and D of the Annex to Schedule 6 (Post Completion Adjustments), setting out the total consideration payable under Clause 3.1 and the Balancing Amount payable by the Purchaser or the Sellers under Clause 3.13, as agreed and determined in accordance with Schedule 6 (Post Completion Adjustments) and the Accounting Policies, and including the Actual Cash Statement, the Actual Debt Statement and the Actual Working Capital Statement:

"Conditions"

together the Sellers Conditions, the Purchaser Condition and the Joint Condition, and "Condition" means each and any of them:

"Confidential Information"

with respect to any person, all information of a confidential or proprietary nature (whether or not specifically labeled or identified as "confidential"), in any form or medium, that relates to the business, products, financial condition, services or research or development of such person or their respective suppliers, distributors, customers, independent contractors or other business relations, including the following: (i) internal business and financial information (including information relating to strategic and staffing plans and practices, business, finances, training, marketing, promotional and sales plans and practices, cost, rate and pricing structures and accounting and business methods); (ii) identities of, individual requirements of, specific contractual arrangements with, and information about, that person's suppliers, distributors, customers, independent contractors or other business relations and their confidential information; (iii) trade secrets, know-how, compilations of data and analyses, techniques, systems, formulae, recipes, research, records, reports, manuals, documentation, models, data and data bases relating thereto; (iv) inventions, innovations, improvements, developments, methods, designs, analyses, drawings, reports and all similar or related information (whether or not patentable); and (v) other Intellectual Property;

"Confidentiality Agreement"

the confidentiality agreement dated 21 February 2014 and made between Hanesbrands Inc, and the Company;

"Consideration"

has the meaning given in Clause 3.1;

"Covenant Material any effect that would be materially adverse to the business,

Adverse Effect"

assets, financial condition or results of the operations of the Group taken as a whole; <u>provided</u>, <u>however</u>, that any adverse effect resulting from or relating to compliance with the terms of, or the taking of any action required by, this Agreement or any of the other Transaction Documents shall not be deemed, either alone or in combination, to constitute, and shall not be taken into account in determining whether there has been or will be, a Covenant Material Adverse Effect;

"Data Room"

the "Project Sublime" data room operated by Merrillcorp containing the documents listed in the Data Room Index;

"Data Room Index"

the index to the Data Room in the agreed form;

"Disclosed"

fairly disclosed in writing in the Disclosure Letter or the Disclosure Documents prior to the execution of the Offer Letter with sufficient detail to identify to the Purchaser the nature and scope of the matter disclosed;

"Disclosure Documents" those documents in the Data Room that are specifically referenced in the Disclosure Letter and all documents appended to the Disclosure Letter;

"Disclosure Letter"

the letter dated the Offer Date from the Sellers to the Purchaser together with the Disclosure Documents, which, for the avoidance of doubt, may not be amended or supplemented at any time after the Offer Date;

"DP Laws"

has the meaning given in paragraph 23(a)(i) of Schedule 5 (Business Warranties);

"Effective Time"

time immediately prior to Completion;

"Employment Tax Liability" has the meaning given in Clause 3.11;

"Encumbrance"

any claim, charge, mortgage, lien, option, equity, power of sale, hypothecation, usufruct, retention of title (other than any ordinary course retention of title arrangements applying to assets which have not been paid for), right of pre-emption, right of first refusal or other third party rights or security interest of any nature whatsoever or an agreement, arrangement or obligation to create any of the foregoing but, with respect to the Warranties contained in Clause 8.1(d)(ii) and Clause 8.2(c)(ii) only, excluding right of pre-emption or right of first refusal arising by operation of law which, while not applicable to the Transaction, could arise in the future;

"Environmental Laws" has the meaning given in paragraph 25(a) of Schedule 5 (Business Warranties);

"Escrow Bank"

a bank or financial institution which is reasonably acceptable to each of the Purchaser and the Institutional Seller to be agreed upon by such Parties between the date of execution of this Agreement and Completion or, failing such agreement within five Business Days prior to Completion, Barclays Bank Plc;

"Estimated Cash Amount"

the Sellers' good faith estimate of the Actual Cash Amount;

"Estimated Consideration"

has the meaning given in Clause 6.2;

"Estimated Debt Amount" the Sellers' good faith estimate of the Actual Debt Amount;

"Estimated Net Debt the Sellers' good faith estimate of the Actual Net Debt Amount' Amount;

"Estimated Working the Sellers' good faith estimate of the Actual Working Capital Capital Amount" Amount;

"Estimated Working the amount (if any) by which the Estimated Working Capital Capital Excess"

Amount exceeds the Working Capital Target;

"Estimated Working the amount (if any) by which the Estimated Working Capital Capital Shortfall" Amount is less than the Working Capital Target;

"European Works Council" the DBA European Works Council as constituted by the European Works Council Agreement;

"European Works Council Agreement"

the DBA European Works Council Agreement, dated 4 June 1996 as amended on 12 December 2007:

"Existing Shareholder Debt"

the amount (including, for the avoidance of doubt, principal plus accrued interest) outstanding and any premium or prepayment fees (if any), required to be paid, on redemption or repayment to the holders of the following instruments in accordance with their respective terms and conditions:

- the eight hundred thousand (800,000) 3% convertible preferred equity certificates issued by the Company to the Institutional Seller, dated 3 February 2006;
- the twenty two million five hundred and sixty nine thousand eight hundred and fifty eight (22,569,858)
 6.53% preferred equity certificates issued by the Company to the Institutional Seller, dated 3 February 2006 (and as amended on 27 March 2009);
- (iii) the one million six hundred and seventeen thousand six hundred and forty two (1,617,642) 7% contingent

preferred equity certificates issued by the Company to the Institutional Seller, dated 3 February 2006 (and as amended and supplemented on 27 March 2009);

 (iv) an intra-group loan agreement in the principal amount of €750,000 between the Company and the Institutional Seller, dated 7 November 2006 (and as amended on 27 March 2009);

in each case after taking into account the PEC Capitalisation Transaction; and

 the loan between DBApparel S.A.S. and the Institutional Seller entered into as consideration for the execution of the Settlement Agreement,

as set out in Schedule 6 (Post Completion Adjustments);

"Existing Shareholder Debt Redemption Amount" an aggregate amount equal to the principal plus accrued interest (if any) (including, for the avoidance of doubt, any premium or pre-payment fees (if any), required to be paid, on redemption or repayment of the Existing Shareholder Debt as at the Completion Date as set out in Schedule 6 (Post Completion Adjustments);

"Facilities"

has the meaning given in paragraph 6(a) of Schedule 5 (Business Warranties);

"FAS"

has the meaning given to it in Clause 4.3;

"Final Sri Lanka Escrow Release Date" the date after which no further Sri Lanka Claims may be brought against the Purchaser's Group pursuant to the Sri Lanka SPA, as a result of any combination of the following: (i) the expiry of the time periods within which Sri Lanka Claims may be brought or (ii) the Sri Lanka SPA prohibiting the bringing of Sri Lanka Claims or (iii) the obligations of the Purchaser's Group under the Sri Lanka SPA having terminated or been fully satisfied and discharged (as confirmed by the purchaser of Isabella (Private) Ltd.);

"Former Land and Buildings"

has the meaning given in paragraph 11(i) of Schedule 5 (Business Warranties);

"French Works Councils" (i) the Central works council ("comite central d'entreprise")
 of DIM S.A.S and (ii) the "Délégation unique du personnel"
 of DBApparel Direct Marketing S.A.S;

"Fundamental Obligations" the obligations of a Party listed in Schedule 4 (Completion Obligations) that are identified as such in that Schedule;

"German Employee all costs, liabilities, claims, obligations, losses, damages,

Reclassification Losses"

penalties, late payment charges, fines, interest and reasonable expenses (collectively, the "Reclassification Losses"), including (without limitation) legal, accounting and expert's fees and the incremental cost of employment of such German Freelancer(s) (including, without limitation, social security contributions, penalties, late payment charges, interest and fines), properly incurred by any member of the Purchaser's Group as a result of the classification under applicable Law (as such Law is in force at Completion) of a service relationship between any German Group Company and any such German Freelancer(s) as an employment relationship (as opposed to their status as freelancers (freier Mitarbeiter)), excluding all Reclassification Losses that arise as a result of changes to the service relationship made by the Purchaser's Group following Completion, save for any such changes made in accordance with reasonable commercial practices undertaken by the Purchaser or the relevant member of the Purchasers Group, unless said classification would not have occurred but for those changes; provided that, the aggregate amount of German Employee Reclassification Losses shall be limited to, and shall in no event exceed, €2,200,000;

"German Freelancers"

any freelancer (freier Mitarbeiter) that currently provides, or has provided at any time in the past, services to any of the German Group Companies, including without limitation, shelf fillers and shelf agents;

"German Group Company"

each of DBA Deutschland GmbH (formerly NUR DIE GmbH), DBA Bodywear Germany GmbH (formerly ELBEO GmbH, merged with DBA Intimates Deutschland GmbH, NDB Private Label GmbH and DBA Lovable Deutschland GmbH), VSE Verwaltungs GmbH, and Belfein GmbH;

"Governmental Entity"

any supra national, national, state, municipal or local government (including any subdivision, administrative agency or commission or other authority thereof) or any quasi governmental or private body exercising any regulatory, taxing, importing or other governmental or quasi governmental authority, including the European Union;

"Grants"

has the meaning given in paragraph 21(a) of Schedule 5 (Business Warranties);

"Group"

the Company and all of the Group Companies, taken together;

"Group Benefit Schemes"

has the meaning given in paragraph 13(a)(v) of Schedule 5 (Business Warranties);

"Group Company Guarantees"

any security, guarantees and indemnities given by or binding upon any Group Company in relation to or in connection with a liability of any Seller or any member of the Institutional Seller's Group or their respective Related Persons;

"Group Companies"

the Company and each of its subsidiaries, each being a "Group Company";

"Guarantee Beneficiaries" has the meaning given in Clause 13.1;

"Hazardous Substances"

has the meaning given in paragraph 25(a) of Schedule 5 (Business Warranties);

"Indebtedness"

in relation to a Group Company, all loans, borrowings, or other financing liabilities or obligations of the Group Company together with interest accrued and including the aggregate amount of all Employment Tax Liabilities which are the primary obligation of the relevant Group Company, in each case determined in a manner consistent with the Accounting Policies, and including, without limitation, the Facilities;

"Indemnified Claim" a claim by a third party against any member of the Purchaser's Group which gives rise to a claim for indemnification pursuant to Clause 8.9;

"Indemnity Escrow Account"

the escrow account at the Escrow Bank in the joint names of the Institutional Seller's Solicitors and the Purchaser's Solicitors for the purposes set out in the Indemnity Escrow Letter:

"Indemnity Escrow Amount"

the sum of €2,187,000 to be paid into the Indemnity Escrow Account in accordance with paragraph 6(h) of Schedule 4 (Completion Obligations);

"Indemnity Escrow Letter"

the letter agreement from the Sellers and the Purchaser to the Institutional Seller's Solicitors and the Purchaser's Solicitors in the agreed form setting out the terms on which the Indemnity Escrow Account is to be operated;

"Indemnity Escrow Release Date"

the date that is the third anniversary of the Completion Date;

"Information Memorandum" the information memorandum dated June 2012 and issued to potential purchasers including the Purchaser in connection with the sale of the Company;

"Initial Completion" completion of the sale and purchase of the Institutional Seller's Shares, and the Parties' compliance with their obligations, pursuant to Clause 6.5;

"Institutional Seller" has the meaning given in the Introduction;

"Institutional Seller's in relation to the Institutional Seller, the Institutional Seller Group" and its Related Persons (but excluding any Group Company);

"Institutional

the release letter in the agreed form delivered by the Shareholder Release" Institutional Seller at Completion in relation to the release and discharge of all obligations and liabilities of the Group Companies to the Institutional Seller and its Related Persons (excluding the limited partners in Sun Capital Partners Group IV, L.P. and Sun Capital Securities Fund, L.P. other than in their capacity as limited partners of those partnerships), save in respect of the shareholders agreement of the Company which shall continue only as between the Sellers (and not, for the avoidance of doubt, the Company) solely for the purposes of allocating the Consideration;

"Institutional Seller's Kirkland & Ellis International LLP of 30 St Mary Axe, Solicitors" London, EC3A 8AF, United Kingdom;

"Institutional Seller's the Kirkland & Ellis International LLP client account Solicitors' Account" (account number 58694155 at Barclays Bank, London, sort code 20-77-67, IBAN: GB46BARC20776758694155, swift code: BARCGB22);

"Intellectual Property"

all intellectual property including: (i) all trade marks, service marks, trade names, domain names, get-up, logos, slogans, brand names, internet domain names and corporate names (whether or not registered); (ii) all patents, inventions, registered and unregistered design rights, copyrights, copyrightable works (whether published or not), moral rights, website content, rights in software and in electronic data and database rights; (iii) all rights in know-how and other confidential information and trade secrets; (iv) all other similar rights in any part of the world (whether registered or capable of registration or not); and (v) all registrations and applications for registration in respect of any of (i) to (iv) above;

"IT System"

computer hardware and software which is in each case owned or used by the Group;

"Law"

with respect to any person, any legislation or other governmental, federal, provincial, state, local or municipal law (including common law), ordinance, regulation, rule, code, injunction, judgment, determination, directive or treaty applicable to such person;

"Licences" has the meaning given in paragraph 19(a) of Schedule 5

(Business Warranties);

"Licensed Intellectual Property" has the meaning given in paragraph 9(a) of Schedule 5

(Business Warranties);

"Local GAAP" the generally accepted accounting principles applicable in

each of the jurisdictions in which the relevant Group

Companies are incorporated;

"Local Transfer Agreement" an agreement in such form and content as is required to satisfy the minimum requirements of the applicable local Law for the transfer of the relevant assets or shares in the relevant jurisdiction, provided that a Party shall not, and shall procure that its Related Persons shall not, bring any claim (including, without limitation, for breach of any warranty, representation, undertaking, covenant or indemnity relating to the Transaction) against the other Party or any of its Related Parties in respect of, or based upon, any such Local Transfer Agreement, except to the extent necessary to enforce the transfer of such asset or shares in a manner consistent with the terms of this Agreement. All such claims (except as referred to above) shall be brought and be subject to the provisions, rights and limitations set out in this Agreement and no Party shall be entitled to recover damages or obtain payment, reimbursement, restitution or indemnity in respect thereof under or pursuant to any such Local Transfer Agreement;

"Long Stop Date" 31 December 2014;

"Lux Companies" each of DBA Lux 1 Sàrl, DBA Lux 2 Sàrl, DBA Lux Holding

S.A and DBA Lux Finance Sarl;

"Management Accounts" the unaudited management accounts relating to the Group drawn up for the nine month period ending on 31 March

2014;

"Management Equity the Warrants and Options; Securities"

"Management Equity the agreement in the agreed form between the Company, the Exercise Agreement" Institutional Seller and certain Management Sellers in relation

to the exercise of the Options, dated on or around the date of

this Agreement;

"Management Sellers" has the meaning given in the Introduction;

"Management Christian Duval or any replacement appointed pursuant to

Sellers' Representative" Clause 14.8;

"Material Adverse Effect"

any effect that would be materially adverse to the business, assets, financial condition or results of operation of the Group; provided, however, that none of the following shall be deemed, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there has been or will be, a Material Adverse Effect: (a) any failure by a Group Company to meet any internal or published projections, forecasts, or revenue or earnings predictions for any period ending on or after the Offer Date; (b) any adverse effect to the extent attributable to the announcement of the Transaction or that the Completion of the Transaction remains pending (including any cancellations of or delays in customer orders, any reduction in sales, any disruption in supplier, distributor, partner or similar relationships or any loss of employees); (c) any adverse effect attributable to conditions affecting (i) the industries in which the Group Companies operate, (ii) the European or world economy, capital markets, currency markets, financial or credit markets in general or changes affecting the availability or cost of financing, or (iii) the economy of any country in which a Group Company has operations or sales; (d) any adverse effect resulting from or attributable to any conditions caused by any national emergencies, acts of terrorism or war (whether or not declared), strikes, conflicts or civil disturbances (or the escalation of any of the foregoing) or any natural or manmade disaster or acts of God; (e) any adverse effect resulting from or relating to compliance with the terms of, or the taking of any action required by, this Agreement or any of the other Transaction Documents; (f) any adverse effect arising from or relating to any change in accounting requirements or principles or any change in Law or the interpretation or enforcement thereof; (g) any adverse effect arising from or relating to actions required to be taken by Law, a contract or other agreement which has been Disclosed; (h) any existing event, occurrence or circumstance with respect to which the Purchaser has knowledge as of the Offer Date or (i) any adverse effect on the business of the Group that is cured by the Sellers before the Completion Date. References in this Agreement to € or other monetary thresholds shall not be deemed to be evidence of a Material Adverse Effect;

"Material Contract" has the meaning given in paragraph 8 of Schedule 5 (Business Warranties);

"Net Debt Excess" the amount (if any) by which the Actual Net Debt Amount

exceeds the Estimated Net Debt Amount;

"Net Debt Shortfall" the amount (if any) by which the Actual Net Debt Amount is

less than the Estimated Net Debt Amount;

"Non-substantiated Claim"

a Seller Claim in respect of the Business Warranties which is

not a Substantiated Claim:

"Non-substantiated Indemnified Claim"

an Indemnified Claim which is not a Substantiated

Indemnified Claim;

"Non-substantiated Sri Lanka Claim"

a Sri Lanka Claim which is not a Substantiated Sri Lanka

Claim:

"Normal Business Hours"

has the meaning given in Clause 14.3;

"Offer Date" the date of the Offer Letter;

"Offer Letter" the letter agreement dated 25 June 2014 between the Sellers,

> the Purchaser and the Purchaser's Guarantor in which the Purchaser irrevocably undertook to enter into this Agreement

if required to do so by the Sellers;

"Options" the options issued to the Management Sellers to acquire 1,075

B Shares held in treasury by the Company, as set out at

column 3 of Schedule 1 (The Sellers);

"Option Exercise Consideration"

has the meaning given in Clause 3.3;

"Original Purchase Agreement"

the First Amended and Restated Purchase Agreement, dated 3 February 2006, made between Sara Lee Corporation and Branded Apparel France SAS, as the same may be novated pursuant to the novation agreement dated on or around the date hereof and made between the Institutional Seller, Sun Capital Partners IV, L.P., DBApparel SAS and DE Master

Blenders 1753 B.V.;

Property"

"Owned Intellectual has the meaning given in paragraph 9(a) of Schedule 5 (Business Warranties);

"Parties" the Institutional Seller, the Management Sellers, the

Purchaser and the Purchaser's Guarantor;

Transaction"

"PEC Capitalisation the capitalisation of the principal amount of €22,569,858 preferred equity certificates; and up to €2,078,749 contingent preferred equity certificates, together with all interest accrued thereon on the date immediately prior to the Completion

Date:

"Permitted deposits to secure the performance of trade contracts (other Encumbrance"

than for borrowed money), conditional sales contracts, leases, statutory obligations, surety bonds, performance bonds and other obligations of a like nature incurred by the Group in the ordinary course of business;

"Portfolio Company" a member of the Institutional Seller's Group that is not managed on a day to day basis by representatives of Sun Capital Partners IV, L.P., its general partner, or their respective investment advisors;

"Pre-Contractual Statement"

has the meaning given in Clause 15.18(b);

"Proceeding"

action, arbitration, mediation, audit, investigation, litigation, or suit (whether civil, criminal, administrative, judicial, or investigative) commenced, brought, conducted, or heard by or before, or otherwise involving, any governmental authority or arbitrator;

"Process Agent"

has the meaning given in Clause 15.33;

"Purchase Price Adjustment Escrow Account"

the escrow account at the Escrow Bank in the joint names of the Institutional Seller's Solicitors and the Purchaser's Solicitors for the purposes set out in the Purchase Price Adjustment Escrow Letter;

"Purchase Price Adjustment Escrow Amount"

the sum of €2,500,000 to be paid into the Escrow Account in accordance with paragraph 6(g) of Schedule 4 (Completion Obligations);

"Purchase Price Adjustment Escrow Letter"

the letter agreement from the Sellers and the Purchaser to the Institutional Seller's Solicitors and the Purchaser's Solicitors in the agreed form setting out the terms on which the Purchase Price Adjustment Escrow Account is to be operated;

"Purchaser"

has the meaning given in the Introduction;

"Purchaser Condition"

has the meaning given in Clause 4.2;

"Purchaser's Accountants" KPMG of 303 Peachtree Street, NE, Suite 2000 Atlanta, GA 30308-3210;

"Purchaser's Group" the Purchaser and its Related Persons (including, following Completion, the Group (other than, if notice is given pursuant to Clause 6.10, DIM RUS pending the Separate Russia Completion));

"Purchaser's Guaranteed

has the meaning given in Clause 13.1;

Obligations"

"Purchaser's Guarantor" has the meaning given in the Introduction;

"Purchaser's Solicitors" Cleary Gottlieb Steen & Hamilton LLP of City Place House, 55 Basinghall Street, London EC2V 5EH;

"Qualifying Sri Lanka SPA" a Sri Lanka SPA on terms which:

- (a) provide that the Group shall have no liability in relation to the sale of Isabella (Private) Ltd., other than any claims relating to capacity and authority of the relevant member of the Group to enter into the transactions contemplated by the Sri Lanka SPA and title of the relevant member of the Group to the shares of Isabella (Private) Ltd.;
- (b) are otherwise in all material respects consistent with those set out in the Sri Lanka Heads of Terms attached as Part B of Schedule 8 (Sri Lanka Sale) to this Agreement; and
- (c) provide that the sale of Isabella (Private) Ltd. shall be completed no later than 31 March 2015;

"Related Person"

- in the case of a body corporate, any holding company or subsidiary of that body corporate, or any subsidiary of any holding company of that body corporate;
- (b) in the case of an individual, any spouse and/or lineal descendant by blood or adoption of that individual or any person(s) acting in the capacity of trustee(s) of a trust of which that individual is the settlor; and
- (c) in the case of a limited partnership, any nominee or trustee of the limited partnership, the partners in that limited partnership or their nominees, any investment manager or investment adviser to the limited partnership, any parent undertaking or subsidiary undertaking of that investment manager or investment adviser whether direct or indirect and any other investment fund managed or advised by any such person and/or any investor in any fund that directly or indirectly holds interests in the limited partnership;

"Relevant Property" has the meaning given in paragraph 11(a) of Schedule 5 (Business Warranties);

"Relevant

the percentage in respect of each Seller set forth in column (6) of Schedule 1, provided that where some only of the

Proportion" Sellers are liable in respect of a Seller Claim, each Seller's

Relevant Proportion shall be that proportion which its percentage bears to the aggregate percentages of all Sellers'

who are so liable;

"Russia Purchaser" has the meaning given to it in Clause 6.14(a);

"Russia Seller" has the meaning given to it in Clause 6.14(a);

"Russia SPA" has the meaning given to it in Clause 6.14(a);

"Russian Antitrust

Approvals"

has the meaning given to it in Clause 4.3;

"Seller Claim" any claim, proceeding, suit or action against any Seller or any

member of the Institutional Seller's Group in respect of the Warranties, the undertakings in Clause 5.1 and Schedule 3 (Conduct of Business prior to Completion), in Clause 7.3 (Access to Premises), Clause 7.4 (Further Assurances) and in

Clause 7.5 (Provision of Certain Information);

"Seller Conditions" has the meaning given to it in Clause 4.1;

"Sellers" the Institutional Seller and the Management Sellers, each a

"Seller";

"Sellers' Ernst & Young LLP of 1 More London Place, London SE1

Accountants" 2AF, United Kingdom;

"Sellers' any security, guarantees and indemnities given by or binding Guarantees" upon any Seller or any member of the Institutional Seller's

upon any Seller or any member of the Institutional Seller's Group or any of their Related Persons in relation to or in connection with a liability of any Group Company, that have

been Disclosed;

"Senior Employee" has the meaning given in paragraph 13(a)(i) of Schedule 5

(Business Warranties);

"Separate Russia has the meaning given in Clause 6.10;

Completion"

"Separate Russia the date on which Separate Russia Completion is to take

place, determined in accordance with Clause 6.13;

"Separate Russia €1.00;

Completion Deduction"

Agreement"

Completion Date"

"Settlement the settlement agreement in the agreed form to be entered into

on Completion by and between DE Master Blenders 1753

B.V. (as successor to the Sara Lee Corporation), the

Institutional Seller and DBApparel S.A.S.;

"Shares"

the 99,999 A Ordinary Shares of €1.25 each in the capital of the Company, and the 1,075 B Ordinary Shares of €1.25 each in the capital of the Company to be sold hereunder which, together with the shares held by the Company in treasury, will comprise the entire issued share capital of the Company at Completion;

"Sri Lanka Actual Working Capital Allocation" the working capital of Isabella (Private) Ltd. which is included in the Actual Working Capital Amount in the Completion Statement;

"Sri Lanka Actual Cash Allocation" the Cash of Isabella (Private) Ltd. which is included in the Actual Cash Amount in the Completion Statement (but which, for the avoidance of doubt, shall exclude any cash received in relation to the Sri Lanka Net Proceeds):

"Sri Lanka Actual Debt Allocation" the Indebtedness and other debt-like items of Isabella (Private) Ltd. which is included in the Actual Debt Amount in the Completion Statement;

"Sri Lanka Claim"

a claim by the purchaser of Isabella (Private) Ltd. or any of its Related Persons against a member of the Purchaser's Group pursuant to the terms of the Sri Lanka SPA or any ancillary agreement entered into in connection with the sale of Isabella (Private) Ltd.;

"Sri Lanka Completion" completion of the sale of Isabella (Private) Ltd. pursuant to the terms of the Sri Lanka SPA;

"Sri Lanka Escrow Account" the escrow account at the Escrow Bank in the joint names of the Institutional Seller's Solicitors and the Purchaser's Solicitors for the purposes set out in the Sri Lanka Escrow Letter;

"Sri Lanka Escrow Letter" the letter agreement from the Sellers and the Purchaser to the Institutional Seller's Solicitors and the Purchaser's Solicitors in the agreed form setting out the terms on which the Sri Lanka Escrow Account is to be operated;

"Sri Lanka Escrow Release Date" each date on which there is a reduction of the Sri Lanka Potential Liability Amount pursuant to the terms of the Sri Lanka SPA (the amount of such reduction being referred to as the "Sri Lanka Reduction");

"Sri Lanka Net Proceeds" (a) the sale proceeds (after deduction of any Tax paid or payable by the relevant member of the Purchaser's Group thereon) received in cash by the Group on or after completion of the sale of Isabella (Private) Ltd. in

accordance with the Sri Lanka SPA; less

(b) all reasonable and properly documented costs incurred by the Group in completing the sale of Isabella (Private) Ltd. in accordance with the Sri Lanka SPA;

Liability Amount"

"Sri Lanka Potential an amount equal to the lesser of: (a) £4,000,000 (less the aggregate sum of all prior Sri Lanka Reductions (if any)) and (b) the aggregate potential monetary liability of the Purchaser's Group in respect of any Sri Lanka Claims (disregarding any monetary liability for those matters that comply with the requirements of a Qualifying Sri Lanka SPA);

"Sri Lanka SPA"

the share purchase agreement entered into by the relevant member of the Group for the sale of 100% of the issued and outstanding shares of Isabella (Private) Ltd. to any person not being any of the Sellers or any of their Related Persons;

Capital Excess"

"Sri Lanka Working the amount (if any) by which the Sri Lanka Actual Working Capital Allocation exceeds the Sri Lanka Target Working Capital Allocation;

Capital Shortfall"

"Sri Lanka Working the amount (if any) by which the Sri Lanka Actual Working Capital Allocation is less than the Sri Lanka Target Working Capital Allocation;

"Sri Lanka Target Working Capital Allocation"

€913,000;

"Subsequent Completion"

the completion of the sale and purchase of (i) 8,075 B Ordinary Shares, the subject of exercise of the Options; and (ii) the Warrants, and the Parties' compliance with their obligations pursuant to Clause 6.6;

"Substantiated Claim"

a Seller Claim in respect of the Business Warranties in respect of which liability is admitted or settled by the relevant Seller(s), or which has been adjudicated on by a court of competent jurisdiction from which there is no right of appeal or in respect of which adjudication the parties are debarred by passage of time or otherwise from making an appeal;

"Substantiated Indemnified Claim"

an Indemnified Claim in respect of which liability is admitted or settled by the relevant Seller(s), or which has been waived or withdrawn by the claimant or adjudicated on by a court of competent jurisdiction from which there is no right of appeal or in respect of which adjudication the parties are debarred by passage of time or otherwise from making an appeal, or in respect of which adjudication, the relevant member of the Purchaser's Group is not required to effect an appeal, judicial review or other challenge pursuant to Clause 8.10;

"Substantiated Sri Lanka Claim"

a Sri Lanka Claim in respect of which liability is admitted or settled by the relevant Seller(s), or which has been waived or withdrawn by the claimant or adjudicated on by a court of competent jurisdiction from which there is no right of appeal or in respect of which adjudication the parties are debarred by passage of time or otherwise from making an appeal;

"Surviving Provisions"

Clauses 1, 4, 9, 12, 13, 14 and 15;

"Tax" or "Taxation" all forms of taxation whether of The Grand Duchy of Luxembourg, France or elsewhere and whether direct or indirect and whether levied by reference to actual, deemed, gross or net income, profits, gains, net wealth, asset values, turnover, added value, receipt, payment, sale, use, occupation, franchise or values or other reference and statutory, governmental, state, provincial, local governmental or municipal impositions, duties, contributions, rates and levies (including without limitation social contributions and any other payroll taxes), whenever and wherever imposed (whether imposed by way of a withholding or deduction for or on account of tax or otherwise) and in respect of any person and all related penalties, charges, surcharges, fines, costs and interest relating thereto;

"Tax Authority"

any taxing or other authority competent to impose any liability in respect of Taxation or responsible for the administration and/or collection of Taxation or enforcement of any Law in relation to Taxation;

"Tikehau Bonds"

the €75,000,000 secured bonds due 2019 issued by DBApparel SAS;

"Tikehau Deed of Release"

the deed of release in a form which is reasonably acceptable to the Purchaser to be delivered by Tikehau Investment Management, in its capacity as bondholders' representative and U.S. Bank Trustees Limited, in its capacity as security agent at Completion, releasing the security in respect of the obligations of DBApparel SAS and the guarantors relating to the Tikehau Bonds, subject only to payment of the Tikehau Redemption Amount at Completion;

"Tikehau Redemption Amount"

the amount required to be paid at Completion to the holders of the Tikehau Bonds in accordance with the terms and conditions of the Tikehau Bonds by way of redemption of the Tikehau Bonds at Completion;

"Transaction"

the transactions contemplated by the Transaction Documents;

"Transaction Documents"

this Agreement, the Offer Letter, the Management Equity Exercise Agreement, the Disclosure Letter, each document in the agreed form and any other document entered into or to be entered into pursuant to this Agreement;

"VAT"

means, within the European Union, such Tax as may be levied in accordance with (but subject to derogations from) Directive 2006/112/EC, and elsewhere, any Taxation levied by reference to added value or sales and including any interest or penalties in respect thereof;

"Warranties"

the warranties in Clause 8.1 and Clause 8.2 and the Business Warranties:

"Warrants"

the warrants issued to the Management Sellers in respect of 4,175 Shares, as set out at column 4 of Schedule 1 (The Sellers):

Adverse Effect"

"Warranty Material any effect that would be materially adverse to the business, assets, financial condition or results of the operations of the Group taken as a whole; provided, however, that none of the following shall be deemed, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there has been or will be, a Warranty Material Adverse Effect: (a) any failure by a Group Company to meet any internal or published projections, forecasts, or revenue or earnings predictions for any period ending on or after the Offer Date; (b) any adverse effect to the extent attributable to the announcement of the Transaction or that the Completion of the Transaction remains pending (including any cancellations of or delays in customer orders, any reduction in sales, any disruption in supplier, distributor, partner or similar relationships or any loss of employees); or (c) any adverse effect resulting from or relating to compliance with the terms of, or the taking of any action required by, this or any of the other Transaction Documents. References in this Agreement to € or other monetary thresholds shall not be deemed to be evidence of a Warranty Material Adverse Effect;

"Working Capital"

at any given date, (x) the current assets of the Group minus (y) the current liabilities of the Group as set out in Part B of the Annex to Schedule 6 (Post Completion Adjustment), in each case determined in accordance with Schedule 6 (Post Completion Adjustments);

"Working Capital Excess"

the amount (if any) by which the Actual Working Capital Amount exceeds the Estimated Working Capital Amount;

"Working Capital

the amount (if any) by which the Actual Working Capital Amount is less than the Estimated Working Capital Amount; Shortfall" and

"Working Capital €68,529,000, subject to Clause 3.2. Target"

- 1.2 References to the "Parties" are to the parties to this Agreement, and each is a "Party".
- 1.3 References to "Clauses" are to the clauses of this Agreement.
- 1.4 References to the "Introduction" and the "Schedules" are to the introduction and schedules to this Agreement, which form part of this Agreement and have the same force and effect as if set out in the body of this Agreement.
- 1.5 Where any capitalised term is defined within a particular Clause in the body of this Agreement, that term shall bear the meaning ascribed to it in that Clause wherever it is used in this Agreement.

Interpretation

- 1.6 The table of contents and headings to Clauses and Schedules are included for ease of reference only, and are not to affect the interpretation of this Agreement.
- 1.7 In this Agreement, unless expressly stated otherwise:
 - (a) the words "include" or "including" (or any similar term) are not to be construed as implying any limitation, and general words shall not be given a restrictive meaning by reason of the fact that they are preceded or followed by words indicating a particular class of acts, matters or things;
 - (b) words indicating gender shall be treated as referring to the masculine, feminine or neuter as appropriate;
 - (c) a reference to a statute, statutory provision or subordinate legislation ("legislation") refers to such legislation as it is amended and in force from time to time and to any legislation that (either with or without modification) re-enacts, consolidates or enacts in rewritten form any such legislation; provided that as between the Parties no such amendment, re-enactment or modification shall apply for the purposes of this Agreement to the extent that it would impose any new or extended obligation, liability or restriction on, or would otherwise adversely affect the rights of, any Party;
 - (d) any reference to any document other than this Agreement is a reference to that other document as amended, varied, supplemented, or novated (in each case, other than in breach of the provisions of this Agreement) at any time;
 - a reference to a document "in the agreed form" means a form of document agreed by each of the Parties and signed by or on behalf of each Party for the purposes of identification;
 - (f) references to the time of day are to Central European Time;

- (g) references to "€" are to Euro, the lawful currency of certain member states of the European Union as constituted from time to time;
- (h) a reference to something being "in writing" or "written" includes any mode of representing or reproducing words in visible form that is capable of reproduction in hard copy form, including words transmitted by fax or email but excluding any other form of electronic or digital communication;
- a reference to a document or communication being "signed" by or on behalf
 of any person means signature in manuscript by that person or his duly
 authorised agent or attorney (which manuscript signature may be affixed
 and/or transmitted by fax or email) and not any other method of signature;
- (j) any reference to a "person" includes any individual, body corporate, trust, partnership, joint venture, unincorporated association or governmental, quasi-governmental, judicial or regulatory entity (or any department, agency or political sub-division of any such entity), in each case whether or not having a separate legal personality, and any reference to a "company" includes any company, corporation or other body corporate, and any limited partnership or limited liability partnership wherever and however incorporated or established;
- (k) any reference to a "holding company" or a "subsidiary" means a "holding company" or "subsidiary" as defined in section 1159 of the Companies Act 2006, save that a company shall be treated for the purposes of the membership requirement contained in sections 1159(1)(b) and (c) as a member of another company even if its shares in that other company are registered in the name of (i) its nominee or (ii) another person (or its nominee) by way of security or in connection with the taking of security. Any reference to an "undertaking" shall be construed in accordance with section 1169 of the Companies Act 2006 and any reference to a "parent undertaking" or a "subsidiary undertaking" means respectively a "parent undertaking" or "subsidiary undertaking" as defined in section 1162 of the Companies Act 2006, save that an undertaking shall be treated for the purposes of the membership requirement in sections 1162(2)(b) and (d) and section 1162(3)(a) as a member of another undertaking even if its shares in that other undertaking are registered in the name of (i) its nominee or (ii) another person (or its nominee) by way of security or in connection with the taking of security. Such references to an "undertaking", a "subsidiary undertaking" or a "parent undertaking" shall be amended, where appropriate, by the Limited Liability Partnerships (Accounts and Audit) (Application of Companies Act 2006) Regulations 2008;
- a person shall be deemed to be connected with another if that person is connected with another within the meaning of section 1122 of the Corporation Tax Act 2010; and
- (m) any reference to any English legal term for any action, remedy, method of judicial proceeding, legal document, legal status, court, official or any legal concept or thing shall, in respect of any jurisdiction other than England, be deemed to include what most nearly approximates in that jurisdiction to the English legal term.

1.8 The obligations of each Seller under this Agreement are several and not joint. No claim may be made against any Seller in respect of any breach of this Agreement by any other Seller.

2 SALE AND PURCHASE

Agreement to sell and purchase

- 2.1 On and subject to the terms of this Agreement, each Seller and each Manager Holding Company, shall sell, and the Purchaser shall purchase:
 - the A Ordinary Shares listed opposite such Seller's name in column 2 of Schedule 1 (The Sellers);
 - (b) such number of B Ordinary Shares as will, upon the Initial Completion and the consequential exercise of the Options, correspond with the number of Options opposite a Management Seller's name in column 3 of Schedule 1 (The Sellers); and
 - (c) the Warrants listed opposite a Management Seller's or Manager Holding Company's name in column 4 of Schedule 1 (The Sellers),
 - free from all Encumbrances and together with all rights attached or accruing to such Shares and/or Warrants (as the case may be) at the Completion Date.
- 2.2 The Sellers, the Manager Holding Companies and the Purchaser agree that the conveyance of the Shares and Warrants to complete such sale pursuant to this Agreement shall be effected by the delivery by each Seller, the Manager Holding Companies and the Purchaser of the transfer declaration forms referred to in paragraphs 1(a) and 2 of Schedule 4 (Completion Obligations) to this Agreement.

Waiver of pre-emption rights

- 2.3 Provided that such waiver shall be effective only as of the Subsequent Completion as regards the Management Sellers and Manager Holding Companies, and to the extent permitted by applicable Law, each Seller irrevocably waives any and all rights in respect of the Shares and each applicable Seller and each Manager Holding Company waives any and all rights in respect of the Warrants that may have been conferred on such Seller or Manager Holding Company by the Articles or otherwise, and such Seller and Manager Holding Company, shall, on or prior to Completion, procure the waiver of any rights in respect of such Seller's or Manager Holding Company's Shares or Warrants (as applicable) held by any other person pursuant to the Articles or otherwise, including, in each case:
 - (a) any rights of redemption, pre-emption, first refusal or transfer;
 - (b) any rights relating to the terms of transfer and/or the consideration, interest and/or dividends receivable for or on any Shares or Warrants; and
 - (c) any rights to acquire any Shares or Warrants.

Management Seller Undertaking

2.4 Christian Duval and François Riston hereby undertake separately in favour of the Purchaser to procure, on Completion, the transfer by their respective Manager Holding Companies of such number of Warrants as is set forth against their respective names in column 4 of Schedule 1 (the Sellers).

3 CONSIDERATION

Amount

- 3.1 Subject to Clauses 3.12 and 3.16, the total consideration for the purchase of the Shares and the Warrants under this Agreement (the "Consideration") shall be the aggregate of:
 - (a) €400,000,000.00 (the "Bid Value"); less
 - (b) the Actual Debt Amount; plus
 - (c) the Actual Cash Amount; plus
 - (d) if:
 - the Actual Working Capital Amount exceeds the Working Capital Target, the amount of such difference; or
 - (ii) the Actual Working Capital Amount is less than the Working Capital Target, the amount of such difference (expressed as a negative number);
 - (e) plus or less, as applicable, the adjustment in accordance with Clause 3.2 in relation to the sale of Isabella (Private) Ltd.

For the avoidance of doubt, in the event of the Separate Russia Completion, an amount equal to the Separate Russia Completion Deduction shall be deducted from the Estimated Consideration payable at Completion and shall be paid at the Separate Russia Completion instead.

3.2 The Consideration shall be adjusted in accordance with Part A of Schedule 8 (Sri Lanka Sale) in relation to the sale of Isabella (Private) Ltd. and any amounts payable to the Parties thereunder shall be paid by telegraphic transfer of immediately available funds to the Institutional Seller's Solicitors' Account or the Purchaser's Account (as applicable) or such other account as the relevant Party shall direct.

Consideration in respect of Options

3.3 Each Management Seller shall settle his or her exercise price in respect of the Options held by him or her by delivering a cheque in the requisite amount to the Company. Each Management Seller hereby irrevocably authorises the Purchaser to withhold upon Subsequent Completion such proportion of the Consideration as is equal to the portion of the exercise price of the Options (the "Option Exercise Consideration") which the relevant Management Seller has paid for by a cheque until the Company has confirmed that this cheque has been duly cleared. Upon clearance of such cheque, the Purchaser shall promptly transfer the Option Exercise Consideration to the

relevant Management Seller. If the cheque issued by the relevant Management Seller has not cleared by the tenth Business Day following the Completion Date, the Purchaser may retain the Option Exercise Consideration in respect of such Management Seller in satisfaction of such Management Seller's obligation to pay the exercise price in respect of the Options held by him or her.

Sri Lanka Escrow Account

- 3.4 The monies in the Sri Lanka Escrow Account shall be held under the terms of the Sri Lanka Escrow Letter for disbursement in accordance with Clauses 3.5, 3.6, 0 and 3.10.
- 3.5 As soon as reasonably practicable (and in any event within five Business Days) following a Sri Lanka Claim becoming a Substantiated Sri Lanka Claim, there shall be paid out of the Sri Lanka Escrow Account to the Purchaser a sum equal to the Substantiated Sri Lanka Claim (together with any interest that has accrued in the Sri Lanka Escrow Account in respect of such amount) or, if less, the total amount then standing to the credit of the Sri Lanka Escrow Account.
- 3.6 On each Sri Lanka Escrow Release Date, an amount (if any) equal to the Sri Lanka Reduction less:
 - the aggregate amount of any Non-substantiated Claims as at that date (together with any interest that has accrued in the Sri Lanka Escrow Account in respect of such amount); and
 - (b) any Substantiated Sri Lanka Claims which have not been paid as at that date (together with any interest that has accrued in the Sri Lanka Escrow Account in respect of such amounts),

shall be paid to the Sellers in their Relevant Proportion.

- 3.7 On the Final Sri Lanka Escrow Release Date, after deduction of:
 - the aggregate amount of any Non-substantiated Claims as at that date (together with any interest that has accrued in the Sri Lanka Escrow Account in respect of such amount); and
 - (b) any Substantiated Sri Lanka Claims which have not been paid as at that date (together with any interest that has accrued in the Sri Lanka Escrow Account in respect of such amounts),

all monies remaining in the Sri Lanka Escrow Account (after payment of all amounts pursuant to Clause 3.6(b)) shall be paid to the Sellers in their Relevant Proportion.

3.8 Following the Final Sri Lanka Escrow Release Date, as soon as reasonably practicable (and in any event within five Business Days) following any Non-substantiated Sri Lanka Claim which has been asserted by the Purchaser on or prior to the Final Sri Lanka Escrow Release Date becoming a Substantiated Sri Lanka Claim, there shall be paid out of the Sri Lanka Escrow Account:

- (a) to the Purchaser, a sum equal to the Non-substantiated Claim which has become a Substantiated Sri Lanka Claim (together with any interest that has accrued in the Sri Lanka Escrow Account in respect of such amount) or, if less, the total amount then standing to the credit of the Sri Lanka Escrow Account; and
- (b) to the Sellers in their Relevant Proportion, the balance (if any) of the monies remaining in the Sri Lanka Escrow Account (after deduction of the amount of all remaining Non-substantiated Sri Lanka Claims (if any, and for the avoidance of doubt excluding any Non-substantiated Sri Lanka Claims which have been withdrawn or dismissed by a court of competent jurisdiction) together with any interest that has accrued in the Sri Lanka Escrow Account in respect of such amounts).
- 3.9 The Institutional Seller and the Purchaser shall jointly instruct the Institutional Seller's Solicitors and the Purchaser's Solicitors and the bank holding the Sri Lanka Escrow Account in accordance with the terms of the Sri Lanka Escrow Letter to make the appropriate transfers from the Sri Lanka Escrow Account as required by the provisions of Clauses 3.5, 3.6, 3.7 and 0.

Apportionment

3.10 The Sellers agree that the Consideration to be paid in accordance with this Agreement shall be apportioned upon such payment pro-rata among the Sellers according to the number of Shares and Warrants sold by each of them, or, in the case of Christian Duval and François Riston, their Manager Holding Companies on Completion, but, in the case of each of the Warrants, having first deducted an amount equal to the exercise price that would be required to be paid under the terms of the agreement pursuant to which such Warrant was issued, and paying the remainder to the Warrantholder; provided that, the Purchaser shall not be concerned with such apportionment and upon payment of the Consideration by the Purchaser in accordance with this Agreement, the Purchaser shall be absolutely discharged from any and all obligations to pay the Consideration.

Employment Tax Liability

3.11 To the extent that the exercise, vesting, exchange or release of any Management Equity Securities on or before Completion, or the Sale of any Shares acquired through the exercise of any such Management Equity Securities, or a payment of the Consideration to a Management Seller results in any Group Company being required to account to any Tax Authority for any income tax, social security contributions or other employment related taxes in respect of the relevant Management Seller or Management Sellers (each, an "Employment Tax Liability") then it shall be a condition of the payment of the Consideration to such Management Seller or Management Sellers that an amount equal to such Employment Tax Liability (as determined in the discretion of the Purchaser, acting reasonably and in cooperation with the Sellers) is deducted from such payment and paid instead to the relevant Group Company and the Purchaser shall procure that the relevant Group Company shall promptly, and in any event within the periods prescribed by applicable Law, account to the relevant Tax Authority for the Employment Tax Liability; provided however that any Employment Tax Liability that is a primary obligation of the

relevant Group Company and not the relevant Management Seller or Management Sellers shall instead be taken into account in increasing the Actual Debt Amount and shall not be deducted from the Consideration under this Clause 3.11. The deductions and withholdings from the portion of the Consideration due to each Management Seller shall be in full and final settlement of the Sellers' liability, and the Sellers shall have no further obligations, in respect of the same.

Estimated Consideration

3.12 The Estimated Consideration shall be paid or satisfied in accordance with Clause 6 (Completion).

Settlement of Consideration

- 3.13 After final agreement on, or final determination of, the Completion Statement in accordance with Schedule 6 (Post Completion Adjustments), the following payments shall be made:
 - (a) if the Balancing Amount is zero, no payment shall be made by Sellers or the Purchaser and the balance of the Purchase Price Adjustment Escrow Account shall be paid to the Sellers in the Relevant Proportion.
 - (b) if the Balancing Amount is a positive number:
 - the balance standing to the credit of the Purchase Price Adjustment Escrow Account shall be released therefrom; and
 - (ii) an amount equal to the Balancing Amount shall be paid by the Purchaser,

to the Sellers in the Relevant Proportion.

- (c) if the Balancing Amount is a negative number, the lesser of:
 - the balance standing to the credit of the Purchase Price Adjustment Escrow Account; and
 - (ii) the amount equal to the Balancing Amount,

shall be paid to the Purchaser from the Purchase Price Adjustment Escrow Account. The balance of the amount standing to the credit of the Purchase Price Adjustment Escrow Account following the payments made pursuant to the preceding sentence (if any) shall be paid to the Sellers in the Relevant Proportion. If the amount standing to the credit of the Purchase Price Adjustment Escrow Account is insufficient to satisfy the entirety of the Balancing Amount (a "Negative Insufficiency") then each of the Seller shall pay to the Purchaser an amount equal to such Seller's proportion of the Negative Insufficiency based on the Relevant Proportion.

3.14 Any payment to be made under Clause 3.13 shall be made within five Business Days after (and excluding) the date on which the Completion Statement is agreed or determined in accordance with Schedule 6 (Post Completion Adjustments) by telegraphic transfer to:

- (a) in the case of payments to a Seller, account details of which shall be notified by the Institutional Seller to the Purchaser and the Purchaser's Solicitors at least five Business Days prior to the due date for payment; and
- (b) in the case of payments to the Purchaser, account details of which shall be notified by the Purchaser to the Sellers and the Sellers' Solicitors at least five Business Days prior to the due date for payment,

and such payment shall be an absolute discharge to the Person making the payment who shall not be concerned with the subsequent application of the amount paid.

- 3.15 If either the Purchaser or a Seller defaults in the payment when due of any sum payable under Clauses 3.13 and 3.14, such sum shall bear interest at the rate of 3% per annum for the period from but excluding the due date up to and including the date of actual payment (after as well as before judgment). Such interest shall accrue from day to day and shall be compounded monthly.
- 3.16 If any payment is to be made by any Seller to the Purchaser in respect of any Seller Claim, the payment shall so far as possible under applicable Law (but not so as to limit the amount payable, where not wholly possible) be made by way of adjustment of the Consideration paid by the Purchaser for the Shares and the Warrants under this Agreement, which shall be deemed to have been reduced by the amount of such payment.

Business Warranties Escrow Amount

- 3.17 The Business Warranties Escrow Amount shall be held under the terms of the Business Warranties Escrow Letter for disbursement in accordance with Clauses 3.18, 3.19, 3.20 and 3.21.
- 3.18 As soon as reasonably practicable (and in any event within five Business Days) following a Seller Claim in respect of the Business Warranties becoming a Substantiated Claim, there shall be paid out of the Business Warranties Escrow Account to the Purchaser a sum equal to the Substantiated Claim (together with any interest that has accrued in the Business Warranties Escrow Account in respect of such amount) or, if less, the total amount then standing to the credit of the Business Warranties Escrow Account.
- 3.19 On the Business Warranties Escrow Release Date, after deduction of:
 - the aggregate amount of any Non-substantiated Claims as at that date (together with any interest that has accrued in the Business Warranties Escrow Account in respect of such amount); and
 - (b) any Substantiated Claims which have not been paid as at that date (together with any interest that has accrued in the Business Warranties Escrow Account in respect of such amounts),

- all monies remaining in the Business Warranties Escrow Account (after payment of all amounts pursuant to Clause 3.19(b)) shall be paid to the Sellers in their Relevant Proportions.
- 3.20 Following the Business Warranties Escrow Release Date, as soon as reasonably practicable (and in any event within five Business Days) following any Non-substantiated Claim which has been asserted by the Purchaser on or prior to the Business Warranties Escrow Release Date becoming a Substantiated Claim, there shall be paid out of the Business Warranties Escrow Account:
 - (a) to the Purchaser, a sum equal to the Non-substantiated Claim which has become a Substantiated Claim (together with any interest that has accrued in the Business Warranties Escrow Account in respect of such amount) or, if less, the total amount then standing to the credit of the Business Warranties Escrow Account; and
 - (b) to the Sellers, the balance (if any) of the monies remaining in the Business Warranties Escrow Account (after deduction of the amount of all remaining Non-substantiated Claims (if any, and for the avoidance of doubt excluding any Non-substantiated Claims which have been withdrawn or dismissed by a court of competent jurisdiction) together with any interest that has accrued in the Business Warranties Escrow Account in respect of such amounts).
- 3.21 The Institutional Seller and the Purchaser shall jointly instruct the Institutional Seller's Solicitors and the Purchaser's Solicitors and the bank holding the Business Warranties Escrow Account in accordance with the terms of the Business Warranties Escrow Letter to make the appropriate transfers from the Business Warranties Escrow Account as required by the provisions of Clauses 3.18, 3.19 and 3.20.

4 CONDITIONS TO COMPLETION

- 4.1 The obligation of the Purchaser to complete the sale and purchase of the Shares and the Warrants under this Agreement shall be conditional on the following conditions (the "Seller Conditions") having been fulfilled or waived in accordance with this Agreement:
 - (a) (i) each of the Warranties contained in Clause 8.1 and Clause 8.2 (disregarding all qualifications and exceptions contained therein relating to materiality) being accurate in all material respects immediately prior to Completion by reference to the facts and circumstances subsisting at the Completion Date (on the basis that any reference in such Warranties, whether express or implied, to the Offer Date is substituted by a reference to the Completion Date) and (ii) each of the Business Warranties (disregarding all qualifications and exceptions contained therein relating to materiality, including references to "Material Adverse Effect") being accurate immediately prior to Completion by reference to the facts and circumstances subsisting at the Completion Date (on the basis that any reference in such Warranties, whether express or implied, to the Offer Date is substituted by a reference to the Completion Date), save to the extent that, in each case, any such inaccuracy would not have a Warranty Material Adverse Effect as at the Completion Date;

- (b) the Sellers having performed and complied with the covenants and obligations set out in Clause 5.1 and Schedule 3 of this Agreement on or before the Completion Date (and for such purpose, the words "use all corporate powers available to it to" and "use best endeavours to" shall be deleted from Clause 5.1), save to the extent that any such non-performance or non-compliance would not have a Covenant Material Adverse Effect as at the Completion Date; and
- (c) no Material Adverse Effect having arisen between the Offer Date and the Completion Date which is continuing as at the Completion Date.
- 4.2 The obligation of the Sellers to complete the sale and purchase of the Shares and the Warrants under this Agreement shall, unless otherwise waived by the Institutional Seller, be conditional on, save as specifically disclosed by the Purchaser to the Sellers in writing on or before the Offer Date, each of the warranties of the Purchaser contained in Clause 8.7 (disregarding all qualifications and exceptions contained therein relating to materiality) being accurate in all material respects immediately prior to Completion by reference to the facts and circumstances subsisting at the Completion Date (on the basis that any reference in such Warranties, whether express or implied, to the Offer Date is substituted by a reference to the Completion Date) (the "Purchaser Condition").
- 4.3 Subject to Clause 6.10, the obligation of the Parties to complete the sale and purchase of the Shares and the Warrants under this Agreement shall be conditional on the Federal Antimonopoly Service of the Russian Federation ("FAS") having adopted a decision approving the Transaction and permitting its Completion without any breach of applicable Law (the "Russian Antitrust Approval" or the "Joint Condition").
- 4.4 To the extent that the Russian Antitrust Approval has not been obtained by the time of the execution of this Agreement, the Parties shall continue to be bound by, and shall continue to comply with, the provisions of Clause 5 and Schedule 2 to the Offer Letter until such time as the Russian Antitrust Approval is obtained (and the provisions of Schedule 2 shall accordingly be deemed to be incorporated herein in their entirety).
- 4.5 The Purchaser may at any time waive in whole or in part any of the Seller Conditions in Clause 4.1 by notice to the Sellers. The Institutional Seller, on behalf of all Sellers, may at any time waive in whole or in part the Purchaser Condition by notice to the Purchaser.
- 4.6 Each Party shall notify the other Parties promptly upon, and in any event within two Business Days of, becoming aware that a Condition has been fulfilled or is no longer capable of fulfillment.
- 4.7 Subject to Clause 6.10, if the Conditions have not been satisfied or waived by the Long Stop Date (or such later date as the Parties may agree in writing) this Agreement shall automatically terminate (other than the Surviving Provisions).
- 4.8 In the event that this Agreement is terminated pursuant to Clause 4.7, no Party (nor any of their respective Related Persons) shall have any claim under this Agreement of any nature whatsoever against any other Party (or any of their respective Related

Persons) except in respect of any rights and liabilities which have accrued before termination or under any of the Surviving Provisions.

5 PERIOD BEFORE COMPLETION

- 5.1 Except as otherwise agreed with the Purchaser, and subject to Clause 5.2 below, from and including the Offer Date until Completion the Institutional Seller shall use all corporate powers available to it to procure, and the Management Sellers shall use best endeavours to procure, that the business of each Group Company (other than, in respect of a Management Seller, the Lux Companies), including the management of its working capital, is carried on only in the ordinary course and consistent with past practice and shall comply with the obligations set out in Schedule 3 (Conduct of Business prior to Completion).
- 5.2 Notwithstanding anything to the contrary in Clause 5.1, or any other provision of this Agreement or any other Transaction Document, no Seller, member of the Institutional Seller's Group or the Group shall be prevented from undertaking, be required to obtain the Purchaser's consent in relation to, or incur any liability as a result of effecting, any of the following on or prior to Completion:
 - (a) any matter required by Law;
 - the implementation of any transaction or the taking of any action required by any Transaction Document;
 - any matter which has been Disclosed in the Disclosure Letter with specific reference to this clause;
 - (d) the PEC Capitalisation Transaction;
 - the contribution of any funding to any Group Company by any other Group Company; or
 - (f) entering into a Sri Lanka SPA.
- 5.3 If a Seller becomes aware of any fact or circumstance that has resulted, results or is reasonably likely to result in a breach of Clause 5.1, it shall notify the Purchaser as soon as practicable and, in any event, prior to Completion, setting out such details as are then available to it, provided that such Seller shall not be liable for failure to provide such notice except to the extent, and only to that extent, that such Seller's failure causes the Purchaser's losses to be greater than they would have been if the Seller had given notice within the required period.

6 COMPLETION

Date and place

6.1 Completion shall take place at such place and time as the Purchaser and the Institutional Seller shall agree at least four Business Days prior to the anticipated Completion Date or, failing such agreement, at the offices of the Institutional Seller's Solicitors at 17:30 BST:

- (a) on the later of 1 August 2014 and the earliest Friday that is reasonably practicable following the date when the last of the Conditions (other than Conditions to be satisfied on the Completion Date and the Joint Condition in the event that the provisions of Clause 6.10 apply) are satisfied and/or waived by the Party entitled to grant such waiver; or
- (b) at such other date and time as the Parties may agree,

but, in any event, no later than the second Business Day after the Long Stop Date.

Notification of Estimated Consideration

- 6.2 The estimated consideration for the sale and purchase of the Shares and the Warrants (the "Estimated Consideration") shall be:
 - (a) the Bid Value; less
 - (b) the Estimated Debt Amount; plus
 - (c) the Estimated Cash Amount; less
 - (d) the Estimated Working Capital Shortfall (if any); plus
 - (e) the Estimated Working Capital Excess (if any);
 - (f) plus or less, as applicable, the adjustment in accordance with Clause 3.2 in relation to the sale of Isabella (Private) Ltd.; less
 - (g) the Separate Russia Completion Deduction (if applicable pursuant to Clause 6.12).
- 6.3 Four Business Days prior to Completion, the Institutional Seller shall provide to the Purchaser a statement which shall include the Estimated Debt Amount, the Estimated Cash Amount, the Estimated Working Capital Shortfall, (if any) and the Estimated Working Capital Excess (if any), together with a copy of a written confirmation issued by the Seller's Accountants to the Institutional Seller that they have reviewed such amounts, and such supporting material as reasonably evidence the calculation of such amounts, and its calculation of the Estimated Consideration (such calculation shall, in the absence of manifest error, be binding on the Purchaser for the purposes of Clauses 3.12 and 6.7).
- 6.4 The amounts notified by the Institutional Seller in accordance with Clause 6.3 shall be expressed in Euro.

Completion arrangements

6.5 Immediately prior to the transfer of (i) such number of additional B Ordinary Shares as will, upon the Initial Completion and the consequential exercise of the Options, correspond with the number of Options; and (ii) the Warrants at Subsequent Completion in accordance with Clause 6.6, Initial Completion in respect of the transfer of the Institutional Seller's Shares shall take place in accordance with this Clause 6. At Initial Completion, the Institutional Seller shall deliver or cause to be

- delivered the documents and confirmations required by paragraph 1 of Schedule 4 (Completion Obligations).
- 6.6 At Subsequent Completion, the Sellers shall deliver or cause to be delivered the documents required by paragraph 2 and paragraph 3 of Schedule 4 (Completion Obligations).
- 6.7 At Completion, and to the extent not completed at Initial Completion or Subsequent Completion the Sellers and the Purchaser shall comply with their respective obligations as specified in Schedule 4 (Completion Obligations). The Institutional Seller and the Purchaser's performance of their obligations under Clause 6.5, and the delivery of the transfer declaration form referred to in paragraph 1(a) of Schedule 4 (Completion Obligations), are conditional upon the Management Sellers' and the Manager Holding Companies' performance of their respective obligations pursuant to Clause 6.6 immediately following Initial Completion, failing which the acts undertaken by the Institutional Seller and Purchaser pursuant to Clause 6.5 shall be disregarded and treated as void, ab initio, and the transfer declaration form shall be returned to the Institutional Seller.
- 6.8 Notwithstanding anything to the contrary in this Agreement, Completion shall not occur unless and until the Sellers and the Manager Holding Companies have transferred all of the Shares and all of the Warrants, and not some only, to the Purchaser. All actions to be performed at Completion shall be deemed a single transaction so that, at the option of the Party for whose benefit an action is to be performed, Completion shall not be deemed to have taken place unless and until all such actions have been performed.
- 6.9 If any of the Fundamental Obligations of the relevant Party (including, for these purposes, the relevant Manager Holding Company) under Schedule 4 (Completion Obligations) is not complied with on the Completion Date, the Purchaser, in the case of non-compliance by the Sellers (including, for these purposes, the Manager Holding Companies) (or any of them), or the Institutional Seller, in the case of non-compliance by the Purchaser, may:
 - defer Completion (so that the provisions of this Clause 6 shall apply to Completion as so deferred);
 - (b) proceed to Completion as far as practicable (without prejudice to any rights or remedies such Party (including, for these purposes, the relevant Manager Holding Company) may have under this Agreement and/or applicable Law); or
 - (c) terminate this Agreement by notice to the other Parties (including, for these purposes, the Manager Holding Companies) without liability on the part of the terminating Party (including, for these purposes, the relevant Manager Holding Company), whereupon all of the provisions of this Agreement other than the Surviving Provisions shall cease to have effect, except that all rights and liabilities which have accrued before termination or under any of the Surviving Provisions shall continue to exist.

Separate Russian Completion

- 6.10 Notwithstanding anything to the contrary contained in this Agreement, in the event that the Seller Conditions and the Purchaser Conditions have been fulfilled or waived in accordance with this Agreement (other than Conditions to be satisfied on the Completion Date) but the Joint Condition has not yet been fulfilled, then the Purchaser or the Institutional Seller (on behalf of all of the Sellers) may, by notice in writing to the other Parties, waive the Joint Condition such that Completion shall occur in all respects other than with respect to the transfer of DIM RUS (Registered Number 1057749497673), for which a subsequent completion (the "Separate Russia Completion") shall be effected as provided in Clauses 6.11, 6.12, 6.13, 6.14 and 6.15 and in any event by no later than 31 December 2015; unless, prior to the Completion Date, the Joint Condition is or becomes fulfilled, in which case the remainder of this Clause 6 below shall be disregarded in its entirety.
- 6.11 On or prior to the Completion Date, the Sellers shall procure the transfer of 99% of the shares of DIM RUS from DIM S.A.S. to such other person, not being a Group Company, as the Institutional Seller may nominate in its sole and absolute discretion (the "Russia Seller"). The Parties acknowledge that:
 - (a) a Russian law Local Transfer Agreement shall be signed between DIM S.A.S. and the Russia Seller and notarized by a Russian public notary on the date of transfer in order for the transfer to be effective; and
 - (b) subsequent to such notarization and transfer, the Sellers shall procure that the transfer is registered with the Russian unified state register of legal entities in accordance with applicable Law.
- 6.12 In the event of a Separate Russia Completion, the portion of the Consideration payable at the initial Completion shall be reduced by an amount equal to the Separate Russia Completion Deduction. For the avoidance of doubt, notwithstanding the Separate Russia Completion, DIM RUS shall be taken into consideration for the purposes of the Completion Statement and in determining the Balancing Amount, such that the Separate Russia Completion Deduction payable on the Separate Russia Completion Date shall not be subject to any subsequent adjustment.
- 6.13 The Separate Russia Completion shall take place at such place and time as the Purchaser and the Institutional Seller shall agree at least four Business Days prior to the anticipated Completion Date or, failing such agreement, at the offices of the Institutional Seller's Solicitors at 17:30 BST:
 - (a) on the earlier of the earliest Friday that is reasonably practicable following the date when the Joint Condition is satisfied, taking into consideration, without limitation, the documents required to be agreed with and provided to a Russian public notary in connection with the Separate Russia Completion, and 31 December 2015; or
 - (b) at such other date and time as the Parties may agree.
- 6.14 At the Separate Russia Completion:

- (a) the Sellers shall procure that the Russia Seller shall transfer to the Purchaser or to such person as the Purchaser may nominate, in its sole and absolute discretion (the "Russia Purchaser"), 99% of the shares in DIM RUS, free from all Encumbrances and together with all rights attached or accruing to such shares at the Separate Russia Completion Date;
- (b) the Sellers shall procure that the Russia Seller shall, and the Purchaser shall, or shall procure that the Russia Purchaser shall, duly execute and deliver to the other the Local Transfer Agreement with respect to the shares of DIM RUS (the "Russia SPA"). The Parties shall procure the notarization of the Russia SPA with a Russian public notary in accordance with applicable Law;
- (c) the Sellers shall procure that the Russia Seller shall execute before a Russian public notary an application to the Russian tax authorities in Form P14001 "Application for the making amendments to the information contained in the Unified state register of legal entities" (the "Form P14001") or such other application form as may be required in connection with the transfer of the shares in DIM RUS under the Russia SPA;
- (d) the Sellers shall or shall procure that the following are delivered or made available to the Purchaser or to the Russia Purchaser to the Purchaser's reasonable satisfaction:
 - the share register of DIM RUS written up to three Business Days prior to the Separate Russia Completion Date showing the Russia Seller as the shareholder of DIM RUS;
 - (ii) an extract from the Russian unified state register of legal entities in relation to DIM RUS confirming the Russia Seller as the shareholder of DIM RUS and the identity of the general director of DIM RUS, dated not more than three Business Days prior to the Separate Russia Completion Date;
 - (iii) powers of attorney granted by the Russia Seller to its representatives to execute the Russia SPA and Form P14001;
 - (iv) offers, waivers and notifications in the agreed form in connection with the pre-emptive rights of shareholders in accordance with applicable Law and the constitutional documents of DIM RUS;
 - (v) the share register of DIM RUS written up to the Separate Russia Completion, including the transfer of the shares in DIM RUS to the Purchaser or the Russia Purchaser, as the case may be, with effect from the Separate Russia Completion;
 - (vi) all registers of DIM RUS duly written up to the Separate Russia Completion;
 - evidence of the termination of any arrangement between DIM RUS and any member of the Institutional Seller's Group in respect of any management, advisory or monitoring fees;

- (viii) if the Institutional Seller is, or has appointed any person as, director of DIM RUS, a letter of resignation in the agreed form in respect of such appointment;
- (ix) to the extent requested by the Purchaser in writing at least 10 Business Days' before the Separate Russia Completion and the Sellers having used reasonable endeavours to obtain the same, the written resignations of the auditors of DIM RUS to take effect on the Separate Russia Completion Date;
- (e) each Seller shall use all corporate powers available to such Seller to:
 - procure the passing of board resolutions of the Russia Seller approving the share transfers and the Russia SPA referred to in Clauses 6.14(a) and (b);
 - (ii) procure the passing of a board resolution of DIM RUS approving the acceptance of and registration of the share transfers referred to in Clause 6.14(a);
 - (iii) procure the passing of board or shareholder resolutions of DIM RUS (as applicable pursuant to relevant Law and the constitutional documents of DIM RUS):
 - (A) accepting the resignations referred to in Clauses 6.14(d)(viii) and (ix);
 - (B) appointing such persons as the Purchaser or the Russia Purchaser, as the case may be, may nominate to be the director(s) of DIM RUS;
 - (C) changing the registered office of DIM RUS;
- (f) the Purchaser shall or shall procure:
 - payment of an amount equal to the Separate Russia Completion Deduction to the Sellers by telegraphic transfer of immediately available funds to the Institutional Seller's Solicitors' Account; and
 - (ii) if applicable, delivery to the Institutional Seller of evidence that the Russia Purchaser is authorised to purchase the shares of DIM RUS in accordance with this Agreement.
- 6.15 Notwithstanding anything to the contrary in this Agreement, the Separate Russia Completion shall not occur unless and until the Sellers have transferred 99% of the shares in DIM RUS (other than the one share retained by DIM S.A.S), and not some only, to the Purchaser or the Russia Purchaser, as the case may be. All actions to be performed at the Separate Russia Completion shall be deemed a single transaction so that, at the option of the Party for whose benefit an action is to be performed, the Separate Russia Completion shall not be deemed to have taken place unless and until all such actions have been performed.

- 6.16 From and after the Completion Date until the Separate Russia Completion Date:
 - (a) the Purchaser shall procure that the Group continues to provide such services to DIM RUS as were provided to DIM RUS by the Group prior to the Completion Date, so as to enable DIM RUS to continue to conduct its business during that period on substantially the same terms as in effect as at the Completion Date;
 - (b) the Purchaser shall, forthwith following demand from the Institutional Seller, provide the minimum amount of financial support necessary to allow DIM RUS to continue to trade in the ordinary course on substantially the same terms as in effect as at the Completion Date;
 - (c) if at any time the business of DIM RUS is loss making, the Sellers shall consider and, as a condition of receiving additional financial support, implement the reasonable directions of the Purchaser and act in accordance with prudent business practices in order to mitigate such losses (provided that the minimum amount of funding necessary pursuant thereto shall be provided by the Purchaser forthwith following demand from the Institutional Seller);
 - (d) if at any time the business of DIM RUS is, or is reasonably likely to be, adversely affected due to the occurrence of an event outside of the ordinary course of trading, the Sellers shall consult with the Purchaser in respect thereof and, following consultation with the Purchaser, make the unilateral decision either to (i) take all steps necessary or desirable to liquidate the business of DIM RUS, provided that all reasonable and properly documented costs of liquidating the business of DIM RUS shall be borne by the Purchaser and the Sellers shall have no liability in respect of the same or (ii) allow DIM RUS to continue to trade (in which event the provisions of Clause 6.16(b) and (c) above shall continue to apply); and
 - (e) the Sellers shall continue to comply with all covenants and agreements contained in Clause 5 and Schedule 3 (Conduct of Business prior to Completion), Clause 7.3, Clauses 7.4(c) and (f) and Clauses 7.5(b), (c), (d) and (e), solely in respect of DIM RUS, and, unless the context clearly requires otherwise, all references in such covenants and agreements to the "Completion" or the "Completion Date" shall be deemed to refer to the Separate Russia Completion Date, respectively.

The Parties agree that the fact that the Separate Russia Completion has not occurred will not, in and of itself, prejudice the effectiveness of Completion.

7 RELEASE OF GUARANTEES, ACCESS TO PREMISES AND BOOKS AND RECORDS AND FURTHER ASSISTANCE

7.1 The Purchaser shall use reasonable endeavours to procure at Completion or, to the extent not done by Completion, as soon as reasonably practicable thereafter the full and unconditional release of each Seller and each member of the Institutional Seller's Group or any person connected with any of them from any Sellers' Guarantees that have been Disclosed in the Disclosure Letter with specific reference to this clause.

- Pending such release, the Purchaser shall indemnify and keep indemnified each Seller and each member of the Institutional Seller's Group from and against any amounts paid by any of them pursuant to any such Sellers' Guarantees.
- 7.2 The Sellers shall procure by Completion or, to the extent not done by Completion, as soon as reasonably practicable thereafter, the full and unconditional release of each Group Company from any Group Company Guarantees. Pending such release, the Sellers shall severally indemnify and keep indemnified the Purchaser (on its own behalf and on behalf of each of the Group Companies) from and against any amounts paid by any Group Company pursuant to any Group Company Guarantees.
- 7.3 From the Offer Date, the Institutional Seller shall use all corporate powers available to it to, and the Management Sellers shall use their respective reasonable endeavours to, procure that the Purchaser is given access at the sole expense of the Purchaser to the premises of, and the books and records of or relating to, the Group Companies, as well as to the Group's directors and employees. In each case such access shall be subject to applicable Law, be on reasonable notice during normal business hours and shall be subject to there being no material interruption to, or interference with, the ordinary course of business of the Group.
- 7.4 In the period between the Offer Date and the Completion Date, the Sellers shall cooperate with and provide or use reasonable endeavours to procure the provision by the Group of such assistance to the Purchaser as it may reasonably request in connection with:
 - (a) repayment of any of the Facilities at Completion as the Purchaser may determine in its sole and absolute discretion, discharge of all amounts owed under any associated hedging arrangements and release of all security in relation thereto, including without limitation in relation to agreeing the form of the Tikehau Deed of Release with the security agent in respect of the Tikehau Bonds and delivering such notices as may be required in relation to the prepayment of the Tikehau Bonds at Completion;
 - refinancing any of the Facilities, as the Purchaser may determine in its sole and absolute discretion;
 - (c) receipt of change of control consents under agreements entered into by the Group, which obligation shall be discharged and satisfied by the Group's sending of letters in the agreed form to the relevant counterparty and responding to such information requests as may reasonably be made by, and clarifications requested by, the relevant counterparty;
 - (d) raising debt to finance the payment of the Consideration, which obligation shall be discharged and satisfied by making the CFO of the Group available to attend a reasonable number of meetings of reasonable duration and on reasonable notice;
 - (e) effecting amendments to, or delivering unilateral waivers to the counterparties in respect of certain provisions of, the distribution agreements entered into by the Group and franchise agreements entered into by the Group with respect to its operations in Italy at the cost of the Group;

- registration with Rospatent of the franchise agreements entered into by the Group with respect to its operations in Russia at the cost of the Group;
- (g) making the following Intellectual Property filings at the cost of the Group:

Trademark	Country	Filings
NER DIE and NER DER Logos	China	File trademark applications for NUR DIE and NUR DER logos to cover all subclasses in Class 25. File for copyright registration for the NUR DIE and NUR DER logos and record with customs.
ABANDERADO Logo	China	File trademark application for the ABANDERADO logo to cover all subclasses in Class 25. File for copyright registration for the ABANDERADO logo and record with customs.
BELINDA, OCEAN and PRINCESA Logos	China	File trademark applications for the BELINDA, OCEAN and PRINCESA logos to cover all subclasses in Class 25. File for copyright registrations for the BELINDA, OCEAN and PRINCESA logos and record with customs.
NUR DIE, NUR DER, BELLINDA Logos	Vietnam	File trademark applications for NUR DIE, NUR DER and BELLINDA logos to cover Class 25.
SHOCK ABSORBER and BELLINDA Logos	Philippines	File trademark applications for the SHOCK ABSORBER and BELLINDA logos to cover Class 25.
SHOCK ABSORBER and BELLINDA Logos	Turkey	File trademark applications for the SHOCK ABSORBER and BELLINDA logos to cover Class 25.

(h) objecting to the following third party trade mark registrations in China at the cost of the Group:

Third Party Trademark	Action	
NURDIE, Reg. No. 5945862	Investigate trademark registration for NURDIE by Wenzhou Sea Culture Communication Co., Ltd. File non-use cancellation as appropriate.	
ABANDERADO, App. No. 11999774	Investigate trademark application for ABANDERADO by France Penguin Int'l Company Ltd.	
	 Monitor status of application and file opposition to application when it is published. 	

 obtaining a waste water discharge permit from the Laguna Lake Development Authority, Philippines in respect of the activities carried out by Piplay, Inc. at the cost of the Group,

provided that, in each case such cooperation and assistance shall be subject to applicable Law and shall be subject to there being no material interruption to, or interference with, the ordinary course of business of the Institutional Seller, any member of the Institutional Seller's Group or the Group and provided always that there shall be no obligation on any Group Company, any Seller or any member of the Institutional Seller's Group to enter into any binding agreement or arrangement in relation to the matters set out in sub-clauses (a), (b), (c) and (d) which is not conditional on the occurrence of Completion.

- 7.5 In the period between the Offer Date and the Completion Date (or, in the case of subclauses (b), (c), (d) and (e) below to the extent applicable to DIM RUS only, between the Offer Date and the Separate Russia Completion Date), the Sellers shall, and shall procure that the Group Companies shall:
 - (a) keep the Purchaser regularly informed of the progress of, and promptly provide the Purchaser with all information as it may reasonably request from time to time in connection with, the sale of Isabella (Private) Ltd., including without limitation, a copy of any agreement in relation thereto that is proposed to be entered into by the Sellers or any Group Company prior to its execution;
 - (b) keep the Purchaser regularly informed of the progress of, and promptly provide the Purchaser with all information as it may reasonably request from time to time in connection with, the transfer of the shares of DIM RUS from DIM S.A.S. in relation to the Separate Russia Completion;
 - (c) provide the Purchaser with copies of the interim monthly financials and monthly consolidation workbook relating to the Group as soon as reasonably practicable after they are available, and in any event within 20 days after the end of the month to which they relate;
 - (d) provide the Purchaser with copies of the consolidated, audited balance sheet, profit and loss statement and cash flow statement of the Group (including any notes thereon) as at 30 June 2014 in accordance with International Financial Reporting Standards as adopted by the European Union and the unaudited balance sheet, profit and loss statement and cash flow statements (including any notes thereon) of each of the individual Group Companies as at such date prepared under the relevant Local GAAP, in each case, as soon as reasonably practicable after they are available; and
 - (e) provide the Purchaser, at the Purchaser's cost, with all such information as the Purchaser may reasonably request in connection with any filings or notifications required to be made by the Purchaser or any member of the Purchaser's Group under applicable Law in relation to the Transaction, including without limitation, in order to prepare any pro forma financial statements that the Purchaser or any member of the Purchaser's Group may need to file under applicable Law.

Sri Lanka Sale

- 7.6 The Sellers shall procure that the Group shall not become a party to, or otherwise enter into, a Sri Lanka SPA where the aggregate potential monetary liability of the Purchaser's Group thereunder exceeds €4,000,000.
- 7.7 In the event that a Sri Lanka SPA is entered into prior to the Completion Date, but the sale thereunder is not completed prior to the Completion Date in accordance with its terms, the Purchaser shall procure that with effect from Completion the Group shall prosecute the completion of that sale with all such diligence as is reasonably practicable in the circumstances.
- 7.8 If a Sri Lanka SPA is not entered into prior to the Completion Date:
 - (a) the Institutional Seller (or its nominee) shall be entitled on ten Business Days' notice to acquire Isabella (Private) Ltd. from the Group at any time prior to the Completion Date for an aggregate consideration of €1.00 on the same terms as a Qualified Sri Lanka SPA (save as to price and the completion date of the sale), and the Purchaser shall procure that the Group takes all actions as may be reasonably necessary to effect the transfer of Isabella (Private) Ltd. to the Institutional Seller or its nominee pursuant thereto within 30 days of such notice; and
 - (b) the provisions of columns (1) and (2) in row (1) of Part A of Schedule 8 (Sri Lanka Sale) shall apply mutatis mutandis to such sale.
- 7.9 If a Sri Lanka SPA is entered into prior to the Completion Date, but the completion of the sale of Isabella (Private) Ltd. in accordance with the Sri Lanka SPA does not occur prior to 31 March 2015 (other than a result of the seller defaulting on its obligations thereunder) and as a result, the Sri Lanka SPA terminates with no obligation or liability of any member of the Purchaser's Group:
 - (a) the Institutional Seller (or its nominee) shall be entitled on ten Business Days' notice to acquire Isabella (Private) Ltd. from the Group at any time prior to 31 March 2015 for an aggregate consideration of €1.00 on the same terms as a Qualified Sri Lanka SPA (save as to price), and the Purchaser shall procure that the Group takes all actions as may be reasonably necessary to effect the transfer of Isabella (Private) Ltd. to the Institutional Seller or its nominee pursuant thereto within 30 days of such notice; and
 - (b) the provisions of columns (1) and (2) of row (2) of Part A of Schedule 8 (Sri Lanka Sale) shall apply mutatis mutandis to such sale.

8 WARRANTIES AND UNDERTAKINGS

Institutional Seller's Warranties

8.1 The Institutional Seller warrants to the Purchaser that as of the date hereof and in all material respects as at Completion and, in the case of DIM RUS only, as at the Separate Russia Completion, such that all references in this Clause 8.1 to the "Institutional Seller", "Shares", "Company" or "Group Company" and "Completion"

shall be deemed to refer to the seller of DIM RUS, the shares of DIM RUS, DIM RUS and the Separate Russia Completion, respectively together with such other amendments as are required in the context of the Separate Russia Completion:

- it is the sole legal and beneficial owner of the Shares set forth against its name in column 2 of Schedule 1 (The Sellers);
- save for any Encumbrance that will be discharged upon Completion, there is no Encumbrance in relation to any such Share;
- the Shares comprise the whole of the issued share capital of the Company;
- (d) the shares in each Group Company set out in Schedule 2 (The Company and the Subsidiaries):
 - comprise the whole of the issued and allotted share capital of that Group Company; and
 - (ii) are, save as set out in Schedule 2 (The Company and the Subsidiaries), directly or indirectly held by the Company or another Group Company and, save as set out in Schedule 2 (The Company and the Subsidiaries), the Company or such other Group Company is the sole legal and beneficial holder of such shares which are free from all Encumbrances;
- (e) other than this Agreement, the Management Equity Exercise Agreement and the terms of the Management Equity Securities, there is no legally binding agreement, arrangement or obligation (whether exercisable now or in the future and whether or not contingent) requiring the allotment or issue or the grant to any person of the right to require the allotment or issue of any share or loan capital or other securities giving rise to a right over or interest in, the capital of the Company or any other Group Company;
- the Institutional Seller is a company duly incorporated and in existence under the Laws of the Cayman Islands;
- (g) the Institutional Seller has full power and authority and has taken all corporate action required of it to enter into, deliver and perform its obligations under this Agreement and each other Transaction Document to which the Institutional Seller is a party;
- (h) this Agreement and each other Transaction Document to which the Institutional Seller is a party will, when executed, constitute valid, binding and enforceable obligations of the Institutional Seller in accordance with their respective terms;
- (i) the execution and delivery of, and the performance by the Institutional Seller of its obligations under this Agreement and each other Transaction Document to which the Institutional Seller is a party will not:

- conflict with or result in a breach of any provision of the articles of association or other constitutional documents of the Institutional Seller;
- (ii) conflict with, result in a breach of or constitute a default (with or without the lapse of time and/or the giving of any notice, certificate, declaration or demand) under any agreement or instrument to which the Institutional Seller is a party;
- (iii) conflict with or result in a breach of any Law or regulation, or of any order, injunction, judgment or decree of any court or Governmental Entity, that applies to the Institutional Seller; and
- (iv) save as set out in this Agreement, require the Institutional Seller to obtain any consent or approval of, or give any notice to or make any registration with, any Governmental Entity that has not been unconditionally and irrevocably obtained or made at the date of this Agreement (save for any legal or regulatory entitlement to revoke the same other than by reason of any misrepresentation or misstatement).

Management Sellers' Warranties

- 8.2 Each Management Seller severally warrants to the Purchaser that (save as specified in sub-clause 8.2(a) below) as of the date hereof and in all material respects as at Completion and, in the case of DIM RUS only, as at the Separate Russia Completion, such that all references in this Clause 8.2 to the "Management Seller", the "Management Equity Securities", "Company" or "Group Company" and "Completion" shall be deemed to refer to the seller of DIM RUS, the shares of DIM RUS, DIM RUS and the Separate Russia Completion, respectively together with such other amendments as are required in the context of the Separate Russia Completion:
 - (a) he or she, or, where indicated in Schedule 1, his or her spouse or civil partner or such Management Seller's Manager Holding Company (as applicable), is the sole legal and beneficial owner of (i) as at the date hereof (and not as at Completion), the Options set forth against his or her name in column 3 of Schedule 1 (the Sellers); (ii) as at Completion (and not as at the date hereof) such number of B Ordinary Shares as will, upon the Initial Completion and the consequential exercise of the Options, correspond with the number of Options set forth against his or her name in column 3 of Schedule 1 (the Sellers); and (iii) the Warrants set forth against his, her or its name in column 4 of Schedule 1 (the Sellers);
 - save for any Encumbrance that will be discharged upon Completion, there is no Encumbrance in relation to any such Shares or Warrants;
 - (c) the shares in each Group Company set out in Schedule 2 (The Company and the Subsidiaries) (save that the Management Sellers make no warranty in respect of the Lux Companies):
 - comprise the whole of the issued and allotted share capital of that Group Company; and

- (ii) are, save as set out in Schedule 2 (The Company and the Subsidiaries), directly or indirectly held by the Company or that other Group Company and, save as set out in Schedule 2 (The Company and the Subsidiaries), the Company or such other Group Company is the sole legal and beneficial holder of such shares which are free from all Encumbrances:
- (d) other than this Agreement and the terms of the Management Equity Securities, there is no agreement, arrangement or obligation (whether exercisable now or in the future and whether or not contingent) requiring the allotment or issue or the grant to any person of the right to require the allotment or issue of any share or loan capital or other securities giving rise to a right over or interest in, the capital of the Company or any other Group Company (save that the Management Sellers make no warranty in respect of the Lux Companies);
- (e) he or she or his or her Manager Holding Company (as applicable) has the capacity to enter into this Agreement and any other Transaction Document to which he, she or it (as applicable) is a party and perform his, her or its (as applicable) obligations under it; and
- (f) this Agreement and each other Transaction Document to which the Management Seller and the relevant Management Seller's Manager Holding Company is a party will, when executed, constitute valid and binding obligations of such Management Seller and the relevant Manager Holding Company in accordance with their respective terms.

Business Warranties

- 8.3 Each of the Sellers severally warrants to the Purchaser that at the Offer Date each of the Business Warranties was true and accurate and not misleading by reference to the facts and circumstances then subsisting (save that the Management Sellers make no warranty in respect of the Lux Companies but without prejudice to the Business Warranty at paragraph 4 of Schedule 5 (Business Warranties) in respect of the 2012 Audited Accounts, 2013 Audited Accounts and Management Accounts to the extent they relate to individual Group Companies other than the Lux Companies), provided always that the Purchaser shall not be entitled to, and shall not, make any claim for breach of any of the Business Warranties other than in the circumstances contemplated in Clause 8.5.
- 8.4 In the absence of fraud on the part of the Sellers, the Business Warranties are qualified by reference to those matters Disclosed in the Disclosure Letter.
- 8.5 The Sellers shall be liable for any Seller Claim arising out of the Purchaser's reliance on any Business Warranty subject to those matters that are Disclosed, provided that the maximum aggregate liability of the Sellers in respect of such Seller Claims shall not exceed €2,000,000 and no Seller shall have any liability in respect of any such Seller Claim unless the Purchaser shall have served on the Sellers a written notice of such claim on or prior to the date that is 15 months from the Completion Date (or, with respect to any Seller Claim in respect of DIM RUS, 15 months from the Separate Russia Completion Date (if applicable)).

8.6 If, at any time during the period from (and including) the Offer Date and until (but excluding) the Completion Date a Seller becomes aware of any fact or circumstance that has resulted, results or is reasonably likely to result in (i) any of the Warranties given by that Seller being untrue, inaccurate or misleading as of the Offer Date or (ii) any of the Warranties contained in Clause 8.1 and Clause 8.2 as applicable (disregarding all qualifications and exceptions contained therein relating to materiality) being untrue, inaccurate or misleading in any material respect immediately prior to Completion by reference to the facts and circumstances subsisting at the Completion Date (on the basis that any reference in such Warranties, whether express or implied, to the Offer Date is substituted by a reference to the Completion Date) or (iii) any of the Business Warranties (disregarding all qualifications and exceptions contained therein relating to materiality, including references to "Material Adverse Effect") being untrue, inaccurate or misleading immediately prior to Completion by reference to the facts and circumstances subsisting at the Completion Date (on the basis that any reference in such Warranties, whether express or implied, to the Offer Date is substituted by a reference to the Completion Date), it shall notify the Purchaser as soon as practicable and, in any event, prior to Completion, setting out such details as are then available to it, provided that such Seller shall not be liable for failure to provide such notice except to the extent, and only to that extent, that such Seller's failure causes the Purchaser's losses to be greater than they would have been if the Seller had given notice within the required period.

Purchaser's warranties

- 8.7 The Purchaser warrants to the Sellers that:
 - the Purchaser is a company duly incorporated and in existence under the Laws of the Grand Duchy of Luxembourg;
 - (b) the Purchaser has full power and authority and has taken all corporate action required of it to enter into, deliver and perform its obligations under this Agreement and each other Transaction Document to which the Purchaser is a party;
 - (c) this Agreement and each other Transaction Document to which the Purchaser is a party will, when executed, constitute valid, binding and enforceable obligations of the Purchaser in accordance with their respective terms;
 - (d) the execution and delivery of, and the performance by the Purchaser of its obligations under this Agreement and each other Transaction Document to which the Purchaser is a party will not:
 - conflict with or result in a breach of any provision of the articles of association or other constitutional documents of the Purchaser;
 - (ii) conflict with, result in a breach of or constitute a default (with or without the lapse of time and/or the giving of any notice, certificate, declaration or demand) under any agreement or instrument to which the Purchaser is a party;

- (iii) conflict with or result in a breach of any Law or regulation, or of any order, injunction, judgment or decree of any court or Governmental Entity, that applies to the Purchaser;
- (iv) save as set out in this Agreement, require the Purchaser to obtain any consent or approval of, or give any notice to or make any registration with, any Governmental Entity that has not been unconditionally and irrevocably obtained or made at the date of this Agreement (save for any legal or regulatory entitlement to revoke the same other than by reason of any misrepresentation or misstatement); and
- (e) at Completion, the Purchaser will have the necessary cash resources to meet its obligations under this Agreement immediately available on an unconditional basis.

Purchaser's undertakings

8.8 The Purchaser acknowledges and agrees that, in the absence of fraud, the Purchaser has no rights against and may not make any claim against any employee, director, agent, officer or (except to the extent such adviser has entered into a reliance letter with the Purchaser) adviser of a Seller or any of its Related Persons on whom it may have relied before agreeing to any term of, or entering into, this Agreement or any other Transaction Document, and each and every such person shall be entitled to enforce this Clause 8.8 under the Contracts (Rights of Third Parties) Act 1999.

German Employee Reclassification Indemnity

- 8.9 Subject to Clause 9.14(b), each Seller severally undertakes to indemnify on demand and hold the Purchaser and each member of the Purchaser's Group harmless from and against all German Employee Reclassification Losses suffered or incurred by any member of the Purchaser's Group, provided that no claim under this Clause 8.9 may be made unless written notice of such claim has been given by the Purchaser to the Institutional Seller in accordance with Clause 9.5 by the third anniversary of the Completion Date.
- 8.10 The Purchaser shall, and shall procure that each other member of the Purchaser's Group shall, use its and their respective commercially reasonable endeavours to, in good faith defend, dispute, resist, appeal and compromise each Indemnified Claim following Completion and, subject to applicable Law, afford the Sellers the rights to information, consultation and conduct of claims provided in this Clause 8.10. Without prejudice to the generality of this Clause 8.10, the Purchaser shall, and shall procure that each other member of the Purchaser's Group shall, in any event:
 - (a) take all steps necessary for that purpose (including making and progressing appropriate submissions, notifications and filings with all due diligence) and, to the extent permitted by applicable Law, in consultation with the Institutional Seller;
 - (b) jointly with the Institutional Seller select counsel to assume the defense of the Indemnified Claim, such counsel to be reasonably satisfactory to the Institutional Seller:

- provide promptly (and in any event in accordance with any applicable time limits) all information which is requested or required by the relevant Tax Authority, Governmental Entity or court;
- (d) to the extent permitted by applicable Law, promptly notify the Institutional Seller (and provide copies or, in the case of non-written communications, details) of any communications with or from any Tax Authority, Governmental Entity, court or other person in connection with an Indemnified Claim; provided that: (x) any information that in the Purchaser's reasonable judgment would result in the disclosure of any of the Purchaser's Group's Intellectual Property or Confidential Information or violate any of its obligations with respect to confidentiality, or (y) any privileged information or confidential competitive information of the Purchaser, shall be provided to the Institutional Seller either:
 - redacted only to the extent necessary to preserve such trade secrets, comply with such confidentiality obligations or preserve such privilege; or
 - (ii) on an "outside counsel to outside counsel" basis, in which case it shall not be disclosed by such outside counsel to employees, officers or directors of the Institutional Seller or any member of the Institutional Seller's Group unless express permission is obtained in advance from the Purchaser or its advisers;
- subject to applicable Law, provide the Institutional Seller and the Institutional (e) Seller's Solicitors with a draft of all submissions, notifications, filings and other communications to be submitted to the relevant Tax Authority. Governmental Entity, court or other person in connection with an Indemnified Claim, including any supporting documentation or information requested by the Institutional Seller or the Institutional Seller's Solicitors, at least five Business Days prior to submission and take account of any reasonable comments of the Institutional Seller and the Institutional Seller's Solicitors and other advisers on such drafts prior to their submission; provided that: (x) any information contained in any such submissions, notifications, filings and other communications that in the Purchaser's reasonable judgment would result in the disclosure of any of the Purchaser's Group's Intellectual Property or Confidential Information or violate any of its obligations with respect to confidentiality, or (y) any privileged information or confidential competitive information of the Purchaser, shall be provided to the Institutional Seller either:
 - redacted only to the extent necessary to preserve such trade secrets, comply with such confidentiality obligations or preserve such privilege; or
 - (ii) on a "outside counsel to outside counsel" basis, in which case it shall not be disclosed by such outside counsel to employees, officers or directors of the Institutional Seller or any member of the Institutional Seller's Group unless express permission is obtained in advance from the Purchaser or its advisers;

- (f) allow persons nominated by the Institutional Seller (which may include the Institutional Seller's Solicitors) to make oral submissions with any Tax Authority, Governmental Entity, court or other person in connection with an Indemnified Claim, to the extent permitted by applicable Law;
- (g) regularly review, and consult on, with the Institutional Seller and the Institutional Seller's Solicitors (i) the progress of each Indemnified Claim; and (ii) any notifications or filings to any Tax Authority, Governmental Entity, court or other person in connection with an Indemnified Claim;
- (h) not settle or compromise or make any admission in relation to an Indemnified Claim without the prior written agreement of the Institutional Seller (such agreement not to be unreasonably withheld or delayed); and
- (i) subject to being paid its reasonable third party costs and expenses, give all such information and assistance including access to premises and personnel, and the right to examine and copy or photograph any assets, accounts, documents and records, to the extent the same are not privileged with respect to any Indemnified Claim as the Institutional Seller may reasonably request, provided that the Institutional Seller shall exercise its rights hereunder to mitigate, as far as reasonably practicable, any disruption to the business of the relevant member(s) of the Purchaser's Group,

provided that nothing in this Clause 8.10 shall require any member of the Purchaser's Group, or permit the Institutional Seller, any member of the Institutional Seller's Group or any Management Seller, to act or refrain from acting in such a way as to prejudice the reputation of any member of the Purchaser's Group or have a material adverse impact on the business and operations of any member of the Purchaser's Group, and in the event that the Purchaser reasonably believes that its or any of the Purchaser's Group companies' reputation or business and operations may be so prejudiced or materially adversely impacted, then the Institutional Seller and the Purchaser shall use their respective best endeavours to negotiate in good faith in order to agree the most appropriate course of action; and

- 8.11 If an Indemnified Claim becomes a Substantiated Indemnified Claim then the Sellers shall, subject to Clause 9.14(b) and in accordance with Clauses 8.13, 8.14 and 8.15, pay, or shall procure the payment of, the relevant amount within the period required by such relevant judgment, award or settlement.
- 8.12 The Indemnity Escrow Amount shall be held under the terms of the Indemnity Escrow Letter for disbursement in accordance with Clauses 8.13, 8.14 and 8.15.
- 8.13 As soon as reasonably practicable (and in any event within five Business Days) following an Indemnified Claim becoming a Substantiated Indemnified Claim, there shall be paid out of the Indemnity Escrow Account to the Purchaser a sum equal to the Substantiated Indemnified Claim (together with any interest that has accrued in the Indemnity Escrow Account in respect of such amount) or, if less, the total amount then standing to the credit of the Indemnity Escrow Account.
- 8.14 On the Indemnity Escrow Release Date, after deduction of:

- the aggregate amount of any Non-substantiated Indemnified Claims as at that date (together with any interest that has accrued in the Indemnity Escrow Account in respect of such amount); and
- (b) any Substantiated Indemnified Claims which have not been paid as at that date (together with any interest that has accrued in the Indemnity Escrow Account in respect of such amounts),

all monies remaining in the Indemnity Escrow Account (after payment of all amounts pursuant to Clause 8.14(b)) shall be paid to the Sellers in their Relevant Proportion.

- 8.15 Following the Indemnity Escrow Release Date, as soon as reasonably practicable (and in any event within five Business Days) following any Non-substantiated Indemnified Claim which has been asserted by the Purchaser on or prior to the Indemnity Escrow Release Date becoming a Substantiated Indemnified Claim, there shall be paid out of the Indemnity Escrow Account:
 - (a) to the Purchaser, a sum equal to the Non-substantiated Indemnified Claim which has become a Substantiated Indemnified Claim (together with any interest that has accrued in the Indemnity Escrow Account in respect of such amount) or, if less, the total amount then standing to the credit of the Indemnity Escrow Account; and
 - (b) to the Sellers in their Relevant Proportions, the balance (if any) of the monies remaining in the Indemnity Escrow Account (after deduction of the amount of all remaining Non-substantiated Indemnified Claims (if any, and for the avoidance of doubt excluding any Non-substantiated Indemnified Claims which have been withdrawn or dismissed by a court of competent jurisdiction) together with any interest that has accrued in the Indemnity Escrow Account in respect of such amounts).
- 8.16 The Institutional Seller and the Purchaser shall jointly instruct the Institutional Seller's Solicitors and the Purchaser's Solicitors and the bank holding the Indemnity Escrow Account in accordance with the terms of the Indemnity Escrow Letter to make the appropriate transfers from the Indemnity Escrow Account as required by the provisions of Clauses 8.13, 8.14 and 8.15.

9 LIMITATION OF LIABILITY

General

- 9.1 The Purchaser acknowledges and agrees that:
 - (a) the warranties set out in Clause 8.1 and, subject to Clause 8.5, the Business Warranties, are the only warranties of any kind given by or on behalf of the Institutional Seller which will survive Completion;
 - (b) the warranties set out in Clause 8.2 and, subject to Clause 8.5, the Business Warranties, are the only warranties of any kind given by or on behalf of the Management Sellers which will survive Completion; and

(c) any Seller Claim shall, in the absence of fraud on the part of the relevant Seller, be subject to the provisions of this Clause 9.

Monetary limits

- 9.2 The aggregate liability of any Seller in respect of any and all claims against that Seller under the Transaction Documents (including claims for the recovery of any costs and expenses incurred in connection with such claim) shall be limited to, and shall in no event exceed, 100% of the amount received by such Seller (and, in the case of Christian Duval and François Riston, by their respective Manager Holding Companies) at Completion pursuant to this Agreement together with such amounts as may be subsequently released to such Seller from the Business Warranties Escrow Account, the Purchase Price Adjustment Escrow Account, the Indemnity Escrow Account or the Sri Lanka Escrow Account or subsequently paid to such Seller in connection with the sale of Isabella (Private) Ltd. Or on the Separate Russia Completion Date, less any amounts paid by the Seller to the Purchaser as a result of a Negative Insufficiency (such aggregate liability of a Seller being referred to hereinafter as the "Total Cap"); save that, in the case of any and all breaches of Clause 5.1 by a Management Seller, the Total Cap of such Management Seller in respect of such breaches shall be reduced by an amount equal to all Taxes suffered or incurred by such Management Seller on the payments comprising the Total Cap which, in the case of Christian Duval and François Riston only shall be calculated as if they had sold their Warrants to the Purchaser directly, disregarding their respective Manager Holding Companies (such reduced aggregate liability of a Management Seller being referred to hereinafter as the "Reduced Cap") and, in those circumstances, in addition to the Institutional Seller's Total Cap, the Institutional Seller shall also be liable for an amount in excess of the Management Seller's Reduced Cap up to but not exceeding the Total Cap that would otherwise have applied to such Management Seller.
- 9.3 Subject to the proviso to Clause 9.2, each Seller's liability in respect of any and all claims shall be limited to, and shall in no event exceed, such Seller's Relevant Proportion of such claims.
- 9.4 The Sellers shall have no liability in respect of a claim unless and until the amount of such claim, when aggregated with the amount of any other claim arising out of the same events, facts or circumstances, equals or exceeds, €15,000. In determining whether a monetary limit has been met for the purposes of this Clause 9.4, the amount of any claim shall not include any costs and expenses incurred in connection with such claim.

Notice of claims

9.5 No Seller shall be liable in respect of any Seller Claim or Indemnified Claim unless written notice of such Seller Claim or Indemnified Claim is given by the Purchaser to such Seller, specifying, in such detail as is reasonably available to the Purchaser at the time, the legal and factual basis of the Seller Claim or Indemnified Claim and, if reasonably practicable, the amount likely to be claimed and, in the case of a Seller Claim pursuant to Clause 5, such written notice shall be given on or before 30 June 2015 (or with respect to any Seller Claim in respect of DIM RUS and in the event of a

- Separate Russia Completion, on or before the expiry of a period of twelve months from the date of the Separate Russia Completion).
- 9.6 Where a breach giving rise to a Seller Claim is capable of remedy, the Purchaser shall not be entitled to make any claim (whether for damages or otherwise) in respect of such breach if the breach is remedied within 30 days after notice of the Seller Claim is given under Clause 9.5.
- 9.7 If notice of any Seller Claim is served by the Purchaser under Clause 9.5, no Seller shall be liable in respect of such Seller Claim (if such Seller Claim has not been satisfied or settled) unless:
 - legal proceedings in respect of such Seller Claim are both issued and served within six months after (and excluding) the date on which notice is served; and
 - (b) such proceedings are being and continue to be pursued with reasonable diligence.

Purchaser actions; Events after the date of this Agreement

- 9.8 Nothing in the Transaction Documents is to be construed as relieving the Purchaser from any duty it may have to mitigate any loss or damage.
- 9.9 The Sellers shall not be liable in respect of any Seller Claim to the extent it arises or is increased as a result of:
 - any matter or thing permitted or required to be done pursuant to and in compliance with the Offer Letter, this Agreement or any other Transaction Document at the written request of, or with the written approval of, the Purchaser;
 - (b) any voluntary act, omission or transaction of any member of the Purchaser's Group, or their respective directors, officers, employees or agents or successors in title, after Completion (or in relation to DIM RUS, after the Separate Russia Completion) save (i) in the ordinary course of business or (ii) pursuant to any requirement of Law or binding obligation;
 - (c) the passing of, or any change in (including the interpretation thereof), any Law or administrative practice of any Governmental Entity after the Offer Date, including (without prejudice to the generality of the foregoing) any increase in the rates of Taxation or any imposition of Taxation or any withdrawal of relief from Taxation not actually (or prospectively) in effect at the Offer Date or any Law or published administrative practice or change thereto that has retrospective effect, in each case, other than (i) as a result of fraud or default by a Seller of its obligations under such Law or administrative practice or (ii) any change targeted specifically at countering a tax avoidance scheme to which a Group Company was party;
 - (d) any change in accounting or Taxation policy, bases or practice of any member of the Purchaser's Group introduced or having effect after Completion (or in relation to DIM RUS, after the Separate Russia Completion) (save where such

change is necessary in order to comply with applicable Law or accounting standards in force and applicable as at Completion (or in relation to DIM RUS, as at the Separate Russia Completion) and where the previous policy, bases or practice did not comply with such Law or accounting standards).

Contingent liabilities

9.10 No Seller shall be liable to make a payment in respect of any liability which is contingent unless and until such contingent liability becomes an actual liability and is due and payable.

Provisions

9.11 Except to the extent otherwise expressly provided in this Agreement, no Seller shall be liable under this Agreement in respect of a Seller Claim relating to a breach of Warranty to the extent that the fact, matter, event or circumstance giving rise to such claim or on which it is based is specifically allowed, provided or reserved for in the Completion Statement.

Losses

9.12 The Sellers shall not be liable for any indirect or consequential losses save to the extent that such losses were reasonably foreseeable at the Offer Date.

Corresponding benefits

9.13 In calculating the amount of any Seller Claim or an Indemnified Claim, there shall be taken into account the amount of any net quantifiable financial benefit accruing to the Purchaser's Group or the Group Companies including an amount equal to any Tax benefit (including a Tax reduction or the creation or the increase of carried back or forward Tax losses) obtained and retained or which will be obtained and retained by any member of the Purchaser's Group or the Group Companies, attributable to the fact, event or matter giving rise to such a Seller Claim.

Source of Recovery

- 9.14 The Purchaser's sole source of recovery in respect of:
 - breach of the Business Warranties under Clause 8.5 shall be from the Business Warranties Escrow Amount; and
 - (b) the indemnity contained in Clause 8.9 shall be from the Indemnity Escrow Amount.

Insurances

9.15 The Sellers shall not be liable in respect of any Seller Claim to the extent that the matters in respect of which such claim is made are recovered under a policy of insurance held by a member of the Purchaser's Group, provided that the liability of such Seller shall only be excluded by an amount equal to the sum recovered by the relevant member of the Purchaser's Group under said policy, less (i) any Tax payable by such member of the Purchaser's Group in respect of the sum recovered after taking

account of any Tax relief available in respect of the matter giving rise to the claim (save where such relief has already been taken into account in computing the amount of the claim pursuant to Clause 9.13) and (ii) the costs and expenses of such member of the Purchaser's Group in making such recovery.

Recovery from third parties

- 9.16 If the Sellers have paid an amount in discharge of any Seller Claim, and the Purchaser or any Group Company recovers from a third party a sum that indemnifies or compensates the Purchaser or Group Company (in whole or in part) for the losses which are the subject matter of such Seller Claim, the Purchaser or the relevant Group Company shall pay to the Sellers as soon as practicable after receipt of such sum an amount equal to the lesser of:
 - (a) the sum recovered from the third party less any costs and expenses incurred in obtaining such recovery and any Tax payable by it in respect of the sum recovered; and
 - (b) the amount previously paid by the Sellers to the Purchaser.
- 9.17 The Purchaser shall not be concerned with the apportionment between the Sellers of any amounts paid pursuant to Clause 9.16.

Information memorandum

9.18 The Purchaser acknowledges that the Information Memorandum was provided to it on the basis that neither the Sellers nor any of their advisers makes any representation or warranty as to the accuracy or completeness of such information or accepts any duty of care to the Purchaser in respect of the provision of such information.

No right of rescission

9.19 Without prejudice to the termination rights available to the Purchaser under Clause 4.7, the Purchaser is not entitled to rescind this Agreement or treat this Agreement as terminated by reason of any breach of this Agreement or circumstances giving rise to any Seller Claim, and the Purchaser hereby waives any and all rights of rescission it may have in respect of any such matter (other than any such rights arising from the fraud of any Seller).

Fraud

9.20 None of the limitations contained in this Clause 9 shall apply to any claim against a Seller that arises or is increased, or is delayed, as a result of, fraud or (with the exception of Seller Claims) willful misconduct on the part of any Seller.

10 TAX INFORMATION

10.1 Each Seller severally undertakes to provide such assistance and such information in its possession or under its control as may reasonably be required by the Purchaser in order to negotiate, refute, settle, compromise or otherwise deal with any claim, investigation or enquiry by a Tax Authority regarding a Group Company that relates to income, profits or gains earned, accrued or received (or treated for Tax purposes as

- earned, accrued or received) or any event occurring (or treated for Tax purposes as occurring) on or before Completion (or in relation to DIM RUS, before the Separate Russia Completion) or to any period commencing prior to Completion (or in relation to DIM RUS, prior to the Separate Russia Completion).
- 10.2 The Purchaser undertakes to provide such assistance and such information in its possession or under its control as may reasonably be required by any Seller in order to negotiate, refute, settle, compromise or otherwise deal with any claim, investigation or enquiry by a Tax Authority regarding a Seller or a member of an Institutional Seller's Group that relates to income, profits or gains earned, accrued or received (or treated for Tax purposes as earned, accrued or received) or any event occurring (or treated for Tax purposes as occurring) on or before Completion (or in relation to DIM RUS, before the Separate Russia Completion) or to any period commencing prior to Completion (or in relation to DIM RUS, prior to the Separate Russia Completion).
- 10.3 To enable compliance with the Purchaser's obligations under Clause 10.2, the Purchaser shall and shall cause each Group Company to:
 - (a) retain and maintain all relevant records, reports, returns, declarations, notices, forms or similar statements or documents filed, maintained or submitted or required to be filed, submitted or maintained with respect to any Taxation for the period for which the same are required by applicable Law to be retained and maintained; and
 - (b) upon being given reasonable notice by a Seller, and subject to such Seller giving any confidentiality undertaking reasonably required by the Purchaser, allow such Seller and its officers, employees, agents, auditors and representatives, at the Seller's cost, to:
 - inspect, review and make copies of such records and information for and only to the extent necessary for that purpose; and
 - have reasonable access within five Business Days to any employee, officer, adviser or premises of any Group Company during Normal Business Hours.

11 RESTRICTIONS ON THE SELLERS

- 11.1 The Institutional Seller undertakes to the Purchaser that it shall not, shall procure that no member of the Institutional Seller's Group that is not a Portfolio Company shall, and shall not procure or encourage any member of the Institutional Seller's Group that is a Portfolio Company to, either alone or in conjunction with or on behalf of any other person:
 - (a) during the period of 12 months following Completion, solicit or entice away from the employment of any Group Company, or offer employment to or employ or offer to enter into any contract for services with, any Senior Employee, or induce any such employee to terminate his employment with any Group Company;

- disclose to any other person, or use in any way that may be detrimental to the business of any Group Company, any Confidential Information relating to the Group;
- do or say anything which is reasonably likely or intended to damage the goodwill or reputation of any Group Company or of any business carried on by any Group Company;
- (d) use any trade or business name or distinctive mark, style or logo used by or in the business of any Company or anything intended or likely to be confused with it; or
- (e) knowingly assist any other person to do any of the foregoing things.
- 11.2 Each Management Seller undertakes to the Purchaser that it shall not, either alone or in conjunction with or on behalf of any other person:
 - (a) during the period of 12 months following Completion, solicit or entice away from the employment of any Group Company, or offer employment to or employ or offer to enter into any contract for services with, any Senior Employee, or induce any such employee to terminate his employment with any Group Company. For the avoidance of doubt, nothing in the foregoing shall preclude the dismissal of any Senior Employee carried out in the proper performance of the relevant Management Seller's duties as an employee of the Group;
 - (b) disclose to any other person, or use in any way that may be detrimental to the business of any Group Company, any Confidential Information relating to the Group;
 - do or say anything which is reasonably likely or intended to damage the goodwill or reputation of any Group Company or of any business carried on by any Group Company;
 - (d) use any trade or business name or distinctive mark, style or logo used by or in the business of any Company or anything intended or likely to be confused with it; or
 - (e) knowingly assist any other person to do any of the foregoing things.
- 11.3 The undertakings in Clause 11.1 and Clause 11.2 do not prohibit any Seller or any member of the Institutional Seller's Group from fulfilling any obligation pursuant to this Agreement or any other Transaction Document.
- 11.4 Each Seller agrees that:
 - each undertaking contained in Clause 11.1 and Clause 11.2 shall be construed as a separate undertaking;
 - (b) the undertakings contained in Clause 11.1 and Clause 11.2 are no greater than is reasonable and necessary for the protection of the interests of the Purchaser

- and the Group and have been specifically negotiated and agreed between sophisticated parties; and
- (c) if any such undertaking is held to be void, against the public interest or unlawful or in any way an unreasonable restraint of trade, but would be valid if deleted in part or reduced in application, such undertaking shall apply with such minimum deletion or modification as may be necessary to make it valid and enforceable and the remaining undertakings shall continue to bind the Sellers.

12 ANNOUNCEMENTS AND CONFIDENTIALITY

Announcements

- 12.1 The Purchaser (and/or any member of the Purchaser's Group) and the Institutional Seller shall make the Completion Announcement.
- 12.2 Subject always to Clause 12.3 and Clause 12.4, each of the Purchaser or any member of the Purchaser's Group, and the Institutional Seller may issue, or cause to be issued, any other written public announcement, public statement or press release in connection with the existence or the subject matter of this Agreement or any other Transaction Document, provided that the disclosing Party or person shall (i) use reasonable endeavours to disclose to, or notify the other Party of, the content of such proposed written public announcement, public statement or press release sufficiently in advance of the proposed disclosure or issue and (ii) in good faith consider any reasonable requests and proposals raised by such other Party (if any) in respect of the content of such public announcement, public statement or press release.
- 12.3 Nothing in Clause 12.2 shall restrict the disclosure of the existence or the subject matter of this Agreement or any other Transaction Document by the Institutional Seller to a member of the Institutional Seller's Group or investors (including prospective investors) in funds managed or advised by members of the Institutional Seller's Group, subject to the recipient keeping such information confidential.
- Nothing in Clause 12.2 shall restrict the ability of the Purchaser or any member of the Purchaser's Group to have any communication with the press in connection with the existence or the subject matter of this Agreement or any other Transaction Document or schedule any press conference or conference call with investors or analysts in connection with the existence or the subject matter of this Agreement or any other Transaction Document, provided that the Purchaser shall, and shall procure that each member of the Purchaser's Group shall, use reasonable efforts to restrict disclosure relating to the Institutional Seller to matters which have previously been disclosed by the Institutional Seller or any member of the Purchaser's Group (otherwise than in breach of this Agreement or the Offer Letter).

Confidentiality

12.5 The Confidentiality Agreement shall terminate on Completion (without prejudice to any rights, liabilities or obligations that have accrued prior to termination).

- 12.6 Subject to Clause 12.7, each Party shall treat as strictly confidential and shall not disclose or use any information received or obtained in connection with or as a result of entering into this Agreement or any other Transaction Document that relates to:
 - (a) the provisions of this Agreement or any Transaction Document;
 - the negotiations relating to this Agreement and all other Transaction Documents; or
 - any Group Company's business or financial or other affairs; or
 - (d) any other Party or its Related Persons.
- 12.7 Notwithstanding Clause 12.6, a Party may disclose or use information if and to the extent that:
 - such disclosure or use is required by applicable Law, including the rules of any recognised stock exchange, or by any court or Governmental Entity;
 - such disclosure or use is required for the purpose of any judicial proceedings arising out of this Agreement or any other Transaction Document;
 - disclosure is made to a Tax Authority in connection with the Tax affairs of the disclosing Party;
 - (d) disclosure is made to the professional advisers, auditors or bankers of a Party on a need-to-know basis, provided that the recipient has undertaken to comply with this Clause 12 in respect of such information as if it were a Party;
 - (e) the disclosure is made by the Institutional Seller to a member of the Institutional Seller's Group or investors (including prospective investors) in funds managed or advised by members of the Institutional Seller's Group, subject to the recipient undertaking to comply with this Clause 12 in respect of such information as if it were a Party;
 - (f) the disclosure is made on a confidential basis by the Purchaser or a member of the Purchaser's Group to any prospective lender in connection with the raising of debt to finance the payment of the Consideration or refinance any of the Facilities;
 - (g) the information is or becomes publicly available (other than by breach of this Agreement or the Confidentiality Agreement); or
 - (h) in the case of a Seller, the Purchaser has given its prior written consent or, in the case of the Purchaser, the Institutional Seller has given its prior written consent.
- 12.8 Before any information is disclosed pursuant to Clause 12.7(a), 12.7(b), or 12.7(c), the Party concerned shall (unless prohibited by Law and to the extent reasonably practicable) promptly notify the other Parties to whom the disclosure relates of the circumstances of the disclosure and the information to be disclosed with a view to

providing such other Parties with the opportunity to contest, limit or agree the timing and content of such disclosure.

13 GUARANTEE

- 13.1 The Purchaser's Guarantor unconditionally and irrevocably guarantees to the Sellers (in the case of the Institutional Seller, on the Institutional Seller's behalf and on the behalf of each other member of the Institutional Seller's Group) (the "Guarantee Beneficiaries") the due and punctual payment by the Purchaser of any amounts which it is, or becomes, obliged to pay pursuant to any of the Transaction Documents (the "Purchaser's Guaranteed Obligations").
- 13.2 If and whenever the Purchaser defaults for any reason whatsoever in the performance of any of the Purchaser's Guaranteed Obligations, the Purchaser's Guarantor shall forthwith upon demand unconditionally perform (or procure performance of) and satisfy (or procure the satisfaction of) the Purchaser's Guaranteed Obligations in regard to which such default has been made in the manner prescribed by this Agreement or the relevant Transaction Document so that the same benefits shall be conferred on the relevant Guarantee Beneficiary as such person would have received if the Purchaser's Guaranteed Obligations had been duly performed and satisfied by the Purchaser.
- 13.3 This guarantee is to be a continuing guarantee and accordingly is to remain in force until all the Purchaser's Guaranteed Obligations have been performed or satisfied. This guarantee is in addition to and without prejudice to and not in substitution for any rights or security which the Sellers or any member of the Institutional Seller's Group may now or hereafter have or hold for the performance and observance of the Purchaser's Guaranteed Obligations.
- 13.4 As a separate and independent obligation the Purchaser's Guarantor agrees that any of the Purchaser's Guaranteed Obligations (including any moneys payable) which may not be enforceable against or recoverable from any member of the Purchaser's Group by reason of any legal limitation, disability or incapacity on or of any such members or any other fact, matter, event or circumstance (other than any limitation imposed by this Agreement or any other Transaction Document) shall nevertheless be enforceable against and recoverable from the Purchaser's Guarantor as though the same had been incurred by the Purchaser's Representative and the Purchaser's Guarantor were the sole or principal obligor in respect thereof and shall be performed or paid by the Purchaser's Guarantor on demand.
- 13.5 The liability of the Purchaser's Guarantor under this Clause 13:
 - (a) shall not be released or diminished by any variation of the Purchaser's Guaranteed Obligations or any forbearance, neglect or delay in seeking performance of the Purchaser's Guaranteed Obligations or any granting of time for such performance; and
 - (b) shall not be affected or impaired by reason of any other fact or event which in the absence of this provision would or might constitute or afford a legal or equitable discharge or release or a defence to a guarantor.

14 NOTICES

Service of notices

14.1 Any notice to be given under this Agreement must be in English and in writing, and may be served by hand, by post or airmail (pre-paid and signed for in each case) or by fax to the address or fax number (as applicable) given below, or to such other address or fax number as may have been notified by any Party to the other Parties for this purpose (which shall supersede the previous address or fax number (as applicable) from the date on which notice of the new address or fax number is deemed to be served under this Clause 14).

Institutional Seller:

For the attention of: Deryl Couch

Address: c/o Maples Corporate Services Limited

PO Box 309, Ugland House

South Church Street George Town Grand Cayman KY1-1104 Cayman Islands

Fax number: +1 561 394 0540

Email: dcouch@suncappart.com

Copy to: Institutional Seller's Solicitors, for the attention of

Christopher Field (christopher.field@kirkland.com)

Fax number: +44 (0) 207 469 2001

Management Sellers:

Address: As set out in column 1 of Schedule 1

Fax number: As set out in column 1 of Schedule 1

Email: As set out in column 1 of Schedule 1

Copy to: Mr Eric Delattre

16 Avenue Hoche 75008, Paris France

Fax number: +33 (0) 1 70 72 35 18

Email: eric.delattre@avok.fr

Purchaser:

For the attention of: Joia Johnson

Address: Hanesbrands Inc.

1000 East Hanes Mill Road Winston-Salem, NC 27105

Fax number: +1 336 519 0524

Email: Joia.Johnson@hanes.com

Copy to: Rick Moss

Hanesbrands Inc.

1000 East Hanes Mill Road Winston-Salem, NC 27105

Fax number: +1 336 519 0524

Email: Rick.Moss@hanes.com

Copy to: Simon Jay

Cleary Gottlieb Steen & Hamilton LLP City Place House, 55 Basinghall Street

London EC2V 5EH, England

Fax number: +44 20 7600 1698

Email: sjay@cgsh.com

- 14.2 Any notice served in accordance with Clause 14.1 shall be deemed to have been received:
 - (a) if delivered by hand, at the time of delivery;
 - (b) if sent by post, at 9.30 am on the eighth day after (and excluding) the date of posting; or
 - (c) if sent by fax, at the time of transmission by the sender,
 - (d) if sent by e-mail, at the time of receipt by the recipient, provided that such e-mail is accompanied by any of the other means of service of notice set forth above,

provided that if a notice would otherwise be deemed to have been received outside Normal Business Hours, it shall instead be deemed to have been received at the recommencement of such Normal Business Hours.

14.3 For the purposes of Clause 14.2, "Normal Business Hours" means 9.00 am to 5.30 pm local time in the place of receipt on any day which is not a Saturday, Sunday or

- public holiday in that location. In the case of service on any Party by fax, the place of receipt shall be deemed to be the address specified for service on that Party by post.
- 14.4 In proving receipt of any notice served in accordance with Clause 14.1, it shall be sufficient to show that the envelope containing the notice was properly addressed and either delivered to the relevant address by hand or posted as a pre-paid, signed-for first class or airmail letter or that the fax was sent to the correct number and a confirmatory transmission report was received.
- 14.5 A notice given under this Clause 14 shall not be validly served if sent by email.
- 14.6 This Clause 14 shall apply to the service of any proceedings or other documents in any legal action.

Management Sellers' Representative

- 14.7 Save as expressly required by this Agreement or mandatory applicable Law, the Management Sellers' Representative shall act on behalf of the Management Sellers in respect of this Agreement and the Transaction Documents and, in particular:
 - (a) any notice, consent, election or request to be made or given by the Management Sellers to the Purchaser under this Agreement may be made or given by the Management Sellers' Representative on behalf of all the Management Sellers, and the Purchaser is not required to have regard to any notice served by or on behalf of the Management Sellers (or any of them) by any other person;
 - (b) any notice to be given by the Purchaser to the Management Sellers under this Agreement shall be validly served on all the Management Sellers (or any of them) if it is served on the Management Sellers' Representative in accordance with Clause 14.1; and
 - (c) any variation to this Agreement pursuant to Clause 15.15 may be signed on behalf of the Management Sellers by the Management Sellers' Representative.
- 14.8 The Management Sellers may at any time, following a decision taken by the Management Sellers representing at least 50% of the aggregate number of Management Equity Securities held by the Management Sellers immediately prior to Initial Completion, appoint a different person to act as the Management Sellers' Representative by written notice to the Purchaser, which appointment shall take effect from the date on which such notice is deemed to be served pursuant to Clause 14.2 or such later date as may be specified in the notice.
- 14.9 The Management Sellers' Representative shall not be liable to any Management Seller for any claim whatsoever arising from any act or omission undertaken by him in his capacity as the Management Sellers' Representative save in the case of fraud or wilful default, and the Management Sellers' Representative shall be entitled to enforce this Clause 14.9 under the Contracts (Rights of Third Parties) Act 1999.

15 GENERAL

Further assurances

- 15.1 On request by any Party, each Party shall, as soon as reasonably practicable at the requesting Party's cost and insofar as such Party is reasonably able, do or procure the doing of all such acts and execute or procure the execution of all such documents (in a form reasonably satisfactory to the requesting Party) as the requesting Party may reasonably consider necessary or appropriate to carry this Agreement into effect and to give the requesting Party the full benefit of it.
- 15.2 Each Party agrees that it will promptly provide all information requested in writing by any other Party reasonably required to enable the other Parties to comply with their respective obligations under the Proceeds of Crime Act 2002 and the Money Laundering Regulations 2007 (No. 2157) (or analogous legislation in any applicable jurisdiction whether such obligations apply prior to Completion or thereafter.

Costs

- 15.3 Unless expressly provided otherwise in this Agreement, each Party shall bear its own costs and expenses in relation to the negotiation, preparation, execution and implementation of the Transaction Documents, but this Clause 15.3 shall not prejudice any Party's right to seek to recover costs in any litigation or other dispute resolution procedure arising in connection with any Transaction Document.
- 15.4 Each Seller agrees that the Institutional Seller is authorised to instruct the Institutional Seller's Solicitors to deduct an amount equal to (a) each Seller's share (pro-rata to the number of Shares and/or Warrants (as the case may be) to be sold by it pursuant to this Agreement) of any costs, fees and expenses incurred by the Sellers in connection with the transactions contemplated by this Agreement (including, any Transaction cash bonuses to be paid by the Institutional Seller) and (b) each relevant Management Seller's portion of the Employment Tax Liability, in each case when allocating the aggregate amounts due to the relevant Seller under this Agreement.
- 15.5 The Purchaser shall bear all transfer taxes, stamp duties and registration duties payable as a result of the transactions contemplated by this Agreement, and shall be responsible for arranging the payment of such taxes and duties.

Gross-up

- 15.6 Subject to Clause 3.11, all sums payable under this Agreement shall be paid free and clear of all deductions or withholdings whatsoever, save only as required by Law.
- 15.7 Where a sum payable under this Agreement in respect of any indemnity, compensation or reimbursement provision (but excluding, for the avoidance of doubt, any part of the Consideration) is required by Law to be made subject to a deduction or withholding, the payer shall pay to the recipient such additional amount as will result in the receipt by the recipient of a net amount equal to the full amount which would have been received had no such deduction or withholding been required to be made.

15.8 If the recipient of a payment made under this Agreement receives a credit against, relief or remission for, or refund or repayment of, any Taxation payable by it or similar relief or benefit by reason of any deduction or withholding for or on account of Taxation then it shall reimburse to the payer such part of any additional amounts paid to it pursuant to Clause 15.7 as the recipient of the payment certifies to the payer will leave it (after such reimbursement) in no better and no worse position than it would have been if the payer had not been required to make such deduction or withholding.

Termination

- 15.9 If this Agreement is terminated pursuant to Clause 4 the Parties shall have no further obligations under this Agreement, provided that:
 - (a) the Surviving Provisions shall survive termination; and
 - (b) (for the avoidance of doubt) termination shall be without prejudice to any rights, liabilities or obligations that have accrued prior to termination, or to any other rights or remedies available under this Agreement.
- 15.10 Save for the termination provisions set out in Clause 4, no Party is entitled to terminate this Agreement.

Assignment

- 15.11 This Agreement shall be binding on each of the Parties and their respective successors and permitted assigns.
- 15.12 No Party may assign, hold on trust, transfer, charge or otherwise deal with all or any part of its rights or obligations under this Agreement without the prior written consent of the other Parties, provided that:
 - (a) this Agreement and the benefits arising under it may be assigned in whole or in the part by the Institutional Seller to any member of the Institutional Seller's Group (provided that if such assignee ceases to be a member of the Institutional Seller's Group, this Agreement and the benefits arising under it shall automatically transfer back to the Institutional Seller immediately prior to such cessation);
 - (b) at any time prior to Completion, this Agreement and the benefits arising under it may be assigned in whole or in the part by the Purchaser to any member of the Purchaser's Group (provided that if such assignee ceases to be a member of the Purchaser's Group prior to Completion, this Agreement and the benefits arising under it shall automatically transfer back to the Purchaser immediately prior to such cessation);
 - (c) this Agreement and the benefits arising under it may be assigned in whole or in the part by the Purchaser to any person to whom the Purchaser transfers any of the Shares and/or Warrants; and

- (d) this Agreement and the benefits arising under it may be assigned or charged in whole or in part by the Purchaser to its financial lenders or banks as security for any financing or refinancing or other banking or related facilities entered into for the purposes of the transactions contemplated by this Agreement and such benefits may further be assigned to any other financial institution by way of security for the borrowings made under such Agreement or to any person entitled to enforce any such security.
- 15.13 In the case of an assignment pursuant to Clause 15.11 above, the liability of any Party to such an assignee shall not be greater than it would have been had such assignment not taken place, and all the rights, benefits and protections afforded to a Party shall continue to apply to the benefit of that Party as against the assignee as they would have applied as against the assigning Party.
- 15.14 Any purported assignment, declaration of trust, transfer, sub-contracting, delegation, charging or dealing in contravention of Clause 15.11 shall be ineffective.

Variation

15.15 Save as permitted pursuant to Clause 14.7, no variation of this Agreement shall be valid unless it is in writing and signed by or on behalf of each Party.

Rights of third parties

15.16 Except as expressly stated in this Agreement, this Agreement does not confer any rights on any person or party under the Contracts (Rights of Third Parties) Act 1999. The Parties may rescind, vary or terminate this Agreement in accordance with its terms without the consent of or notice to any person on whom such rights are conferred.

Entire agreement

- 15.17 The Transaction Documents constitute the whole agreement between the Parties relating to the Transaction to the exclusion of any terms implied in Law that may be excluded by contract. They supersede and extinguish any and all prior discussions, correspondence, negotiations, drafts, arrangements, understandings or agreements relating to the Transaction.
- 15.18 Each Party agrees and acknowledges that:
 - it is entering into the Transaction Documents in reliance solely on the statements made or incorporated in them;
 - it is not relying on any other statement, representation, warranty, assurance or undertaking made or given by any person, in writing or otherwise, at any time prior to the date of this Agreement ("Pre-Contractual Statement");
 - (c) except as expressly provided in this Agreement, it is entering into this Agreement solely in reliance on its own commercial assessment and investigation and advice from its own professional advisers; and

- (d) the other Parties are entering into this Agreement in reliance on the acknowledgements given in this Clause 15.18.
- 15.19 No Party shall have any liability whatsoever for any Pre-Contractual Statement, whether in contract, in tort, under the Misrepresentation Act 1967 or otherwise.
- 15.20 It is agreed that the only liability of any Party in respect of those statements, representations, warranties, assurances and undertakings made or given by it and set out or incorporated in this Agreement shall be for breach of contract.
- 15.21 The Purchaser irrevocably and unconditionally waives (and such waiver shall constitute a valid and effective waiver for the purposes of Clause 15.24) any right it may have to claim damages in respect of any breach of, or to rescind this Agreement by reason of, any Business Warranty and the Purchaser agrees and acknowledges that:
 - any purported claim under, or in respect of, any Business Warranty shall be unenforceable against any Seller or any member of the Institutional Seller's Group (other than in the circumstances contemplated by Clause 8.5); and
 - (b) the Purchaser is, and shall be, from and including the Completion Date, prohibited from bringing any claim under, or in respect of, any Business Warranty (other than in the circumstances contemplated by Clause 8.5).
- 15.22 This entire agreement Clause does not limit or exclude any liability for fraud.

Inconsistency

15.23 If there is any inconsistency between the provisions of this Agreement and those of any other Transaction Document, the provisions of this Agreement shall prevail.

Waiver

- 15.24 Any waiver of any term or condition of this Agreement, waiver of any breach of any term or condition of this Agreement, or waiver of, or election whether or not to enforce, any right or remedy arising under this Agreement or at Law, must be in writing and signed by or on behalf of the person granting the waiver, and no waiver or election shall be inferred from a Party's conduct.
- 15.25 Any waiver of a breach of any term or condition of this Agreement shall not be, or be deemed to be, a waiver of any subsequent breach.
- 15.26 Failure to enforce any provision of this Agreement at any time or for any period shall not waive that or any other provision or the right subsequently to enforce all provisions of this Agreement.
- 15.27 Failure to exercise, or delay in exercising, any right or remedy shall not operate as a waiver or be treated as an election not to exercise such right or remedy, and single or partial exercise or waiver of any right or remedy shall not preclude its further exercise or the exercise of any other right or remedy.

Severance

- 15.28 If any provision of this Agreement is held to be invalid or unenforceable by any court or Governmental Entity, all other provisions of this Agreement will remain in full force and effect and will not in any way be impaired.
- 15.29 If any provision of this Agreement is held to be invalid or unenforceable but would be valid or enforceable if some part of the provision were deleted or amended, the provision in question will apply with the minimum modifications necessary to make it valid and enforceable.

Counterparts and duplicates

15.30 This Agreement may be executed in any number of counterparts, but shall not be effective until each Party has signed at least one counterpart. Each counterpart constitutes an original, and all the counterparts together constitute one and the same agreement. If this Agreement is executed in duplicate, each duplicate constitutes an original.

Governing Law

15.31 This Agreement and any non-contractual obligations arising out of or in connection with it (including any non-contractual obligations arising out of the negotiation of the transaction contemplated by this Agreement) are governed by and shall be construed in accordance with the Laws of England and Wales.

Jurisdiction

15.32 The Parties irrevocably agree that the courts of England shall have exclusive jurisdiction to settle any dispute or claim that arises out of or in connection with this Agreement (including a dispute relating to any non-contractual obligation arising out of or in connection with either this Agreement or the negotiation of the transaction contemplated by this Agreement).

Agent for service of process

- 15.33 Each of the non-UK Party(ies) undertakes to ensure that at all times a person with an address in England is appointed as its process agent to receive on its behalf service of any proceedings in respect of any dispute or claim that arises out of or in connection with this Agreement or its subject matter or formation (including non-contractual disputes or claims) (the "Process Agent"). Such service shall be deemed completed on delivery to the Process Agent, whether or not it is forwarded to or received by each of the non-UK Party(ies).
- 15.34 At the date of this Agreement, each of the non-UK Party(ies) (other than the Purchaser and the Purchaser's Guarantor) has appointed the persons set out in column (5) of Schedule 1 (The Sellers) in England as its Process Agent. The Purchaser and the Purchaser's Guarantor have each appointed, Worldwide Corporate Advisors LLP (WCA), 32 Threadneedle Street, The City, London, UK, EC2R 8AY as their respective Process Agents. If such person ceases to be able to act as process agent or no longer has an address in England, each of the non-UK Party(ies) shall immediately

- appoint a replacement Process Agent and deliver to the other Party a notice setting out the new Process Agent's name and address together with a copy of the new Process Agent's acceptance of its appointment.
- 15.35 Each of the non-UK Party(ies) irrevocably agrees that any proceedings or document served on the Process Agent will be validly served if delivered in accordance with Clause 14.
- 15.36 Nothing contained in this Agreement shall affect the right to serve process in any other manner permitted by Law.

This share purchase agreement is executed by	y the	Parties	on	the da	ate	written	at t	he	beginn	iing
of this Agreement.										

EXECUTED by)
SLB BRANDS HOLDING, LTD)

EXECUTED)
by CHRISTIAN DUVAL)

EXECUTED)
by FABIO LUXI)

EXECUTED)
by FRANCOIS RISTON)

EXECUTED)
by XAVIER LÉPINGLE)

EXECUTED)
by MARCEL NARDELLI)

EXECUTED
by ANTONIO LANDOLO

EXECUTED by) /s/ Richard D. Moss
MFB INTERNATIONAL) Richard D. Moss
HOLDINGS S.À R.L.) Authorized Representative

EXECUTED by)	/s/ Joia M. Johnson
HANESBRANDS INC.)	Joia M. Johnson
		Chief Legal Officer, General
		Counsel and Corporate Secretary

EXECUTED by SOCIÉTÉ CIVILE DE LA DUNE)

EXECUTED by (COUNTY OF THE PROPERTY OF THE PRO

List of Schedules Omitted from the Agreement

Pursuant to Item 601(b)(2) of Regulation S-K, certain schedules to this Agreement have been omitted. A list briefly identifying the contents of the omitted schedules is set forth below. The Registrant agrees to furnish supplementally a copy of any omitted exhibits or schedules to the Securities and Exchange Commission upon request.

Schedule 1 The Sellers

Schedule 2 The Company and the Subsidiaries

Schedule 6 (Annex) Form and Content of Completion Statement

Schedule 8 Sri Lanka Sale

SCHEDULE 3

Conduct of Business prior to Completion

Without prejudice to the generality of Clause 5.1, the Institutional Seller shall exercise all corporate powers available to it to, and each Management Seller shall use his best endeavours to, procure that, between the time of entry into the Offer Letter and Completion, no Group Company does any of the following things without the prior consent of the Purchaser; provided that, in respect of matters set out in paragraphs 3-7 (inclusive) and 13-26 (inclusive) only, consent shall not be unreasonably withheld, conditioned or delayed and where the Purchaser has failed to respond for a period of six Business Days following a request for consent from the Institutional Seller under those paragraphs, such consent shall be deemed to have been given. For the purposes of this Schedule 3, notice requesting consent shall be validly served only if it is both: (i) delivered by fax in accordance with Clause 14 and (ii) sent by e-mail to the following persons:

Name:	E-mail:
Gerald Evans	Gerald.Evans@hanes.com
Rick Moss	Rick.Moss@hanes.com
Joia Johnson	Joia.Johnson@hanes.com

- Make any substantial change in the nature or organisation of its business;
- Discontinue or cease to operate all or a material part of its business, where "material" shall mean such part of its business as has sales in excess of €10,000,000;
- Assign, licence, charge or abandon or otherwise dispose of any Intellectual Property other than on arm's length terms;
- Incur any additional borrowings (other than by bank overdraft or similar facility in the ordinary course of business and within the limits subsisting at the Offer Date) or incur any other indebtedness, in each case in excess of €15,000,000;
- Make any loan (other than the granting of trade credit in the ordinary and usual course of trading in accordance with its normal practice) to any person, other than to another Group Company;
- Acquire or dispose of any asset with a book or market value in excess of €5,000,000 other than with respect to any acquisition or disposal in the ordinary and usual course of trading and on normal arm's length terms;
- Create or grant any Encumbrance (other than any Permitted Encumbrance) over any
 of its assets or undertaking or any guarantee, indemnity or other agreement to secure
 any obligation of any person other than another Group Company;

- Amend its articles of association or other constitutional document, or pass any resolution that is inconsistent with their provisions;
- Pass any resolutions in a general meeting or class meeting of shareholders or by way of shareholders' written resolution;
- Create, allot or issue, or grant any option over or right to subscribe for or purchase any share capital or loan capital or other securities of any Group Company;
- Repay, redeem, reduce or repurchase any share capital or loan capital or other securities of any Group Company other than the redemption or repurchase of the Options held by Xavier Lepingle pursuant to his ceasing to be an employee of the Group;
- Declare, make or pay any dividend or other distribution to shareholders;
- Make any capital commitment which individually exceeds €5,000,000 or which, together with all other capital commitments entered into between the Offer Date and Completion, exceeds €15,000,000 in aggregate;
- 14. Enter into or terminate, or amend, any Material Contract;
- Offer to engage any new employee or consultant at an annual gross base salary or fee per employee or consultant (on the basis of full time employment or consultancy) in excess of €125,000 per annum;
- 16. Dismiss any employee or consultant earning a gross annual base salary in excess of €125,000 per annum, other than for cause or unless not to do so would materially damage its business or dismiss 10 or more employees, or announce any restructuring of the business of any Group Company which involves dismissal or reclassification of employees;
- Materially amend, including by increasing emoluments (including pension contributions, bonuses, commissions and benefits in kind), the terms of employment of any Senior Employee;
- Discontinue or amend, any pension scheme or commence winding up or termination
 of any pension scheme or cause it to cease to admit new members or communicate to
 employees a plan, proposal or intention to discontinue, amend, wind up, terminate or
 cease to admit;
- Commence, compromise, settle, release or discharge any Proceedings except for routine debt collection where the amount claimed does not exceed €1,000,000;
- 20. In relation to any Real Property:
 - (a) carry out any material change of use of, such Real Property;
 - (b) terminate or serve any notice to terminate, surrender or accept any surrender of or waiver of the terms of any lease, tenancy, licence or other contract which, in each case, is material;

- agree to any new rent or fee payable under any lease, tenancy, licence or other contract which, in each case, is material;
- enter into or vary any lease, tenancy, licence or other contract which, in each case, is material;
- sell, convey, transfer, assign or charge any Real Property or grant any rights or easements over any Real Property or enter into any covenant or grant any other Encumbrance (other than Permitted Encumbrance) affecting any Real Property;
- Knowingly do anything to make any of its policies of insurance which are material to the business of the Group taken as a whole void or voidable;
- Change its accounting reference date or make any other changes to its accounting practices or policies other than as may be required to ensure compliance with applicable Law and accounting standards;
- 23. Take steps with the intention of changing its residence for Tax purposes;
- 24. Acquire any securities of, or other interest in, any person;
- Enter into, or amend, any contract, understanding or arrangement with any member of the Institutional Seller's Group or any Management Seller or any of its Related Persons other than on arm's length terms;
- Enter into, amend in a material respect or terminate any collective bargaining agreement;
- 27. Make (other than in accordance with past practice), change or rescind any election, claim, surrender or disclaimer relating to Tax except where the same is required pursuant to applicable Law, settle or compromise any proceedings relating to Tax or amend any previously submitted Tax return or filing; or
- 28. Authorise or agree to do or take any of the foregoing acts or matters.

SCHEDULE 4

Completion Obligations

On Initial Completion, the Institutional Seller shall deliver to the Purchaser (or make available to the Purchaser's reasonable satisfaction):	
(a) a copy of the Settlement Agreement executed by each party thereto; and	Fundamental Obligation
(b) the Institutional Shareholder Release, duly executed by the Institutional Seller;	Fundamental Obligation
(c) a transfer declaration form with respect to the Institutional Seller's Shares, duly executed by the Institutional Seller and countersigned by the Company in favour of the Purchaser and share certificates for such Shares (to the extent such Shares are in a certificated form) in the name of the Institutional Seller;	Fundamental Obligation
(d) a written declaration of the Company in its capacity as the holder of the class B shares in the Company (held in treasury) that it waives its rights to participate in the Transaction pursuant to articles 2 (Tag-along right) and 3 (Drag-along right) of the Articles, as holder of the class B shares in the Company;	Fundamental Obligation
(e) an updated copy of the share register of the Company evidencing the transfer of the such number of B Ordinary Shares as will, upon the Initial Completion and the consequential exercise of the Options, correspond with the number of Options set against the relevant Seller's name in column (3) of Schedule 1 from the Company (held in treasury) to the Management Sellers effective as of the date of Initial Completion;	Fundamental Obligation
(f) documentary evidence in the form of minutes or otherwise, indicating that the general meeting of shareholders of the Company has consented to the transfer of the Warrants pursuant to the terms of this Agreement;	Fundamental Obligation
(g) a copy of the Management Equity Exercise Agreement executed by each party thereto.	Fundamental Obligation

satisfaction):	
(a) a written notice by each holder of an Option to the Company and acknowledged by the Company, that the Option is exercised by the participant and confirmation that the Company accepts as payment of the exercise price the cheque issued by the relevant holder of the Options and provided therewith as sufficient consideration for the exercise of the Option on the terms of the Option;	Fundamental Obligation
(b) a written notice by each holder of an Option and countersigned by the Company, waiving any share sale restrictions on exercise of the Options as may be contained in the Option terms and conditions; and	Fundamental Obligation
(c) a transfer declaration form with respect to such Person's B Ordinary Shares and Warrants (as applicable), duly executed by the applicable Management Seller and the relevant Manager Holding Company in favour of the Purchaser and share certificates for such Shares (to the extent such Shares are in a certificated form) in the name of the relevant Management Seller;	Fundamental Obligation
(d) a written notice by each holder of an Option and countersigned by the Company, waiving any share sale restrictions on exercise of the Options as may be contained in the Option terms and conditions; and	
3. On Subsequent Completion, the Institutional Seller shall deliver to the Purchaser (or make available to the Purchaser's reasonable satisfaction) a waiver notice executed by the Institutional Seller waiving any rights it may have as registered shareholder of the class A shares in the capital of the Company pursuant to article 1 (pre-emption rights) of the Articles.	Fundamental Obligation
4. On Completion, the Sellers shall deliver to the Purchaser (or make available to the Purchaser's reasonable satisfaction):	
(a) the share register of the Company written up to Completion, including the transfer of the Institutional Seller's Shares and the Management Seller's Shares to the Purchaser with effect from Completion;	Fundamental Obligation
(b) a copy of the notification to the Company notifying it of the transfer of the Institutional Seller's Shares and the Management Sellers' B Ordinary Shares to the Purchaser with effect from Completion;	

(c) a copy of the Warrant register of the Company written up to Completion;	
 (d) all registers of the Company duly written up to Completion, including updated registers to reflect the repayment or redemption of the Existing Shareholder Debt; 	
 (e) share certificates (to the extent the shares are in a certificated form) and share registers written up to Completion, in respect of each Group Company; 	
(f) a duly executed termination deed in the agreed form in respect of the Company Shareholders' Agreement and the liquidity agreements entered into by each of the holders of the Warrants, including a confirmation that DBA Lux Holding S.A. Incentive Share Option Plan 2006, Plan Rules are no longer of any force or effect.;	Fundamental Obligation
(g) the Tikehau Deed of Release, duly executed by each party thereto;	Fundamental Obligation
(h) the Transaction Documents to which the relevant Seller is a party duly executed by such Seller;	
 (i) a letter of resignation in the agreed form in respect of the following persons that the Institutional Seller has appointed as a director: Clarence Terry, Lynn Skillen, Isabelle Arker, Anita Lyse, Laura Spitoni and Marcel Stephany; 	
(j) to the extent requested by the Purchaser in writing at least 10 Business Days' before Completion and the Sellers having used reasonable endeavours to obtain the same, the written resignations of the auditors of the Group Companies to take effect on the Completion Date; and	
(k) evidence that it is authorised to execute this Agreement and any other Transaction Document to which it is a party.	
5. On Completion, each Seller shall use all corporate powers available to such Seller to:	
(a) procure the passing of a board resolution of the Company approving the Transaction and noting the delivery of the notification referred to in paragraph 4(b) of this Schedule;	

(b) procure the passing of board or shareholder resolutions of each Group Company (as applicable pursuant to relevant Law): (i) accepting the resignations referred to in paragraphs 4(i) above; and (ii) appointing such persons as the Purchaser may nominate at least 10 Business Days prior to Completion to be the director(s) and secretary (if any) of each Group Company; and (c) procure that the registers in respect of the Existing Shareholder Debt are written up to reflect the repayment or redemption of such Existing Shareholder Debt. Purchaser's Obligations 6. On Completion, the Purchaser shall: (a) deliver a counterpart of the transfer declaration form referred to in paragraph 1(c) duly executed by the Purchaser as transferee; (b) deliver a counterparts of the transfer declaration forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option; (e) subject to Clause 3.3, pay the Consideration less the			
(ii) appointing such persons as the Purchaser may nominate at least 10 Business Days prior to Completion to be the director(s) and secretary (if any) of each Group Company; and (c) procure that the registers in respect of the Existing Shareholder Debt are written up to reflect the repayment or redemption of such Existing Shareholder Debt. Purchaser's Obligations 6. On Completion, the Purchaser shall: (a) deliver a counterpart of the transfer declaration form referred to in paragraph 1(c) duly executed by the Purchaser as transferee; (b) deliver a counterparts of the transfer declaration forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company, referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;		resolutions of each Group Company (as applicable pursuant	
nominate at least 10 Business Days prior to Completion to be the director(s) and secretary (if any) of each Group Company; and (c) procure that the registers in respect of the Existing Shareholder Debt are written up to reflect the repayment or redemption of such Existing Shareholder Debt. Purchaser's Obligations 6. On Completion, the Purchaser shall: (a) deliver a counterpart of the transfer declaration form referred to in paragraph 1(c) duly executed by the Purchaser as transferee; (b) deliver a counterparts of the transfer declaration forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;			
Shareholder Debt are written up to reflect the repayment or redemption of such Existing Shareholder Debt. Purchaser's Obligations 6. On Completion, the Purchaser shall: (a) deliver a counterpart of the transfer declaration form referred to in paragraph 1(c) duly executed by the Purchaser as transferee; (b) deliver a counterparts of the transfer declaration forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, (ii) appointing new directors to the board of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;		nominate at least 10 Business Days prior to Completion to be the director(s) and secretary (if any) of each Group	
6. On Completion, the Purchaser shall: (a) deliver a counterpart of the transfer declaration form referred to in paragraph 1(e) duly executed by the Purchaser as transferee; (b) deliver a counterparts of the transfer declaration forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, (ii) appointing new directors to the board of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;		Shareholder Debt are written up to reflect the repayment or	
(a) deliver a counterpart of the transfer declaration form referred to in paragraph 1(c) duly executed by the Purchaser as transferee; (b) deliver a counterparts of the transfer declaration forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, (ii) appointing new directors to the board of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;	Purcha	ser's Obligations	
referred to in paragraph 1(c) duly executed by the Purchaser as transferee; (b) deliver a counterparts of the transfer declaration forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, (ii) appointing new directors to the board of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;	6.	On Completion, the Purchaser shall:	
forms referred to in paragraph 2 duly executed by the Purchaser as transferee; (c) deliver a written resolution of the Purchaser, in its capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, (ii) appointing new directors to the board of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;		referred to in paragraph 1(c) duly executed by the Purchaser	
capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, (ii) appointing new directors to the board of the Company, and (iii) changing the registered office of the Company; (d) deliver a written resolution of the board of directors of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;		forms referred to in paragraph 2 duly executed by the	
of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms and conditions of the Option;		capacity as sole shareholder of the Company and each relevant Group Company (post Completion) resolving: (i) to accept the resignation of the directors of the Company referred to in paragraph 4(i) and granting such directors discharge from their duties as directors of the Company, (ii) appointing new directors to the board of the Company, and	
(e) subject to Clause 3.3, pay the Consideration less the Fundamental		of the Company (post Completion and change of board) recording: (i) the acceptance of the exercise of the Options and acknowledging transfer of the applicable Shares under the Options, (ii) acknowledging the transfer of the Warrants and the update to the Warrant register and (iii) confirmation that payment of the Option Exercise Price pursuant to the terms of this Agreement was proper payment on the terms	
		(e) subject to Clause 3.3, pay the Consideration less the	Fundamental

Option Exercise Consideration, the secondary Employment Tax Liability, the Business Warranties Escrow Amount, the Purchase Price Adjustment Escrow Amount, the Indemnity Escrow Amount and the amount, if any, required to be paid into the Sri Lanka Escrow Account under Clause 3.2 and Part A of Schedule 8 (Sri Lanka Sale), to the Sellers by telegraphic transfer of immediately available funds to the Institutional Seller's Solicitors' Account (which payment shall be an absolute discharge to the Purchaser who shall not be concerned to see to the application of any such amount thereafter);	Obligation
(f) pay the Business Warranties Escrow Amount into the Business Warranties Escrow Account to be held on the terms of the Business Warranties Escrow Letter;	Fundamental Obligation
(g) pay the Purchase Price Adjustment Escrow Amount into the Purchase Price Adjustment Escrow Account to be held on the terms of the Purchase Price Adjustment Escrow Letter;	Fundamental Obligation
(h) pay the Indemnity Escrow Amount into the Indemnity Escrow Account to be held on the terms of the Indemnity Escrow Letter;	Fundamental Obligation
(i) pay the amount, if any, under Clause 3.2 and Part A of Schedule 8 (Sri Lanka Sale) into the Sri Lanka Escrow Account to be held on the terms of the Sri Lanka Escrow Letter;	Fundamental Obligation
(j) pay the secondary Employment Tax Liability to the Company;	Fundamental Obligation
(k) procure that the Company redeems the Existing Shareholder Debt (including for the avoidance of doubt, all accrued interest) in full by telegraphic transfer of immediately available funds equal to the Existing Shareholder Debt (including for the avoidance of doubt, all accrued interest) Redemption Amount to the Institutional Seller's Solicitors' Account (which payment shall be an absolute discharge to the Purchaser who shall not be concerned to see to the application of any such amount thereafter or be answerable for the loss or misapplication of such sum);	Fundamental Obligation
(l) procure that DBApparel SAS pays the Tikehau Redemption Amount to the holders of the Tikehau Bonds;	Fundamental Obligation

(m) deliver the Transaction Documents to which the Purchaser and the Purchaser's Guarantor is a party duly executed by the Purchaser and the Purchaser's Guarantor; and	
(n) deliver or make available to the Institutional Seller evidence that the Purchaser is authorised to execute this Agreement and any other Transaction Document to which it is a party.	
(o) pay, or procure the payment by the relevant Group Company of, all Unsettled Transaction Expenses (including, for the avoidance of doubt, accrued but unpaid consulting, management, advisory or monitoring fees due to members of the Institutional Seller's Group) referred to in the Spreadsheet in the Annex to Schedule 6 (Post Completion Adjustments)	Fundamental Obligation
Purchaser's Guarantor's Obligations	
7. On Completion, the Purchaser's Guarantor shall deliver or make available to the Institutional Seller evidence that it is authorised to execute this Agreement and any other Transaction Document to which it is a party.	Fundamental Obligation

SCHEDULE 5

Business Warranties

For the purposes of this Schedule 5, any reference to the awareness of the Sellers or any of them shall be deemed to refer to the awareness of any of the Management Sellers or the Institutional Seller, or in the case of the Lux Companies only, to the awareness of the Institutional Seller, in each case, having made reasonable enquiries.

Group Structure and Constitutional Matters

- (a) No Group Company is a party to any contract or binding arrangement which is material (in the context of the Group) and which is a joint venture, consortium, partnership or profits (or loss) sharing agreement.
- (b) Complete and accurate copies of the articles of association or other equivalent constitutional documents in the relevant jurisdiction of each Group Company are set out in the Disclosure Documents.
- (c) The statutory books of each Group Company are up-to-date and have been maintained in all material aspects in accordance with applicable Law and no written notice or allegation that any of them is inaccurate or should be rectified has been received.
- (d) No Group Company has any branch, division or place of business outside the jurisdiction in which it is incorporated.
- (e) No Group Company is or has agreed to become a member of any joint venture, consortium, partnership or unincorporated association (other than a recognised trade association).
- (f) Each Group Company has at all times carried on its business and affairs in all material respects in accordance with its constitutional documents from time to time in force.

Agreements With Related Persons

There is not outstanding any agreement or binding arrangement to which any Group Company is a party and in which any Seller, their respective Related Persons or any person connected with the Sellers or their Related Persons is directly or indirectly interested, whether relating to the business of the Group or otherwise, other than any employment agreements between the Group Companies and the Management Sellers or any agreement between Group Companies.

Powers of Attorney

No Group Company has given a power of attorney or any other authority (express, implied or ostensible or as agent or otherwise) to any person to enter into any contract or commitment or to do anything on its behalf that is still outstanding or effective, other than to appropriate employees to enter into routine trading contracts in the normal course of their duties that involve or are reasonably likely to involve an

aggregate consideration payable by or to that Group Company of less than €100,000 or to patent attorneys or agents or trademark agents for routine prosecution or maintenance of such Intellectual Property as is registered.

4. Audited Accounts and Management Accounts

- (a) The 2013 Audited Accounts and 2012 Audited Accounts give a true and fair view of the financial condition and the results of operations of the Group and of each of the Group Companies as at the date of, and for the period referred to in, the 2013 Audited Accounts and 2012 Audited Accounts, respectively.
- (b) The 2013 Audited Accounts were prepared in accordance with applicable Law and on the same basis and in accordance with the same accounting principles and practices, consistently applied, as the audited financial statements of the Group for the previous two financial periods.
- (c) None of the Group Companies has any off-balance sheet liabilities or any liabilities that are not adequately provided for or noted in the 2013 Audited Accounts.
- (d) Having regard to the purpose for which the Management Accounts have been prepared, the Management Accounts (i) have been prepared in accordance with accounting policies consistent with those used in preparing the 2013 Audited Accounts for the Group, applied on a consistent basis, and (ii) fairly represent in all material respects the assets and liabilities of the Group at the date to which they are made up and the profits and losses of the Group for each period to which they relate.
- (e) The accounting records of the Group are up-to-date, have been maintained on a proper and consistent basis and in accordance with applicable Law and accounting standards. They contain an accurate and complete record of all matters required to be entered in them or which are otherwise entered in them and no written notice or allegation that any of the accounting records is incorrect or should be rectified has been received.

5. Business Since the Accounts Date

Since the Accounts Date:

- the Group has carried on its business in the ordinary course so as to maintain its business as a going concern;
- there has been no material adverse change in the financial or trading position of the Group and, so far as the Sellers are aware, no circumstance has arisen that is likely to give rise to any such change;
- (iii) no Group Company has allotted or issued or granted any option over or other right to subscribe for or purchase, or agreed to allot or issue or grant any option over or other right to subscribe for or purchase, any share capital or loan stock or other security;

- (iv) no Group Company has declared, made or paid any dividend or other distribution to its members outside of the Group;
- (v) no Group Company has reduced, redeemed or purchased or agreed to reduce, redeem or purchase any of its share or loan capital or other securities;
- (vi) no Group Company has acquired or disposed of any asset nor agreed to acquire or dispose of an asset, in any case with a book value or for a consideration in excess of €250,000, other than in the ordinary course of trading;
- (vii) no Group Company has made or agreed to make any capital expenditure in an amount exceeding €250,000;
- (viii) there has been no material increase or decrease in the level of any Group Company's trading stock (including work-in-progress) or the price paid for its trading stock;
- (ix) the profits and losses of the Group have not been affected by changes or inconsistencies in accounting treatment or by any non-recurring items of income or expenditure by transactions entered into otherwise than on normal commercial terms or by any other factors rendering such profits exceptionally high or low;
- there has been no change in the manner or time of payment of creditors and there has been no change in the manner or time of collection of debts or the policy of reserving for debtors;
- (xi) the Group has not cancelled or delayed, in whole or in part, any capital expenditure or other item of discretionary spending that is set out in the Group's business plan in force as at the Offer Date;
- (xii) no Group Company has paid nor agreed to pay any management, advisory or similar charges to any member of the Institutional Seller's Group;
- (xiii) no resolution of shareholders of any Group Company has been passed or signed; and
- (xiv) no Group Company has changed its accounting reference period.

Borrowings

- (a) Details of all overdrafts, loans and other financial facilities available to each Group Company for an amount in excess of €100,000 (the "Facilities") and the amounts outstanding under them are set out in the Disclosed Document.
- (b) Details of all debentures, charges and guarantees and other security of any nature to secure the Facilities are set out in the Disclosure Documents.

- (c) The total amount borrowed by a Group Company under the Facilities, and the total amount borrowed by a Group Company from any source, do not exceed any limitation on the Group Company's borrowings set out in its articles of association or in any contract binding on it.
- (d) No Group Company has lent or agreed to lend any money to any person that is not a member of the Group, is not responsible for the indebtedness or other liability of any person that is not a member of the Group and has not given any guarantee, indemnity or other assurance of loss in relation to any indebtedness or other liability of any person that is not a member of the Group.
- (e) No event has occurred and, so far as the Sellers are aware, no circumstances exist which: (i) is or with the lapse of time and/or the giving of any notice, certificate, declaration or demand could reasonably be expected to become an event of default under any of the terms of the Facilities; or (ii) entitles, or with the lapse of time and/or the giving of any notice, certificate, declaration or demand could reasonably be expected to entitle any person to call for repayment or the cancellation of the availability of, or place on demand, any of the Facilities or alter to the disadvantage of a Group Company, the terms of any of the Facilities.
- (f) No Group Company has created or agreed to create any Encumbrance (except liens arising by operation of law in the ordinary course of trading) over its assets or entered into any factoring arrangement in relation to any of its assets.
- (g) There has been no breach by any Group Company of any of the provisions of any Encumbrance, or such arrangement or agreement as is referred to in paragraph (f) and all payments to be made by a Group Company in respect thereof have been duly paid by such Group Company.
- (h) No Group Company has any unmatched open positions with respect to any swap, hedge, option, warrant, future, forward purchase and/or sale transaction relating to currencies, commodities, securities, interest rates, indexes, events or conditions or any other derivative instrument and, so far as the Sellers are aware, none of such open positions could reasonably be expected to involve a Group Company in a loss.
- No Group Company owns the benefit of any debt (whether present or future) other than debt owing to it in the ordinary course of trading.

Customers and Suppliers

- (a) The Disclosure Documents contain:
 - (i) a list of top 5 customers of the Group in France, Germany and Spain (by revenues generated by the Group from its clients or customers for the 12 month financial period ending June 2013 and March 2014 respectively);

- (ii) a list of top 20 suppliers of the Group in relation to finished goods (by the amount of expenditure of the Group for the 12 month financial period ending June 2013 and March 2014 respectively).
- (b) So far as the Sellers are aware, no customer or supplier referred to in paragraph (a) has indicated that it shall cease to do business with the Group or materially reduce the volume of such business.

8. Material Contracts

- (a) The Disclosure Documents contain complete and accurate copies of all Material Contracts.
- (b) All Material Contracts constitute valid, binding and enforceable obligations of the applicable Group Company thereto (and, as far as the Sellers are aware, the applicable counterparty), and the terms of each such agreement have been complied with in all material respects by that Group Company and, so far as any of the Sellers are aware, by all other parties thereto.
- (c) No Group Company is in a material breach of any Material Contract and no Group Company has received any written notice (that is still valid) that any Group Company to any Material Contract is in a material breach of such Material Contract.
- (d) There are no grounds for the rescission, avoidance or repudiation of any Material Contract, nor has there been any written allegation received by any Group Company of such a thing, and no written notice of termination or intention to terminate has been given or received in respect of any Material Contract.
- (e) No offer, tender or the like is outstanding (the value of which to a Group Company could exceed €500,000 in any financial year) which is capable of being converted into an obligation of a Group Company by an acceptance or other act of another person.

For the purposes of this Agreement, a "Material Contract" means any written contract, agreement or arrangement to which any Group Company is a party which:

- involves annual expenditure by, or revenue to, such Group Company in excess of €1,000,000; or
- (ii) involves annual expenditure by, or revenue to, or other potential exposure for, such Group Company in excess of €500,000, and:
 - (A) is not in the ordinary course of business; or
 - (B) is not on an arm's length basis; or
 - (C) is long-term (meaning unlikely to be performed in full in accordance with its terms within twelve months of its commencement or being incapable of termination without

- compensation by any Group Company in twelve months or less); or
- (D) cannot readily be performed by the Group Company without undue or unusual expenditure of money or effort; or
- requires or is likely to require any Group Company to pay any commission, finder's fee, brokerage or similar payment whether or not in connection with the Transaction; or
- involves an agency, distributorship, franchise or management relationship; or
- (G) is a material outsourcing agreement or arrangement; or
- (H) is a joint venture, consortium, partnership or other agreement or written arrangement under which any Group Company participates with any other person in any business, including without limitation, any strategic partnership agreements with third parties; or
- on the change of control of any Group Company, can be terminated or varied or would entitle any person to exercise rights not otherwise available to such person; or
- (J) which will or may be in default or trigger termination or amendment rights, or cause an acceleration in payment or result in any Encumbrance over the assets of any Group Company or the shares in any Group Company as a result of the Transaction; and
- relate to the acquisition or disposal of any business or assets by any Group Company and impose obligations on any Group Company that are yet to be satisfied; or
- (iii) is, in the reasonable opinion of the Sellers, material to the business and operations of any Group Company, whether or not it is material in terms of expenditure or revenue expectations.

Intellectual Property Rights

- (a) The Group Companies own or are entitled to use pursuant to a license agreement, all Intellectual Property that is material to the business of the Group Companies in the manner and to the extent in which it is presently conducted (such Intellectual Property that is owned by the Group Companies, the "Owned Intellectual Property" and such Intellectual Property that is licensed or otherwise used pursuant to a contract the "Licensed Intellectual Property").
- (b) The Disclosure Documents contain complete and accurate details of all registered Owned Intellectual Property (and applications for any such right)

- and all material unregistered Owned Intellectual Property and the Group Companies are the sole legal and beneficial owners of all such rights free from all Encumbrances.
- (c) All of the Owned Intellectual Property is valid and enforceable and all renewal, application and other official registry fees and steps required for the maintenance, protection and enforcement of the material registered Owned Intellectual Property have been paid in full and on time (including extensions and reinstatements).
- (d) Nothing has been done or omitted to be done and no circumstances exist by which a person is or will be able to cause cancellation of a registration of any material Owned Intellectual Property.
- (e) So far as the Sellers are aware, all Confidential Information of the Group has been kept confidential, save where it has been disclosed to a third party under a duty of confidentiality.
- (f) The Company has not granted and is not obliged to grant any licence, consent, undertaking, security interest or other right in respect of any Owned Intellectual Property, save as Disclosed in the Disclosure Documents.
- (g) The Disclosure Documents contain complete and accurate copies of all licences or contracts relating to all material Licensed Intellectual Property, and all such licenses or contracts (i) are valid and enforceable in accordance with their terms; (ii) have been materially complied with by the applicable Group Company thereto (and, as far as the Sellers are aware, the applicable counterparty) during the 12 months prior to the date of this Agreement; and (iii) are not the subject of any written notice of termination by any of the parties thereto.
- (h) So far as any of the Sellers are aware, none of the operations of the Group Companies are infringing any Intellectual Property owned by a third party. No written claim (that has not been conclusively settled) has been received by any Group Company from any third party in the past three years alleging any infringing act or process or unauthorized use by a Group Company in respect of any Intellectual Property owned by a third party.
- (i) None of the Owned Intellectual Property or Licensed Intellectual Property is the subject of any opposition, dispute or Proceedings and no opposition, dispute or Proceedings is or are pending or threatened in writing.
- (j) So far as the Sellers are aware, there exists no actual infringement or unauthorised use by any third party of any Owned Intellectual Property, nor has there been any such infringement or unauthorised use in the two years prior to the date of this Agreement.

Insolvency

(a) No order has been made, petition presented to any Group Company, resolution passed or meeting convened by a Group Company for the purpose of winding

- up any Group Company or whereby the assets of any Group Company are to be distributed to creditors, shareholders or other contributories.
- (b) No administration order has been made and no petition for such an order has been presented to any Group Company or application made for an administration order by any Group Company and no administrator has been appointed, in each case in respect of any Group Company.
- (c) No event analogous to any of those described in paragraphs 10(a) or 10(b) above has occurred in any jurisdiction.
- (d) No receiver (including an administrative receiver), liquidator, trustee, administrator, manager, supervisor, nominee, custodian, or any similar or analogous officer or official in any jurisdiction has been appointed in respect of the whole or any part of the business or assets of any Group Company.
- (e) No Group Company is currently proposing any composition or similar arrangement with creditors.
- (f) No Group Company has stopped payment to its creditors as they fall due nor is it insolvent or unable to pay its debts as they fall due within the meaning of the Laws applicable to it.
- (g) No unsatisfied judgment is outstanding against any Group Company.
- (h) So far as the Sellers are aware, no Group Company has at any time in the last two years entered into any transaction with any person at an undervalue or been given a preference by any person.

11. Real Property

- (a) The real estate property details of which are included in the Disclosure Documents (the "Relevant Property") is true and accurate and comprises all and the only real estate property owned, used or occupied by any Group Company or in which any Group Company has any interest, right or liability. Each Relevant Property is used and occupied for the purpose of the Group's business.
- (b) In relation to the Relevant Property, the relevant Group Company is the sole legal and beneficial owner of such Relevant Property and has valid title over it free and clear of all title defects or other Encumbrances, and/or occupies and uses such Relevant Property pursuant to, or under the terms of, valid and enforceable leases, sub-leases, tenancies or licences.
- (c) In relation to each lease under which any part of the Relevant Property is held:
 - all rent due and demanded has been paid to date and there are no arrears;
 - (ii) no Group Company has received written notice from the landlord that it is in a material breach of its obligations under such lease (which breach remains outstanding); and

- (iii) there are no disputes with the landlord or any other person in respect of such Relevant Property.
- (d) There are no material disputes or proceedings relating to the Relevant Property or its use which are likely to prevent or significantly impede the Group Companies from carrying on their businesses in all material respects as currently conducted at the Relevant Property.
- (e) The Group Companies are entitled to and have exclusive possession of each of the Relevant Properties and no person other than the Group Companies has any interest in or has any right (actual or contingent) to possession, occupation or use of the Relevant Properties.
- (f) No Group Company has entered into any agreement or arrangement to acquire or dispose of any interest in land or premises that has not been completed.
- (g) Any documents of title including any leases pursuant to which the Group occupies the Relevant Properties have been registered at the relevant land registry if required.
- (h) The existing use of the Relevant Properties by the Group Companies is permitted both under Law and the terms of the relevant title documents or leases relating to the Relevant Properties and no notice of breach has been received relating to the existing use of the Relevant Properties by the Group Companies.
- (i) No Group Company, nor any person for whose acts or defaults a Group Company may be liable, has any liability in respect of land and buildings that have, at any time before the Offer Date, been owned (under whatever tenure) and/or occupied and/or used by a Group Company, but which are either no longer owned, occupied or used by a Group Company or are owned, occupied or used by one of them but pursuant to a different lease, licence, transfer or conveyance ("Former Land and Buildings").
- (j) No Group Company, nor any person for whose acts or defaults a Group Company may be liable, has given any guarantee or indemnity for any liability relating to any of the Relevant Properties any Former Land and Buildings, or any other land or buildings.

12. Litigation

There are no Proceedings involving any of the Group Companies or any Senior Employee (as defined below) for whose acts or defaults a Group Company may be vicariously liable or threatened in writing against any of the Group Companies or any such person, which, if determined adversely against such company or person, would result in its liability in an amount exceeding €250,000 (either on its own, or when aggregated with the amount of liability under any other Proceedings arising out of the same events, facts or circumstances), and so far as any of the Sellers are aware, there are no facts or circumstances which might give rise to any such Proceedings other than as Disclosed in the Disclosure Documents.

Labour Matters

- (a) The Disclosure Documents contain:
 - (i) an anonymised list of all employees of the Group and all workers engaged by the Group as at the Offer Date, including the following particulars in respect of each employee or consultant of the Group who is entitled to a base salary in excess of €125,000 (or equivalent in other currencies) per annum (a "Senior Employee"): name, employing entity, job title, date of commencement of continuous employment, usual city of work, current salary or wage, notice period (or date of expiry, if a fixed term contract), holiday entitlement and whether or not on a leave of absence for a period in excess of one month (and for what reason);
 - (ii) complete and materially accurate copies of the contracts of employment for all Senior Employee (including details of their respective salaries, length of service and notice periods), together with copies of any document that amends such contract;
 - (iii) copies of the standard terms and conditions of employment or engagement (as applicable) of employees (excluding Senior Employees), seconded employees within the Group and contractors who perform services personally under an agreement which is not a contract of employment, together with details of any material contract that departs from such standard form to any material extent;
 - (iv) copies of all of the staff handbooks, policies and procedures adopted by each Group Company;
 - details of all employee benefit or incentive plans operated by a Group Company for the benefit of employees or former employees ("Group Benefit Schemes"); and
 - (vi) details of all material collective bargaining agreements or arrangements which each Group Company has negotiated and entered into with any trade union, works council or similar body representing all or any category of employees of each Group Company.
- (b) All material governing documentation in relation to the Group Benefit Schemes has been made available in the Disclosure Documents.
- (c) The Group Companies do not participate in or contribute to any (i) retirement plans; (ii) profit-sharing, bonus or other incentive schemes; or (iii) other benefit plans, in all instances, other than those which are compulsory under applicable Law and collective bargaining agreements.
- (d) All amounts due and payable on or before the Completion Date by any Group Company in relation to the Group Benefit Schemes have been or will be duly paid in full on the due dates for such payments and all death in service benefits

- which may be payable in respect of employees are fully insured and all insurance premiums have been duly paid.
- (e) None of the Group Companies has entered into any contracts with any of its Senior Employees which provide, in the event of termination, for a notice period or payment of an indemnity which exceeds that provided for by applicable Law and collective bargaining agreements (if any).
- (f) No Senior Employee has given written notice terminating his contract of employment or consultancy agreement (which has not yet terminated) or is under notice of dismissal or termination.
- (g) No employee or contractor is, or will become, entitled to any (or to any increase or enhancement to, or to any accelerated vesting, funding or time of payment of, any) payment or benefit, or to treat himself as dismissed or released from any obligation, in each case as a result of the Transaction.
- (h) No offer of a contract of employment or engagement has been made by any Group Company to any Senior Employee which has not yet been accepted, or which has been accepted but the individual's employment or engagement has not yet commenced.
- (i) Since the Accounts Date:
 - no Group Company has changed, or agreed to change, or is obliged to change, the terms of employment of any of its Senior Employees and there are no requests or negotiations for any such change which are likely to take place within the next six months;
 - (ii) no Group Company has or is obliged to or has made any provision to increase or vary the salary, fee, pension benefits, bonus, commission or other remuneration or benefits whether in cash or in kind payable to employees or contractors which could increase the Group's total costs in respect of employees and contractors by more than €250,000; and
 - (iii) no Group Company has provided, or agreed to provide, a gratuitous payment or benefit to any of its Senior Employees, or to any of their respective dependents.
- (j) No Group Company has entered into any recognition or other agreement with a trade union or received any request from a trade union for recognition in the last two years, or carried out any act which may be construed as giving a trade union any legal entitlement in relation to recognition.
- (k) There is no dispute between any Group Company and any trade union or a collective bargaining body recognised by the Group which is existing or, so far as the Sellers are aware, is pending or threatened in writing.
- (l) Within the period of three years before the Offer Date, no Group Company has initiated or completed the implementation of any collective dismissals or implemented or entered into a social plan.

(m) With respect to any employee or any person who provides personal services to a Group Company as a contractor or a seconded employee, such Group Company (i) has classified each such person as an "employee", "seconded employee" or "contractor" in accordance with applicable Law and has a reasonable basis for each such determination, (ii) all employer tax returns have been filed on a basis consistent with such classification and (iii) there are no Proceedings by or in respect of any person classified by a Group Company as a contractor to reclassify such person as an employee for any purposes pending or threatened in writing.

Compliance with Laws and collective bargaining agreements

- (a) The Group is conducting, and has during the preceding two year period conducted, its business in material compliance with all applicable Law and collective bargaining agreements and no Group Company is, nor has been during the preceding two year period, in a material breach of any such Law or collective bargaining agreement.
- (b) There is no investigation, disciplinary proceeding or enquiry by, or order, decree, decision, prosecution or judgment of, any court, tribunal, arbitrator or Governmental Entity outstanding against any Group Company or any Senior Employee for whose acts or defaults it may be vicariously liable which will have a material adverse effect upon the assets, business or prospects of the Group.
- (c) No Group Company has received any notice or other communication (official or otherwise) from any court, tribunal, arbitrator or Governmental Entity with respect to an alleged, actual or potential breach of, and/or failure to comply with, any applicable Law or collective bargaining agreement requiring a Group Company, or any person for whose acts or defaults it may be vicariously liable, to take or omit to take any action.

15. Product Liability and Insurance

- (a) So far as the Sellers are aware, no Group Company has given any guarantees or indemnity, in respect of goods or services supplied or accepted any liability which would apply after any such goods or services have been supplied other than as may be implied by Law or contained in its standard terms of business.
- (b) There have not been any regulatory product recalls or written correspondence concerning the same with any relevant Governmental Entity with respect to any products manufactured by the Group during the 12 months period prior to the Offer Date.
- (c) The Group has at all times maintained adequate insurance cover against risks normally insured against by companies carrying on similar businesses or owning property and/or assets of a similar nature to the Group ("Policies") and, in particular, has maintained all insurance required by Law and adequate third party product, environmental liability insurance and ill-health or disablement at work, accident at work and life insurance and has insured its

- assets to their full reinstatement or replacement value with a well-established and reputable insurer.
- (d) All premiums due on the Policies have been duly paid and each of the Policies is in full force and effect and nothing has been done or omitted to be done which would be reasonably likely to make any such Policy void or voidable.
- (e) There are no claims outstanding under the Policies and no event has occurred during the period of two years prior to the Offer Date which might reasonably give rise to a claim.
- (f) Details of all claims made under the Policies during the period of two years prior to the Offer Date are set out in the Disclosure Documents.

16. Tax

- (a) Each Group Company has complied with all its duties under all Tax legislation, kept made or filed with all appropriate Tax Authorities all Tax returns, disclosures, notices and Tax reports required to be filed or made by applicable law on or before the date hereof and supplied all other information required by law to be supplied to all relevant Tax Authorities, and all such information, Tax returns, notices and disclosures were submitted within applicable time limits and were accurate and complete in all material respects.
- (b) Each Group Company has duly paid within the applicable time limits all Tax to the extent that the same ought to have been paid and no Group Company is liable, nor has it been liable, to pay any fine, surcharge, penalty or interest in connection with the payment of Tax.
- (c) None of the Group Companies is, so far as the Sellers are aware, the subject of any dispute, material investigation or proceeding with or by any Tax Authority and there are no facts or circumstances likely to give rise to or be the subject of any such dispute, material investigation or proceeding. No Group Company has received any written notice of any Tax Authority's intent to conduct an investigation, visit, enquiry or audit.
- (d) Each Group Company has within applicable time limits duly submitted all claims, disclaimers, elections, surrenders and applications which have been assumed to have been made for the purposes of the 2013 Audited Accounts. Details of such claims, disclaimers, elections, surrenders and applications are set out in the Disclosure Documents.
- (e) The 2013 Audited Accounts, as at the Accounts Date, make full provision or reserve in respect of any period ended on or before the Accounts Date for all Tax assessed or liable to be assessed on each Group Company or for which it is accountable at the Accounts Date whether or not it has or may have any right of reimbursement against any other person, and proper provision has been made and shown in the 2013 Audited Accounts as at the Accounts Date for deferred Tax in accordance with the relevant generally accepted accounting principles.

- (f) No Taxation Authority has agreed with any Group Company to operate any special arrangement (being an arrangement which is not based on relevant legislation or any published practice) in relation to the affairs of any Group Company.
- (g) Each Group Company has, on or before the relevant due date, complied with its obligations to deduct, withhold or retain amounts of or on account of Tax from any payments made by it and to account for such amounts to the relevant Tax Authority and has complied with all its reporting obligations to the relevant Tax Authority.
- (h) Each Group Company is and has at all times been resident for Tax purposes in the jurisdiction in which it was incorporated and no Group Company is nor has at any time been treated as resident in any other jurisdiction for any Tax purpose (including for the purpose of any double tax treaty). No Group Company is, nor has it ever been, subject to Tax in any jurisdiction other than its place of incorporation by virtue of having a permanent establishment or other place of business in that jurisdiction.
- (i) All transactions entered into by a Group Company have been entered into on an arm's length basis. No notice or enquiry by any Tax Authority has been made in connection with the arm's length nature of any transaction entered into by any Group Company.
- (j) No Group Company will become liable to pay any Tax as a result of Completion.
- (k) No transaction, act, omission or event has occurred in consequence of which any Group Company is or, so far as the Sellers are aware, may be held liable for any Tax or may otherwise be held liable for or to indemnify any person in respect of Tax which is primarily or directly chargeable against or attributable to any person other than the Group Company.
- No Group Company is, nor has it ever been, a member of a group, fiscal unity or similar entity or association for any Tax purpose.
- (m) Each Group Company has complied in all respects with the terms of applicable Laws in respect of VAT (including, for the avoidance of doubt, applicable registration requirements in any jurisdiction) and the Disclosure Documents contain full details of any VAT registrations of the Group Companies. No any Group Company is an agent or fiscal representative for VAT Purposes.
- (n) All documents which any Group Company needs in order to provide title to any asset have been duly stamped and all applicable stamp duty, interest and penalties have been duly paid.
- (o) The Sellers' representations and warranties set forth in this subsection 16 are the only representations and warranties given in respect of Tax.

17. Commission and finder's fees

No person is entitled to receive from any Group Company or any person acting on behalf of a Group Company any finder's fee, brokerage or other commission in connection with the Transaction.

18. Information Technology

- (a) All IT Systems necessary for the carrying on of the business of the Group in substantially the same manner and to substantially the same extent as at the Offer Date, are owned by or validly licensed to the applicable Group Companies and such IT Systems shall be available for use by the applicable Group Companies on Completion on terms and conditions substantially on the same terms as those under which the Group Companies owned or used such IT Systems during the 12 months prior to the Offer Date.
- (b) All material IT Systems are in good working order and function materially in accordance with the specifications applicable to them and there are, and during the 12 months prior to the Offer Date, there have been, no performance reductions or breakdowns of, or logical or physical intrusions to, any IT Systems or loss of data which have had (or are having) a material adverse effect on the use of the IT Systems by the Group Companies and the Sellers are not aware of any fact or matter which may give rise to such a material adverse effect.
- (c) The Group Companies have taken all steps deemed reasonably necessary by them to ensure that the business can continue in all material respects in the event of a failure of the IT Systems and has up to date, appropriate and adequate disaster recovery infrastructure which is regularly tested.
- (d) The Group Companies are compliant in all material respects with the terms of all licences in respect of any third party software, including any open source software, used in the business of the Group.

Licences and consents

- (a) Each Group Company has all material licences, consents, permits and authorisations ("Licences") that are necessary for the carrying on of its business in substantially the manner and in the places in which such business is now carried on and all such Licences are in full force and effect and are being complied with in all material respects.
- (b) So far as the Sellers are aware, there are no circumstances which indicate that any of the Licences may be modified or revoked or not renewed or which confer a right of modification or revocation or non-renewal.

20. Financial Crime

(a) No Group Company has and, so far as the Sellers are aware, no director, consultant, agent or other person associated with any Group Company has, directly or indirectly, in furtherance of or in connection with the business of any Group Company:

- (i) offered, promised or given any financial or other advantage to any person with the intention of influencing a person (who need not be the recipient of the advantage) to perform his or her function improperly, or where the acceptance of such advantage would itself be improper, or in violation of any applicable anti-corruption or anti-bribery law;
- requested, agreed to receive or accepted any financial or other advantage where this would be improper or likely to influence the recipient in the performance of his or her role; or
- (iii) offered, promised or given any financial or other advantage to a public official (or to any other person at the request of, or with the acquiescence of, a public official) with the intention of influencing that official in the performance of his or her public functions, whether or not that performance would be improper.
- (b) The Alleged South African Conduct, as defined and described in the Disclosure Documents, does not involve, in any way, government officials, political party members, political parties, candidates for political office, or employees or representatives of state-owned or controlled enterprises.
- (c) The Group has policies, procedures and controls reasonably designed to ensure that the Group has complied and will comply with any applicable anticorruption or anti-bribery law.
- (d) Details of the anti-bribery policies and procedures adopted by each Group Company are set out in the Disclosure Documents.

21. Grants and Subsidies

- (a) Details of all Governmental Entity investment grants, loan subsidies or financial aid received by or pledged to a Group Company during the last two years, or in respect of which applications are outstanding ("Grants") are set out in the Disclosure Documents.
- (b) The relevant Group Company has complied with all its obligations to which any Grant is subject and there are no outstanding obligations that may be required to be complied with by such Group Company after Completion.
- (c) No Group Company has done or omitted to do any act or thing which could result in any Grant (or any part thereof) becoming repayable or forfeited or any application for any Grant being rejected, nor is any Seller aware of any circumstances which might lead to any such repayment, forfeiture or rejection.
- (d) The entering into and performance of this Agreement any other Transaction Document will not result in any Grant (or any part thereof) becoming repayable or forfeited.

22. Trade

- (a) No Group Company is and never has been a party to or directly or indirectly engaged or involved in any contract, arrangement, understanding or practice (whether legally binding or not, and whether by omission or otherwise) which:
 - (i) infringes Article 101 or 102 of the Treaty on The Functioning of the European Union or any other anti-trust or similar Law in any jurisdiction in which a Group Company has assets or carries on or intends to carry on business or where its activities may have an effect; or
 - (ii) is unenforceable or void (whether in whole or in part) or renders it liable to any Proceedings by virtue of any anti-trust or similar legislation in any jurisdiction in which a Group Company has assets or carries on or intends to carry on business or where its activities may have an effect.
- (b) No Group Company has given an undertaking to, is not subject to any order of, or investigation by, and has not received any request for information from, any Governmental Entity (including any national competition authority or sectoral regulator, or the European Commission) under any anti-trust or similar Law in any jurisdiction in which a Group Company has assets or carries on or intends to carry on business or where its activities may have an effect.
- (c) None of the businesses or activities of any Group Company as currently conducted could reasonably be expected to give rise to the imposition of any duty or other sanction under any applicable trade regulation Laws in respect of any products manufactured by a Group Company or in which a Group Company trades.
- (d) No sanction under any applicable trade regulation Laws is or has been in force in any area in or to which a Group Company manufactures, trades or sells in respect of products manufactured by a Group Company or in which a Group Company trades.
- (e) No undertaking has been given by a Group Company to any Governmental Entity (including the authorities of the European Union) under any trade regulation legislation.
- (f) The Sellers are not aware of any investigation, complaint, action or negative decision in relation to the receipt or alleged receipt by any Group Company of any aid or alleged aid or of any such threatened investigation, complaint, action or negative decision.

23. Data Protection

(a) In respect of the personal data (whether automatically or manually recorded) held by a Group Company:

- all such data has in the last two years been processed by the Group Company fairly and lawfully and in material compliance with all applicable data protection Laws ("DP Laws");
- so far as the Sellers are aware, all necessary consents have been given by the data subjects to the processing of it in the manner in which the Group Company has processed it;
- (iii) so far as the Sellers are aware, all such data is held in a retrievable, permanent and legible form and appropriate measures have been taken against unauthorised processing or accidental loss, damage or destruction of or to it;
- so far as the Sellers are aware, all such data is adequate, relevant and not excessive for the purposes for which it is held and it is accurate and up-to-date and has not been kept for longer than necessary;
- (v) all such data has not been transferred out of the European Economic Area except in compliance with the DP Laws;
- (vi) the Group Company has made notifications to the relevant Governmental Entity, which are up-to-date; and
- (vii) no written complaints have been made in respect of such data and no enforcement notices have been served on the Group Company by any Governmental Entity and no fact or circumstance exists which might give rise to any such complaint, concern or enforcement notice.
- (b) There are no enforcement, deregistration or transfer prohibition notices or any other notices whatsoever under DP Laws outstanding against any Group Company, nor is there any outstanding appeal against such notices and there are no circumstances which may give rise to the giving of any such notices to any Group Company.
- (c) There are no outstanding claims for compensation for inaccuracy, loss or unauthorised disclosure of personal data nor is any personal data held by any Group Company not true and complete, nor has any Group Company lost or made any unauthorised disclosure of any such data.

24. Assets

(a) The Group Companies have legal and beneficial title to and are, where capable of possession, in possession and control of, all material assets included in the 2013 Audited Accounts or which were acquired by the Group since the Accounts Date (except for assets sold, realised or applied in the ordinary course of trading) ("Assets"). All the Assets are free from Encumbrances (save for those Encumbrances created in the ordinary course of business) and no third party has or claims any rights in relation to the Assets (or to the proceeds of sale of the Assets).

- (b) The Assets comprise all the material assets necessary for the carrying on of the business of the Group substantially in the manner and to the extent it is now conducted.
- (c) No person has the right to call for the assignment of or any payment in respect of, any of the Assets, or to retake any Asset into their possession.
- (d) The plant and machinery (including fixed plant and machinery), vehicles and office and other equipment owned or used by a Group Company in connection with its business are in satisfactory working order.
- (e) The Group's inventory of raw materials, work-in-progress and finished goods are in a good and materially undamaged condition and are capable of being used and, in respect of finished goods, sold by the Group in the ordinary course of trading in accordance with the Group's current price list without excessive rebate or allowance in relation to the current trading requirements of the Group's business.
- (f) None of the trade receivables or other rights of payment (including unbilled receivables) of the Group which are included in the 2013 Audited Accounts or which have subsequently arisen have been outstanding for more than three months from its due date for payment nor been released on terms that the debtor has paid less than the full value of its debt.

25. Environmental

- (a) Each Group Company and its activities and operations have been and are in compliance with all applicable Laws and licences relating to: (i) protection, preservation or clean-up of the environment or natural resources; (ii) chemical substances, pollutants, or toxic, hazardous or deleterious materials, wastes or agents, including petroleum or any fraction thereof, asbestos-containing materials, and polychlorinated biphenyls ("Hazardous Substances"); or (iii) health and safety (together, "Environmental Laws").
- (b) No Group Company has been subject to any Proceedings pursuant to Environmental Laws or relating to Hazardous Substances and, so far as the Sellers are aware, no such Proceedings are threatened, pending or expected.
- (c) There are no conditions or circumstances, including the release or presence of, or exposure to, any Hazardous Substance, reasonably anticipated to result in liabilities of any Group Company pursuant to Environmental Laws.

26. Consequences of this Agreement

The entering into and performance of this Agreement and the Transaction Documents will not:

(a) so far as any of the Sellers are aware, cause any person who normally does business with or gives credit to a Group Company to cease to do business with the Group or not to continue to do so on the same basis as previously; or (b) result in a breach or constitute (with or without the lapse of time and/or the giving of any notice, certificate, declaration or demand) a default or give rise to any right of termination, variation, payment or acceleration, under any Material Contract to which a Group Company is a party or result in the imposition of an Encumbrance on the assets or shares of any Group Company.

DEMB

- (a) The Group Companies have complied in all material respects with their obligations under the Original Purchase Agreement.
- (b) All arrangements or understandings between DE Master Blenders 1753 B.V. and any Group Company have been Disclosed.
- (c) Other than with respect to the payment of the deferred consideration to DE Master Blenders 1753 B.V. pursuant to section 2(f) of the Original Purchase Agreement, no Group Company has any obligation or liability towards DE Master Blenders 1753 B.V..
- (d) No event has occurred or, so far as the Sellers are aware, been alleged and no circumstance exists which are reasonably likely to result in any claim by DE Master Blenders 1753 B.V. against any Group Company or vice versa.

28. No other representation and warranty

- (a) The Sellers' representations and warranties set forth in this Schedule 5 (Business Warranties) and in Clause 8.1 and Clause 8.2 are the only representations and warranties given by the Sellers.
- (b) In particular, the Sellers make no representation and warranty with respect to any projection, forecast, analysis, perspective or business plan exchanged with the Purchaser in the context of the Transaction.

SCHEDULE 6

Post Completion Adjustments

PART 1: Preparation of Completion Statement

- After Completion, the Purchaser shall prepare a draft Completion Statement in accordance with Part 2 of this Schedule 6.
- The Purchaser shall procure that the draft Completion Statement is submitted to the Institutional Seller within 25 Business Days after (and excluding) the last day of the month in which Completion occurs.
- Within 25 Business Days after (and excluding) the date of receipt of the draft Completion Statement, the Institutional Seller shall serve notice on the Purchaser that either:
 - (a) the draft Completion Statement is agreed; or
 - (b) the draft Completion Statement is not agreed, in which case such notice (a "Sellers' Disagreement Notice") shall set out in reasonable detail:
 - (i) those amounts with which the Institutional Seller disagrees;
 - to the extent reasonably possible, the quantum (in the Institutional Seller's opinion) of such amounts; and
 - (iii) the reasons why the Institutional Seller disagrees with those items.

Following the service of a Sellers' Disagreement Notice, all matters contained in the draft Completion Statement other than those items that are the subject of the Sellers' Disagreement Notice shall become final and binding on all Parties.

- 4. In order to enable the Institutional Seller and the Sellers' Accountants to review the draft Completion Statement, the Purchaser shall with effect from the date it submits the draft Completion Statement to the Institutional Seller, on reasonable notice and subject to the execution of appropriate hold harmless letters and confidentiality undertakings:
 - make available to the Institutional Seller and the Sellers' Accountants during normal office hours:
 - (i) its working papers; and
 - such books and records relating to the business of the Group and in the possession or under the control of the Purchaser as they may reasonably require; and
 - (b) co-operate with the Institutional Seller with regards to the provision of such information, explanations and responses to queries as is reasonably required by the Institutional Seller in connection with its review of the draft Completion Statement and shall make available the services of the employees

of the Group in connection with the provision of such information, provided that this does not interfere with the proper operation of the Group after Completion.

5. If the Institutional Seller:

- notifies the Purchaser in accordance with Paragraph 3(a) above that the draft Completion Statement is agreed; or
- (b) fails to give notice to the Purchaser as required by Paragraph 3 above,

then the draft Completion Statement shall be final and binding and shall, for the purposes of this Agreement, be treated as agreed on the date on which such notice is served or the next date after the last date on which such notice could be served (as applicable).

- 6. If the Institutional Seller serves a Sellers' Disagreement Notice in accordance with Paragraph 3(b) above, the Purchaser and the Institutional Seller shall use reasonable efforts to resolve the matters in dispute within 10 Business Days after (and excluding) the day of receipt of the Sellers' Disagreement Notice.
- 7. If the disputed items are agreed within the period specified in Paragraph 6, the draft Completion Statement (amended as agreed) shall be final and binding and shall, for the purposes of this Agreement, be treated as agreed on the date on which such agreement is reached.
- 8. If the disputed matters are not agreed within the period specified in Paragraph 6, either the Purchaser or the Institutional Seller may refer the remaining matters in dispute to the Reporting Accountants (as defined in Paragraph 9 below), provided that, upon such referral, all matters contained in the draft Completion Statement other than those items that are the subject of the referral shall become final and binding on all Parties.

Reporting Accountants

- (a) Mazars shall be engaged jointly by the Institutional Seller and the Purchaser as reporting accountants (the "Reporting Accountants") on the terms set out below. If either (i) Mazars is unable or unwilling to act or (ii) the terms of engagement of the Reporting Accountants have not been settled within five Business Days of the referral pursuant to Paragraph 8 (or such longer period as the Institutional Seller and the Purchaser may agree) then, those accountants shall be deemed never to have become the Reporting Accountants and new Reporting Accountants shall be selected by the Institutional Seller and the Purchaser within five days of a notice by one to the other requiring such agreement or failing such agreement to be nominated on the application of either of them by or on behalf of the President of the Institute of Chartered Accountants of England and Wales (or any successor thereto).
- (b) Except to the extent that the Institutional Seller and the Purchaser agree otherwise, the Reporting Accountants:

- (i) shall determine only whether any, and if so which, adjustments to the draft Completion Statement required by the Institutional Seller on which the Institutional Seller and the Purchaser still disagree shall be made and shall not put forward any alternative adjustment(s) that is outside of the range provided by the Institutional Seller and the Purchaser;
- shall apply the accounting principles, policies, procedures, practices, categorisations, definitions, methods and techniques set out in <u>Schedule 7</u>; and
- (iii) for the avoidance of doubt shall not be entitled to determine the scope of their own jurisdiction.
- (c) The Reporting Accountants shall act as experts and not as arbitrators and their determination of any matter falling within their jurisdiction shall be final and binding on the Parties save in the event of manifest error (when the relevant part of their determination shall be void and the matter shall be remitted to the Reporting Accountants for correction). In particular, without limitation, their determination shall be deemed to be incorporated into the draft Completion Statement.
- (d) Each Party shall bear its own costs and expenses in relation to this <u>Schedule 6</u> (including the fees of their respective accountants and other advisers incurred in connection with the determination of the Completion Statement), provided that the fees and expenses of the Reporting Accountants shall be allocated between the Sellers, on the one hand, and the Purchaser, on the other hand, as determined (and recorded in their final determination) by the Reporting Accountants based upon the relative success (in terms of percentages) of each such Party's claims. For example, if the final determination reflects a 60-40 compromise of the Parties' claims, the Reporting Accountants would allocate expenses 40% to the Party whose claims determined to be 60% successful and 60% to the Party whose claim was determined to be 40% successful.
- (e) Nothing in this Paragraph 9 shall entitle a Party or the Reporting Accountants access to any information or document that is protected by legal professional privilege or litigation privilege, provided that a Party shall not be entitled to refuse to supply such part or parts of documents as contain only the facts on which the relevant claim or argument is based.
- (f) Each Party and the Reporting Accountants shall, and shall procure that its accountants and other advisers shall, keep all information and documents provided to them pursuant to this Paragraph 9 confidential and shall not use the same for any purpose, except for disclosure or use in connection with the preparation of the draft Completion Statement, the proceedings of the Reporting Accountants or another matter arising out of this Agreement or in defending any claim or argument or alleged claim or argument relating to this Agreement or its subject matter.

10. Determination

- (a) The determination of the Reporting Accountants shall be based solely on written presentations submitted by or on behalf of the Institutional Seller and the Purchaser which are in accordance with the guidelines and procedures set out below and otherwise in accordance with this Agreement, and not on the basis of an independent review. The Reporting Accountants shall consider only the disputed matters that were included in the Sellers' Disagreement Notice and the Reporting Accountants may not assign a value to any item other than within the range of the values assigned by the Purchaser, on the one hand, or the Institutional Seller, on the other hand.
- (b) The timetable for the Reporting Accountants to make their determination will be governed by the following procedures:
 - (i) Within ten Business Days of retaining the Reporting Accountants, the Institutional Seller and the Purchaser shall each submit to the Reporting Accountants three copies of a memorandum (which must include supporting exhibits) explaining their respective positions of all unresolved disputed items (the "Initial Report").
 - (ii) Within one Business Day of receiving Initial Reports from both the Institutional Seller and the Purchaser, the Reporting Accountants shall distribute a copy of each Initial Report to the other Party.
 - Within ten Business Days of receiving the other Party's Initial Report (iii) from the Reporting Accountants, each of (i) the Institutional Seller and (ii) the Purchaser may (but shall not be obliged to) submit to the Reporting Accountants three copies of a memorandum responding to the Initial Report submitted to the Reporting Accountants by the other Party (the "Rebuttal Report"). If a Party elects not to submit a Rebuttal Report, it shall notify the Reporting Accountants in writing that it has waived its right to do so. The Rebuttal Report may respond solely to the arguments raised, and information submitted, by the other Party in its Initial Report and no Party may introduce new arguments or rely on new information in the Rebuttal Report that was not part of such Party's Initial Report or which are not directly responsive to an argument raised by the other Party's Initial Report, except to the extent such new arguments or new information are used in direct response to arguments raised and information submitted by the other Party in its Initial Report.
 - (iv) Within one Business Day of receiving Rebuttal Reports from both the Institutional Seller and the Purchaser, the Reporting Accountants shall distribute a copy of each Rebuttal Report to the other Party.
 - (v) At any time within ten Business Days of the submission of the Initial Reports or any Rebuttal Reports by the Institutional Seller and the Purchaser, the Reporting Accountants may submit written questions to either Party following the procedures set out in Paragraph 10(c) below.

- (vi) Upon receipt of the Rebuttal Report or notice waiving the right to submit such a report from both the Institutional Seller, on the one hand, and the Purchaser, on the other hand, and receipt of all responses to any written questions submitted by the Reporting Accountants (and responses thereto), the Reporting Accountants shall endeavour to issue a report containing their determination within ten Business Days after the later of (i) receiving Rebuttal Reports from both the Institutional Seller and the Purchaser or notice waiving the right to submit such report, as applicable, or (ii) any responses (if any) to any written questions submitted by the Reporting Accountants to either Party following the procedures set out in Paragraph 10(c) below.
- (vii) Unless requested by the Reporting Accountants in writing pursuant to the terms of Paragraph 10(c) below, neither the Institutional Seller nor the Purchaser may present any additional information or arguments to the Reporting Accountants, either orally or in writing.
- (viii) The Reporting Accountants shall render their decision without conducting a hearing but shall provide reasons for their determination.
- After receiving both Initial Reports and Rebuttal Reports, if any, the Reporting (c) Accountants may submit written questions to the Institutional Seller, on the one hand, and the Purchaser, on the other hand, for written responses or may direct requests for additional information, calculations, or supporting documentation to the Parties reasonably needed by the Reporting Accountants in order to clarify or understand any position or argument made by a Party in its written submission, in which case the Parties agree to co-operate with such requests (including, without limitation, by ensuring that the Reporting Accountants are provided copies of all relevant books and records and other information) in the manner and in accordance with the timing described in this Schedule 6. If any such questions are addressed to only one Party, the Reporting Accountants shall submit the questions to that Party, with a copy to the other Party. Once received, the Party to whom the questions are addressed shall have five Business Days to answer the Reporting Accountants' questions, and shall provide a copy of its written answers to the other Party at the time they are provided to the Reporting Accountants. In response thereto, the other Party may, within five Business Days, submit a response to such answer(s) to the Reporting Accountants and shall provide a copy of its response to the other Party at the time it is provided to the Reporting Accountants. If any such questions are addressed to both the Institutional Seller and the Purchaser, both Parties shall have five Business Days from the date of receipt to respond to the Reporting Accountants and each Party shall provide a copy of its written answers to the other Party at the time they are provided to the Reporting Accountants. In response thereto, each Party may, within five Business Days, submit a response to the other Party's answer(s) to the Reporting Accountants and shall provide a copy to the other Party at the time it is provided to the Reporting Accountants.
- (d) The Parties agree not to engage in any ex parte communication with the Reporting Accountants. The Reporting Accountants will be required to include a representation in their engagement letter that they have not discussed

- the disputed matter with any Party prior to their being retained by the Parties, and to include a covenant in their engagement letter not to engage in ex parte communications with any Party throughout the course of their engagement.
- (e) The Reporting Accountants will be required to agree that, if between the delivery of the Sellers' Disagreement Notice and their engagement, the Institutional Seller and the Purchaser have exchanged certain proposals relating to the disputed items, such proposals were (i) intended solely for purposes of facilitating settlement discussions, (ii) were confidential and (iii) were provided solely on the condition and understanding that such proposals would not be permitted to be disclosed in any court or arbitration hearing, including with respect to the Reporting Accountants' engagement in the dispute. The Reporting Accountants will be instructed to disregard any evidence of such settlement proposals and negotiations in their consideration of the disputed matters.
- (f) Except in connection with the dispute being resolved with respect to this Agreement, during the term of their engagement no member of the Reporting Accountants' team may work on any matters related to the Sellers or the Purchaser or otherwise perform services for any person that in any case may present a conflict of interest that is reasonably likely to affect the Reporting Accountants' services or the unbiased performance of services by any member of the Reporting Accountants' team. The foregoing restrictions on the Reporting Accountants will not apply to employees of the Reporting Accountants not assigned to work on this engagement. On the date they are engaged to resolve any dispute under this Agreement, the Reporting Accountants will be required to confirm to the Parties that they have not had any communication with any Party or discussed the matter with any person except as disclosed in writing to the Parties or a communication in which representatives of both the Parties participated.

11. Institutional Seller Representation

- (a) Each Management Seller hereby irrevocably appoints the Institutional Seller as his or her representative to act on his or her behalf for all purposes under this <u>Schedule 6</u>, including for the purposes of:
 - reviewing the draft Completion Statement and negotiating, disputing and agreeing the Completion Statement and each of the constituent parts thereof on his or her behalf;
 - appointing the Reporting Accountants and making such representations and submissions to them as are contemplated by this Schedule 6;
 - taking any and all actions that may be necessary or desirable, as determined by the Institutional Seller in its absolute discretion, in connection with the Completion Statement;
 - (iv) exercising exclusively on his or her behalf any right given to it under this <u>Schedule 6</u>, which exercise shall be binding on him or her; and

- generally taking any and all actions and doing any and all other things provided in or contemplated by this <u>Schedule 6</u> to be performed by the Sellers.
- and shall refrain from doing anything inconsistent with any of the Institutional Seller's actions taken in its capacity as his or her representative hereunder.
- (b) By entry into this Agreement, each Management Seller hereby agrees to (i) waive any and all claims that he or she may have against the Institutional Seller arising out of or in connection with the actions taken by the Institutional Seller pursuant to this <u>Schedule 6</u>, except for actions taken by it that constitute wilful misconduct on its part and (ii) indemnify and hold harmless the Institutional Seller and its shareholders, directors, officers, employees, agents and other representatives against any and all losses or liabilities that they may suffer or incur by reason of the fact that the Institutional Seller is or was acting as the Management Seller's representative pursuant to the terms of this Schedule 6.

PART 2: Form and Content of Completion Statement

- The Annex to this <u>Schedule 6</u> contains a pro-forma Completion Statement in Part A, together with a pro-forma Actual Working Capital Statement in Part B, a pro-forma Actual Debt Statement in Part C, a pro-forma Actual Cash Statement in Part D and a spreadsheet showing the agreed categorisations of each line item appearing in the "Hyperion" accounts software (the "Spreadsheet").
- 2. In accordance with the categorisations shown in the Spreadsheet, each of the Actual Cash Amount, Actual Debt Amount and Actual Working Capital Amount shall only include the assets and liabilities as set out in the respective relevant constituent line items identified therein (and Actual Cash Statement, Actual Debt Statement and Actual Working Capital Statement, as applicable) and no other categories of assets or liabilities shall be included in respect of any of them. Such constituent line items shall be treated in accordance with the Accounting Policies. No other categories of assets or liabilities other than as specified in the foregoing shall be included in the Completion Statement.

SCHEDULE 7

Accounting Policies

1. Basis of preparation

- The Completion Statement will be prepared in accordance with the following order of priority:
 - (i) By applying the specific accounting principles, policies, procedures, methods, estimation techniques and classifications set out in paragraph 2 below ("Specific Accounting Policies");
 - (ii) To the extent not specified in (i) above, on a consistent basis with the accounting principles, policies, management estimates, treatments, classifications and categorisations as were used in the preparation of the 2013 Audited Accounts;
 - (iii)To the extent not specified in (i) or (ii) above, on a consistent basis with the accounting principles, policies, management estimates, treatments, classifications and categorisations as were used in the preparation of the March 2014 management accounts; and
 - (iv) To the extent not covered by (i), (ii) or (iii) above, IFRS as at 30 June 2013.

For the avoidance of doubt, paragraph (i) above shall take precedence over paragraphs (ii), (iii) and (iv) above, paragraph (ii) above shall take precedence over paragraphs (iii) and (iv) above and paragraph (iii) above shall take precedence over paragraph (iv) above.

The Group cannot create new general ledger accounts versus those existing at 30 June 2013.

2. Specific Accounting Policies

(a) General

The Completion Statement shall be prepared on the following basis:

- (i) The Completion Statement shall be prepared on a consolidated basis, in Euros as at the Effective Time and shall be prepared in the format set out in the Annex to Schedule 6, making reference to the relevant line items in the Spreadsheet.
- (ii) Going concern The Completion Statement shall be prepared on a going concern basis and as if the Completion Date was a financial year end of the Group and include all accounting adjustments and entries as would be incurred in closing applied on a consistent basis as reflected in connection with the 2013 Audited Accounts.
- (iii)Post Balance Sheet Events The Completion Statement shall take into account 'adjusting events' as defined in IFRS 10 'Events after the Balance Sheet Date' up to

the date on which the draft Completion Statement is submitted to the Institutional Seller pursuant to paragraph 2 of Part 1 of <u>Schedule 6</u> (Post Completion Adjustments) (the "Cut-Off Time").

- (iv)Foreign currency balances Asset and liability balances to be taken into account in calculating the Completion Statement that are expressed in a currency other than Euros shall be translated into Euros using the relevant month end exchange rates used by the Group sourced from Banque de France. Profit and loss account transactions in foreign currencies shall be translated at the average Group exchange rate sourced from Banque de France for the relevant month.
- (v) Change of ownership The Completion Statement shall exclude any charges, provisions, reserves or write-offs in respect of any costs, liabilities or charges to be incurred after Completion as a consequence of the change of ownership of the Company or any change in the management strategy, direction or priority or possible closure of any part of the Company after Completion, whether or not resulting from the change in ownership For the avoidance of doubt, any amounts becoming payable on (as opposed to after) Completion as a result of the change of control such as prepayment costs payable in relation to redemption of the Tikehau Bonds, bonuses payable to employees on a change of control or any exit or similar fee payable to the Institutional Seller, shall be included.
- (vi)For the avoidance of doubt, there shall be no double counting of assets or liabilities, in particular in the items included in the calculation of Actual Net Debt Amount and Actual Net Working Capital Amount, in the Completion Statement.
- (vii)Classification of line items in the Completion Statement shall be consistent with the classification of line items in the 2013 Audited Accounts except where an asset or liability classification is specified within the Specific Accounting Policies set out in this Schedule 7, in which case the classification included in the Specific Accounting Policies shall override the classification of that asset and liability in the 2013 Audited Accounts.
- (viii) To the extent that a matter or thing is the subject of an indemnity from the Sellers under this Agreement, no account shall be taken of, and no provision or reserve shall be made in respect of, such matter or thing in the Completion Statement, excluding the provision for the German freelancer issue identified as part of Net Debt. In respect of the German freelancer issue the indemnity has been be reduced by the amount provisioned in net debt in order to prevent the double counting of liabilities.

(b) Revenue recognition

Wholesale – revenue shall be recognised when an entity has delivered products to the wholesaler and the wholesaler has full discretion over the channel and price to sell the products, and there is no unfulfilled obligation that could affect the wholesaler's acceptance of the products. In determining when delivery has taken place, the policies

and procedures applied in each local business unit for year-end reporting purposes shall be applied. A provision for product returns shall be recognized as soon as the business authorises such returns, with no delay for either receipt of product or processing of product.

Retail - revenue shall be recognised when an entity has sold a product to the customer.

The Group provides various sales incentives to resellers and consumers which are recognised as a reduction of revenue such as discounts, rebates, coupons, slotting fees, volume based incentives and cooperative advertising.

(c) Inventory

Inventory shall be valued at the lower of acquisition cost or manufacturing cost, and net realisable value.

Net realisable value shall be deemed to be acquisition cost or manufacturing cost, including all costs of purchase, costs of conversion and other costs in bringing the inventories to their present location and condition, less provisions for obsolete and slow moving inventory. In determining manufacturing cost, standard costs shall be used.

- a) Provisions for obsolete inventory shall be calculated as follows:
- Raw materials and packaging for which no sales or utilization is forecast (including by substitution in another product) – 95 % - 100%, on a basis consistent with that applied in the 2013 Audited Accounts
- Raw materials that are damaged and cannot be used or sold, or have passed their sell by date – 95% - 100%, on a basis consistent with that applied in the 2013 Audited Accounts
- Packaging that is damaged and cannot be used or sold, or has passed its sell by date –
 100%
- Raw materials that have a quality issue and are therefore unusable where DBA has no recourse to the vendor 95% 100%, on a basis consistent with that applied in the 2013 Audited Accounts
- WIP that is sitting in the production line (or plant inventory) for which there is no future requirement and that cannot be reprocessed for alternative use – 75%
- Finished goods for which no sale or utilization are forecasted 75%
- Finished goods that are classified as 'seconds' or 'off quality' (and classified as such on a consistent basis to the classification used in the 2013 Audited Accounts) – 75%

Provisions for slow moving inventory shall be calculated as follows:

- Red flag inventory i.e. inventory representing more than 12 months of forecast sales (other than in a limited number of business units which in the year ended 30 June 2013 calculated red flag inventory on the basis of inventory representing more than 12 months of historical sales ("Business Units Applying Historical Sales"), which shall continue to calculate this on the basis of historical sales), and all products of a "fashion" or seasonable nature:
 - i) Raw materials 75%
 - ii) WIP and finished goods 50%
- Excess inventory i.e. inventory representing more than 17 weeks of future sales but less than 12 months of forecast sales (other than Business Units Applying Historical Sales which shall calculate this by reference to historical sales):
 - Raw materials 50%
 - ii) WIP and finished goods 25%

No other provisions for inventory shall be made with the exception of provisions specific to the following countries (on the basis that no double-counting of provisions shall be included):

- France: (a) inventories over expected sales depreciated at 25% for current products; (b) previous collections classified in three categories: discount 25% reserve, out of collection 50% reserve, and non-current 75% reserve; (c) other slow moving: returns 50% reserve; (d) Yellowing issues to be provisioned on a basis consistent with that applied in the March 2014 Management Accounts
- Russia: (a) obsolete items with no sales forecasted 75% reserve; (b) slow moving second last collection 25% reserve

On a consistent basis with that applied in the March 2014 Management Accounts, the calculation for obsolete, slow moving and excess inventory provisions shall be reduced to the extent that management considers the sale of provided products in excess of their NRV is likely to occur.

Except for the specific provisions noted above, no provisions against inventory shall be made in the Completion Statement.

Standard costs are updated annually by the Group at the start of the financial year and the FY15 budget standard costs shall be applied for the purposes of the Completion Statement. Standard costs shall be calculated using the same methodology as that used to prepare standard costs in the 2013 Audited Accounts.

(d) Trade accounts receivable

Provision shall be made for trade receivables as at the Effective Time using the same criteria as that used by management in the preparation of the 2013 Audited Accounts.

Trade receivables of one year or more are provided against in full (100%).

No general bad debt provision shall be included in the Completion Statement, on a basis consistent with that applied in the March 2014 Management Accounts.

(e) Other receivables

To the extent that customers have not been invoiced on or before the Effective Time in respect of goods or services supplied or dispatched on or before the Effective Time, an amount of accrued income shall be included in the Completion Statement.

Prepayments shall be recognised to the extent a payment has been made, or creditor recognised, prior to the Effective Time, but where the relevant goods, benefit or service will not be received until after the Effective Time.

No provisions shall be made against any receivables, other than the specific bad debt provisions referred to above.

The Completion Statement shall exclude any Sri Lanka Net Proceeds, if any, uncollected from the buyer of Isabella (private) Ltd. at the Completion Date.

The Completion Statement shall exclude amounts due from DE Master Blenders 1753 B.V. (as successor to Sara Lee Corporation) in respect of the warranty claim for tax liabilities.

(f) Intercompany balances

No provisions shall be made in respect of intercompany trading balances due to/from the Institutional Seller or any subsidiaries of the Institutional Seller that do not form part of the Group. All intercompany balances shall be reconciled and outstanding receivable/payable balances shall be agreed between respective parties, and prepared using the same criteria as that used by management in the preparation of the 2013 Audited Accounts.

(g) Liabilities

- (i) To the extent that vendors have not invoiced on or before the Effective Time in respect of goods or services (where contractual criteria have been met such that a liability to the vendor exists) on or before the Effective Time, an amount of accrued expense shall be included in the Completion Statement on a consistent basis with that applied in the 2013 Audited Accounts.
- (ii) The Completion Statement shall include an accrual for external loan interest accrued but not paid as at the Effective Time.
- (iii)Leases classified as operating leases in the 2013 Audited Accounts shall not be reclassified as finance leases in the Completion Statement. Leases entered into since 30 June 2013 shall be classified on a consistent basis and using criteria applied in the 2013 Audited Accounts.

- (iv)The Completion Statement shall recognize a liability for rebate accruals, volume discount accruals and marketing and promotion/incentive accruals, which are accounted for on a monthly basis in a manner consistently applied using criteria applied in the 2013 Audited Accounts.
- (v) The Completion Statement shall include a specific provision for restructuring of €1,498,000. No restructuring projects will be added beyond those reviewed and approved by the Purchaser. No other restructuring provisions, accruals or liabilities shall be included in the Completion Statement save in respect of any new restructuring project initiated prior to Completion which has not been approved by the Purchaser.
- (vi)The Completion Statement shall include a specific provision for commercial litigation of €313,000 in respect of a claim relating to the payroll tax status of the agency workforce used by DBApparel in Germany (the Landsversicherungsanstalt).
- (vii) The Completion Statement shall include the Existing Shareholder Debt Redemption Amount.
- (viii) The French R&D Tax Credit Denial Amount included in the Actual Debt Statement shall be the aggregate of the following amounts:
 - (i) an amount equal to any additional corporate tax liability including any related interest and penalties of DBApparel S.A.S., acting as head of the French DBA tax group and sole tax payer vis-à-vis the French treasury, in respect of the Financial Year 2010, which arises as a result of the denial of some or all of the 2009 R&D tax credit generated by DIM S.A.S., as determined in accordance with the most recent decisions of the Ministry of Research or the French Tax Authorities issued as at the date of the Completion Statements in respect of the R&D tax credit audit currently being undertaken at the level of DIM S.A.S. (the "2009 Denied French R&D Tax Credit");
 - (ii) an amount equal to any additional corporate tax liability including any related interest and penalties of DBApparel S.A.S., acting as head of the French DBA tax group and sole tax payer vis-à-vis the French treasury, in respect of the Financial Year 2011, which arises as a result of the denial of some or all of the 2010 R&D tax credit generated by DIM S.A.S., as determined in accordance with the most recent decisions of the Ministry of Research or the French Tax Authorities issued as at the date of the Completion Statements in respect of the R&D tax credit audit currently being undertaken at the level of DIM S.A.S. (the "2010 Denied French R&D Tax Credit");
 - (iii)an amount equal to any additional corporate tax liability including (for Financial Years 2012 and 2013) any related interest and penalties of DBApparel S.A.S., acting as head of the French DBA tax group and sole tax payer vis-à-vis the French treasury, in respect of the Financial Years 2012, 2013 and 2014, which would arise as a result of a denial of some or all of the 2011, 2012 and/ or 2013 R&D tax credits generated by DIM S.A.S. if the

denial were determined by (1) calculating the percentage of denied R&D tax credits in accordance with the most recent decisions of the Ministry of Research or the French Tax Authorities issued as at the date of the Completion Statements for Financial Year 2010 described in i. above, (2) calculating the percentage of denied R&D tax credits in accordance with the most recent decisions of the Ministry of Research or the French Tax Authorities issued as at the Completion Date for Financial Year 2011 described in ii. above, (3) calculating the average of the two percentages calculated in (1) and (2), and (4) applying that average percentage to the R&D tax credits claimed in Financial Years 2012, 2013 and 2014 (the "2011, 2012, and 2013 Denied French R&D Tax Credits").

- (iv)To the extent that an appeal of any decision of the Ministry of Research or the French Tax Authorities is made by DBApparel S.A.S. or DIM S.A.S resulting after Completion in a final decision from any Taxation Authority denying an amount of French R&D Tax Credits for 2009, 2010, 2011, 2012 and 2013 lower than the 2009, 2010, 2011, 2012 and 2013 Denied French R&D Tax Credits, or the actual amount of denied French R&D Tax Credits for those Financial Years is otherwise found after Completion to be lower than the 2009, 2010, 2011, 2012 and 2013 Denied French R&D Tax Credits, the Sellers hereby agree not to request the repayment of any excess French R&D Tax Credit Denial Amount from the Purchaser.
- (v) In the event that the most recent decisions of the Ministry of Research or the French Tax Authorities mentioned in i. and ii. above do not specify the additional corporate tax liability of DBApparel S.A.S. for the relevant Financial Years, such amounts shall be calculated by the Purchaser and the Institutional Seller acting together in good faith and on the basis of the information provided in such decisions. In the event that the Purchaser and the Institutional Seller are required to make such calculations then the amounts calculated in accordance with iii, above shall also be made on the basis of such calculations. The Purchaser and the Institutional Seller shall endeavour to agree such amounts within a reasonable time and in any event not later than 15 Business Days prior to the date on which the draft Completion Statement is submitted to the Institutional Seller in accordance with Schedule 6 to this Agreement. In the event that the Purchaser and the Institutional Seller do not agree such amounts by that date, the Purchaser and Institutional Seller shall jointly appoint Grande Armée Conseil (the "R&D Advisor") to calculate such amounts. If any member of the team assigned by the R&D Advisor to determine the dispute in question has previously worked, or is currently working, on any matter related to the Sellers or the Purchaser or has otherwise performed or is performing services for any person that in any case may present a conflict of interest that is reasonably likely to affect the services to be provided by the R&D Advisor or the unbiased performance of services by any member of the R&D Advisor's team, then the R&D Advisor shall be

nominated by or on behalf of the President of the Institute of Chartered Accountants of England and Wales (or any successor thereto) on the application of either of the Purchaser or the Institutional Seller. The fees and expenses of the R&D Advisor being shared equally by the Purchaser and the Institutional Seller. The R&D Advisor shall act as expert and not as arbitrator and its determination of the amounts pursuant to this sub-paragraph (v) shall be (A) in accordance with the procedures in Paragraphs 9 and 10 of Schedule 6 (*Post Completion Adjustments*) the provisions of which shall be applied mutatis mutandis and (B) final and binding on the Parties save in the event of manifest error (when the relevant part of their determination shall be void and the matter shall be remitted to the R&D Advisor for correction).

- (ix) Any outstanding Transaction related expenses incurred by the Group pre-Completion in respect of the Transaction, including without limitation payments to employees and advisors.
- (x) The Group operates hedges against certain foreign exchange risks. These derivative financial instruments shall be recognised at fair value. Hedge accounting is not applied and all gains/losses are recognised in the profit and loss account. Current corporation tax receivables and payables shall be calculated as if the Effective Time was the end of a tax accounting period. For the avoidance of doubt, all of the calculated corporation taxes due (i.e., receivables and payables) for the current tax year through the Effective Time (less any interim/estimated cash payments) will be accounted for as a current asset or current liability. More generally, no amounts shall be included for deferred tax assets or liabilities.
- (xi) No provision, accrual or liability shall be included in the Completion Statement in respect of the following items:
 - a) Environmental, dilapidation or similar remediation costs;
 - b) Pension deficits or funding shortfalls;
 - c) Other long term benefits, including but not limited to:
 - 1. Jubilee award schemes;
 - 2. The Italian TFR scheme, a statutory service indemnity; and
 - The Italian sales agent commission reserve;
 - d) Amounts owed to DE Master Blenders 1753 B.V. (as successor to Sara Lee Corporation) in respect of deferred consideration;
 - e) Capital expenditure shortfalls or capital commitments;
 - f) Tax litigation risks other than:
 - French R&D tax credit (to be included in Net Debt as described in paragraph 2(g)(viii) above)

- Tax litigation that should be provided for in accordance with IFRS applied on a basis which is consistent with historical practice (to be included in the Completion Statement)
- g) 'Out of balance' assets or liabilities captured by account codes 8368 and 8369.

(f) Option Exercise Consideration

For the purposes of the Completion Statement, the aggregate amount of the Option Exercise Consideration due from all relevant Management Sellers shall be treated as Cash.

Exhibit B

Acceptance Notice Pursuant to paragraph 2.3 of the Offer Letter

[On the letterhead of the Institutional Seller]

To: [•]
Date: [•]
We refer to the letter from MFB International Holdings s.à r.l. (the "Purchaser") and Hanesbrands Inc. (the "Purchaser's Guarantor") addressed to and countersigned by the Institutional Seller) on [•] 2014 concerning the Purchaser's final binding offer to acquire the Shares and Warrants (the "Offer Letter").
Capitalised terms used but not otherwise defined herein shall have the same meanings as ascribed to them in the Offer Letter.
The Institutional Seller hereby provides notice to the Purchaser of the Institutional Seller's intention to accept the Offer and proceed with the Transaction in accordance with paragraph 2.3.
Enclosed with this notice is an original Share Purchase Agreement duly executed by the Sellers and Manager Holding Companies. This notice together with the duly executed Share Purchase Agreement constitutes acceptance of the Offer for the purposes of paragraph 2.3 of the Offer Letter.
Yours faithfully,
SLB BRANDS HOLDINGS, LTD
Date:

CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Richard A. Noll, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Hanesbrands Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Richard A. Noll

Richard A. Noll Chief Executive Officer

Date: July 24, 2014

CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Richard D. Moss, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Hanesbrands Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Richard D. Moss

Richard D. Moss Chief Financial Officer

Date: July 24, 2014

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Hanesbrands Inc. ("Hanesbrands") on Form 10-Q for the fiscal quarter ended June 28, 2014 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Richard A. Noll, Chief Executive Officer of Hanesbrands, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Hanesbrands.

/s/ Richard A. Noll

Richard A. Noll Chief Executive Officer

Date: July 24, 2014

The foregoing certification is being furnished to accompany Hanesbrands Inc.'s Quarterly Report on Form 10-Q for the fiscal quarter ended June 28, 2014 (the "Report") solely pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and shall not be deemed filed as part of the Report or as a separate disclosure document and shall not be deemed incorporated by reference into any other filing of Hanesbrands Inc. that incorporates the Report by reference. A signed original of this written certification required by Section 906 has been provided to Hanesbrands Inc. and will be retained by Hanesbrands Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Hanesbrands Inc. ("Hanesbrands") on Form 10-Q for the fiscal quarter ended June 28, 2014 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Richard D. Moss, Chief Financial Officer of Hanesbrands, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Hanesbrands.

/s/ Richard D. Moss

Richard D. Moss Chief Financial Officer

Date: July 24, 2014

The foregoing certification is being furnished to accompany Hanesbrands Inc.'s Quarterly Report on Form 10-Q for the fiscal quarter ended June 28, 2014 (the "Report") solely pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and shall not be deemed filed as part of the Report or as a separate disclosure document and shall not be deemed incorporated by reference into any other filing of Hanesbrands Inc. that incorporates the Report by reference. A signed original of this written certification required by Section 906 has been provided to Hanesbrands Inc. and will be retained by Hanesbrands Inc. and furnished to the Securities and Exchange Commission or its staff upon request.